



Water dynamics and use in coffee shaded with *Tabebuia rosea* Bertol. and *Simarouba glauca* D.C. compared to full sun coffee in sub optimal environmental condition for coffee cultivation

Maria da Penha Padovan

Water dynamics and use in coffee shaded with *Tabebuia rosea* Bertol. and *Simarouba glauca* D.C. compared to full sun coffee in sub optimal environmental condition for coffee cultivation

Thesis presented by Maria da Penha Padovan to opt for the degree of Doctor of Philosophy at CATIE – BU joint PhD Program

Tropical Agricultural Research and Higher Education Centre (CATIE), Turrialba, Costa Rica

School of Environment, Natural Resources and Geography, Bangor University (BU), Bangor, Gwynedd, United Kingdom

September, 2016

Doctoral Advisory Committee

Dr. Bruno Rapidel (Major Advisor)

CIRAD (Centre de coopération Internationale en Recherche Agronomique pour le Développement)

Montepellier, France.

CATIE (Centro Agronómico Tropical de Investigación y Enseñanza)

Turrialba, Costa Rica.

Dr. Robert Brook (Co-Advisor)

BU (Bangor University - College of Natural Resources)

Bangor, Gwynedd, United Kingdom.

Dr. Aureliano Nogueira da Costa (Committee member)

INCAPER (Instituto Capixaba de Pesquisa, Assistência Técnica e Extensão Rural) Espirito Santo State, Brazil.

Dra. Carolina Vega Jarquín (Committee member)

UNA (Universidad Nacional Agraria)

Managua, Nicaragua.

DECLARATION AND CONSENT

I hereby agree to deposit the following item in the digital repository maintained by Bangor University and/or in any other repository authorized for use by Bangor University.

Author Name: Maria da Penha Padovan

Title: Water dynamics and use in coffee shaded with *Tabebuia rosea* Bertol. and *Simarouba glauca* D.C. compared to full sun coffee in sub optimal environmental condition for coffee cultivation.

Supervisor/Department: Dr. Robert M. Brook - SENRGy

Qualification/Degree obtained: PhD

This item is a product of my own research endeavours and is covered by the agreement below in which the item is referred to as "the Work". It is identical in content to that deposited in the Library, subject to point 4 below.

Non-exclusive Rights

Rights granted to the digital repository through this agreement are entirely nonexclusive. I am free to publish the Work in its present version or future versions elsewhere.

I agree that Bangor University may electronically store, copy or translate the Work to any approved medium or format for the purpose of future preservation and accessibility. Bangor University is not under any obligation to reproduce or display the Work in the same formats or resolutions in which it was originally deposited.

Bangor University Digital Repository

I understand that work deposited in the digital repository will be accessible to a wide variety of people and institutions, including automated agents and search engines via the World Wide Web.

I understand that once the Work is deposited, the item and its metadata may be incorporated into public access catalogues or services, national databases of electronic theses and dissertations such as the British Library's EThOS or any service provided by the National Library of Wales. I understand that the Work may be made available via the National Library of Wales Online Electronic Theses Service under the declared terms and conditions of use (http://www.llgc.org.uk/index.php?id=4676). I agree that as part of this service the National Library of Wales may electronically store, copy or convert the Work to any approved medium or format for the purpose of future preservation and accessibility. The National Library of Wales is not under any obligation to reproduce or display the Work in the same formats or resolutions in which it was originally deposited.

Statement 1:

This work has not previously been accepted in substance for any degree and is not being concurrently submitted in candidature for any degree unless as agreed by the University for approved dual awards.

Signed (candidate)

Date 18.12.2015

Statement 2:

This thesis is the result of my own investigations, except where otherwise stated. Where correction services have been used, the extent and nature of the correction is clearly marked in a footnote(s).

All other sources are acknowledged by footnotes and/or a bibliography.

Signed Machine (candidate)

Date 18.12.2015

Statement 3:

I hereby give consent for my thesis, if accepted, to be available for photocopying, for inter-library loan and for electronic repositories, and for the title and summary to be made available to outside organisations.

Signed Machine (candidate)

Date 18.12.2015

NB: Candidates on whose behalf a bar on access has been approved by the Academic Registry should use the following version of **Statement 3**:

Statement 3 (bar):

I hereby give consent for my thesis, if accepted, to be available for photocopying, for inter-library loans and for electronic repositories after expiry of a bar on access.

Signed (candidate)

Date

Statement 4:

Choose **one** of the following options

a)	I agree to deposit an electronic copy of my thesis (the Work) in the Bangor			
	University (BU) Institutional Digital Repository, the British Library ETHOS system,	V		
	and/or in any other repository authorized for use by Bangor University and where	Х		
	necessary have gained the required permissions for the use of third party material.			
b)	I agree to deposit an electronic copy of my thesis (the Work) in the Bangor			
	University (BU) Institutional Digital Repository, the British Library ETHOS system,			
	and/or in any other repository authorized for use by Bangor University when the			
	approved bar on access has been lifted.			
c)	I agree to submit my thesis (the Work) electronically via Bangor University's e-			
	submission system, however I $\ensuremath{\text{opt-out}}$ of the electronic deposit to the Bangor			
	University (BU) Institutional Digital Repository, the British Library ETHOS system,			
	and/or in any other repository authorized for use by Bangor University, due to lack			
	of permissions for use of third party material.			

Options B should only be used if a bar on access has been approved by the University.

In addition to the above I also agree to the following:

- That I am the author or have the authority of the author(s) to make this agreement and do hereby give Bangor University the right to make available the Work in the way described above.
- That the electronic copy of the Work deposited in the digital repository and covered by this agreement, is identical in content to the paper copy of the Work deposited in the Bangor University Library, subject to point 4 below.

- 3. That I have exercised reasonable care to ensure that the Work is original and, to the best of my knowledge, does not breach any laws including those relating to defamation, libel and copyright.
- 4. That I have, in instances where the intellectual property of other authors or copyright holders is included in the Work, and where appropriate, gained explicit permission for the inclusion of that material in the Work, and in the electronic form of the Work as accessed through the open access digital repository, *or* that I have identified and removed that material for which adequate and appropriate permission has not been obtained and which will be inaccessible via the digital repository.
- 5. That Bangor University does not hold any obligation to take legal action on behalf of the Depositor, or other rights holders, in the event of a breach of intellectual property rights, or any other right, in the material deposited.
- 6. That I will indemnify and keep indemnified Bangor University and the National Library of Wales from and against any loss, liability, claim or damage, including without limitation any related legal fees and court costs (on a full indemnity bases), related to any breach by myself of any term of this agreement.

Signature:

Nadowan

Date: 18.12.2015

Dedication

To Cloves, Pedro and Daniel

To the coffee producer Olindo Padovan in memorian

my inspiration

Acknowledgements

This thesis was produced with partial financial support of the Espirito Santo State Government – INCAPER (Instituto Capixaba de Pesquisa, Assistencia Tecnica e Extensao Rural) and FONTAGRO Caf Adapt Project which is an alliance of Latin American and Caribbean countries that supports research and innovation in agriculture.

Thanks to the Centre of Agronomic Investigation for International Development (CIRAD) to host us in Montpellier and for the additional financial support over the period of this study.

Many thanks to my supervisor Dr. Bruno Rapidel (CIRAD/CATIE) that trusted me and gave me the opportunity to develop this study and to learn so much with him. Thanks for your patience and dedication to produce this work. To Dr. Robert M. Brook (BU) for your guidance, patience and constant words of encouragement. Thanks again for your very especial attention. Thanks to the committee members Aureliano Nogueira da Costa and Carolina Vega Jarquín for your support and attention in Brazil and in Managua.

I am grateful to the staff from Graduate School at CATIE. Thanks are due to Patricia Hernandez, Margarita Alvarado and Marianella Sanabria for their constant support at Department of Agriculture and Agroforestry. Especial thanks to Javier Brenes from Orton library. To Prof. Sergio Vilchez Mendonza and Dr. Fernando Casanoves from Statistics Department.

I also should thank the support from Federación Cooperativas de Ahorro y Crédito (CENECOOP-FEDECARUNA), Instituto Nicaraguense de Tecnologia Agropecuaria (INTA) and Universidad Nacional Agraria (UNA) in Nicaragua. Thanks to Prof. Leonardo Garcia Centeno and Iris from the soil laboratory and Maria Helena from the plant physiology laboratory at UNA. I would like to especially mention Prof. Rodolfo Munguía for his constant support and advice.

Thanks for the staff from CATIE in Nicaragua especially Mirna Barrios who accompanied me in each stage of the experiment. Thanks to Dalia Sanchez to share your experience. To Ledis Navarrete and Elvin Navarrete for the opportunity to work and learn together. Thanks also to Mauricio Xavier Martinez, Alejandro and Dimas for your help in the fieldwork. Very especial thanks to my assistant in the field Víctor Manuel Jirón Cortez who didn't measure any efforts to support me at any time.

Thanks to Luis Sepulveda and Hebler Narvaez for their help and to host us at Instituto de Capacitación y Desarrollo Rural e Integral (ICIDRE) - Universidad Politécnica de Nicaragua.

Thanks are due to Ana Cecilia Tapia Fernández for the support and the laboratory facilitation at Universidad de Costa Rica (UCR).

Thanks to Jael Bildad Cruz Castilho, Luiz Ramos and Elvin Lagos - students from UNA – Managua and to Anna C. Deffner from SupAgro – Montpellier that integrated their professional practice/graduation project to our experiment over the period of study.

Many thanks to the members of postgraduate committee at INCAPER that understood the need of this investigation including Dr. Antonio Elias de Souza, Dr. Jose Ayres Ventura, Dr. Jose Braz Venturim, Dr. Jose Sergio Salgado. Thanks to Dr. Rogerio Guarçoni for statistical support at Incaper.

To Elisa Bartolozzi, Martha Tristão e Hans J. Andreas Schneebeli that facilitated the videoconferences at UFES.

Especial thanks to Zambeze Almeida, Carlao Henrich and Maria Amélia Almeida for your love and unconditional support in Lisbon and as well as Peter Avilez and Alma Ramirez in Costa Rica. Thanks to my friends Marilyn Villalobos, Elias de Melo and Eliane, Kimi Hamada, Miguel Ditta, Ian, Pilar Bucheli Leon and Karlita, Fredy Argotty, Maria Cristina China, Betinho Milled, Lelia Moura, Esther Miranda, Eduardo, Joao Tocha and Sonia, Praguita Chhetri, Patricia Aldeguer for your help and friendship. Thanks Dra. niña Hilda Lezcano for your constant advices.

No words to thanks my family especially Cloves, Pedro and Daniel to accompany me in this adventure.

I'm grateful to the "three Marias" of my life: Martha, Madalena, Dida. I'm nothing without them.

Forever grateful to my father.

SUMMARY

Water availability is predicted to be reduced and temperature to rise in the global climate change context. Future climate conditions may thus represent a serious risk for coffee cultivation especially in less favorable environment. Agroforestry has been postulated as a promising strategy to adapting to climate changes. Shade tree may minimize radiation and temperature near the soil surface and reduce soil evaporation. Shade tree may enhance infiltration, reduce runoff and increase rainfall water use efficiency by taking up water from deep soil layers. However, shade tree may reduce the water that reaches the soil by rainfall interception. Trees may consume additional water and can establish a competitive relationship depending on tree species characteristics, soil water availability, site conditions and management.

In this experiment water dynamics and use was monitored in a mature agroforestry experiment where coffee shaded by a mixture of *Tabebuia rosea* and *Simarouba glauca* is compared to full sun coffee over 2012 and 2013. The water balance was obtained by two independent approaches: 1) measuring directly all components of water balance (trees and coffee transpiration; soil evaporation; rainfall interception); and 2) measuring changes in the soil water stock through Time Domain Reflectometers (TDR) probes.

Agroforestry (AFS) showed greater transpiration and lower soil surface evaporation compared to full sun (FS). Shade tree did not represent a serious constraint for coffee water use during most of the period of the experiment. Coffee water consumption represented 75% of the total transpiration in agroforestry while *Tabebuia rosea* transpired 17% and *Simarouba glauca* 8%. Complementarity was demonstrated by root niche differentiation between coffee and *Simarouba glauca* that seemed to be more suitable as coffee shade tree compared to *Tabebuia rosea*. We also demonstrated high competition between coffee and shade tree when an atypical very dry season occurred. Transpiration was stabilized although the high evaporative demand and coffee leaf water potential reached its lowest value in AFS which suggested high level of coffee water stress. Adaptation strategies for coping with climate change using shade trees need to be devised taking into account this quantified information into account.

RESUMEN

En el contexto del cambio climático global disminución de la disponibilidad de agua y el aumento de la temperatura han sido esperados en el futuro próximo. Variaciones en las condiciones climáticas futuras pueden por lo tanto representar un grave riesgo para el cultivo del café, especialmente en condiciones menos favorables. Los sistemas agroforestales han sido postulados como una estrategia promisora para la adaptación a los cambios climáticos. Arboles de sombra pueden minimizar la radiación y la temperatura cerca de la superficie del suelo y reducir la evaporación. Además pueden mejorar la infiltración, reducir la escorrentía y aumentar la eficiencia del uso del agua de lluvia, tomando el agua de las capas profundas del suelo. Sin embargo, árboles de sombra puede reducir el agua que llega al suelo mediante la interceptación de la lluvia. Los árboles pueden consumir mucha agua y pueden establecer una relación de competencia en función de las características de las especies de árboles y la disponibilidad de agua del suelo. En este experimento, la dinámica y el uso del agua fueron monitoreados en café bajo la sombra de Tabebuia rosea y Simarouba glauca comparados con el café a pleno sol durante 2012 y 2013. El balance hídrico se obtuvo mediante dos métodos independientes: 1) se midió directamente los componentes del balance hídrico (transpiración de café y árboles, la evaporación del suelo y la intercepción de lluvia); y 2) se midió el cambio en el contenido de agua del suelo por medio de sensores TDR - Time Domain Reflectometers. Es sistema agroforestal presentó mayor transpiración y menor evaporación de la superficie del suelo en comparación con café pleno sol. Árboles de sombra no representaron una limitación para el uso del agua de café durante la mayor parte del período del experimento. El consumo de agua del café representó el 75% del total de la transpiración en AFS mientras que Tabebuia rosea transpiró 17% y Simarouba glauca 8%. La complementariedad fue demonstrada por la diferenciación de nicho de raíces de café y Simarouba glauca la cual pareció ser más adecuada como árbol de sombra para el café en comparación con Tabebuia rosea. Se demostró además una competencia potencial entre café y árbol de sombra cuando se produjo una estación atípica muy seca. La transpiración se estabilizó aunque la gran demanda evaporativa y el potencial de agua en la hoja del café alcanzó su valor más bajo en AFS lo que sugirió alto nivel de estrés hídrico en el café. Las estrategias de adaptación para hacer frente al cambio climático utilizando árboles de sombra deben ser concebidas teniendo en cuenta esta información.

Declaration and consent ii
Dedicationvi
Acknowledgements vii
Summaryix
Resumenx
CHAPTER I1
1. GENERAL INTRODUCTION1
1.1. INTRODUCTION1
1.2. CONTEXT OF THE RESEARCH4
1.3. OBJECTIVES
1.4. HYPOTHESES
1.4.1. Roots distribution and water use in the full sun coffee compared to coffee agroforestry
1.4.2. Rainfall interception, stemflow and throughfall in a coffee shaded by <i>Tabebuia rosea</i> Bertol. and <i>Simarouba glauca</i> DC
1.4.3. Evapotranspiration in a coffee shaded by <i>Tabebuia rosea</i> and <i>Simarouba glauca</i> timber tree species compared to unshaded coffee in sub optimal environmental conditions
1.5. GENERAL OUTLINE OF THE THESIS
CHAPTER II9
LITERATURE REVIEW9
2.1. COFFEE WATER RELATIONS
2.1.1. Coffee transpiration10
2.1.2. Coffee leaf water potential11

2.2. CLIMATE IMPACTS ON COFFEE GROWTH AND YIELD
2.3. WATER DYNAMICS IN COFFEE AGROFORESTRY15
2.4. COFFEE AGROFORESTRY WATER BALANCE16
2.5. COFFEE SHADE TREE CHARACTERISTICS AND PERFORMANCE
2.6. THE BELOW GROUND RELATIONSHIP IN COFFEE AGROFORESTRY20
2.7. GAPS IN THE KNOWLEDGE OF COFFEE AGROFORESTRY WATER RELATIONS21
CHAPTER III
ROOT DISTRIBUTION AND WATER USE IN COFFEE SHADED WITH <i>TABEBUIA ROSEA</i> BERTOL. AND <i>SIMAROUBA GLAUCA</i> DC. COMPARED TO FULL SUN COFFEE IN SUB- OPTIMAL ENVIRONMENTAL CONDITIONS
3.1. INTRODUCTION23
3.2. MATERIAL AND METHODS25
3.2.1. Site description25
3.2.2. Experimental design27
3.2.3. Tree species studied29
3.2.4. Fine root distribution
3.2.5. Soil water content
3.2.5. Data analysis
3.3. RESULTS AND DISCUSSION
3.3.1. Coffee and tree fine root distribution
3.3.2. The effect of talpetate on coffee root distribution
3.33. Soil water content
3.3.4. Soil water uptake
3.4. CONCLUSIONS

3.5. REFERENCES	39
CHAPTER IV	43
RAINFALL INTERCEPTION, STEMFLOW AND THROUGHFALL IN WITH <i>TABEBUIA ROSEA</i> BERTOL. AND <i>SIMAROUBA GLAUCA</i> DC. SUN COFFEE	Compared to full
4.1. INTRODUCTION	43
4.2. MATERIALS AND METHODS	45
4.2.1. Site description and experimental design	45
4.2.2. Species studied	46
4.2.3. Rainfall	47
4.2.4. Leaf area index	48
4.2.5. Stemflow	49
4.2.6. Throughfall	49
4.2.7. Interception loss	51
4.2.8. Statistical analysis	51
4.3. RESULTS	51
4.3.1. Leaf area index	51
4.3.2. Stemflow	52
4.3.3. Throughfall	54
4.3.4. Rainfall interception	58
4.3.5. Rainfall interception modeling	61
4.4. DISCUSSION	65
4.4.1. Stemflow	65
4.4.2. Throughfall	66

4.4.3.	Interception loss	.67
4.5.	CONCLUSIONS	.69
4.6.	REFERENCES	69
CHAP	TER V	73
SHADI	R LOSS BY PLANT TRANSPIRATION AND SOIL EVAPORATION IN COF ED BY <i>TABEBUIA ROSEA</i> BERTOL. AND <i>SIMAROUBA GLAUCA</i> DC. COMPARED ADED COFFEE IN SUB-OPTIMAL ENVIRONMENTAL CONDITIONS	то
5.1.	INTRODUCTION	.73
5.2	MATERIALS AND METHODS	.74
5.2.1	Site description and experimental design	.74
5.2.2	Species studied	.76
5.2.3	Climate	.77
5.2.4.	Soil water content	.77
5.2.5.	Coffee and tree leaf area index	.77
5.2.6.	Coffee and tree transpiration	.79
5.2.7.	Leaf water potential	.81
5.2.8.	Soil evaporation	.81
5.2.9.	Data analysis	.82
5.3.	RESULTS	82
5.3.1.	Climate	.82
5.3.2.	Soil water content	84
5.3.3.	Coffee and tree leaf area index	.84
5.3.4.	Coffee and tree transpiration	.86
5.3.5.	Soil evaporation	90

5.3.6. Coffee leaf water potential94
5.3.7. Total evapotranspiration95
5.4. DISCUSSION
5.4.1. Coffee water use97
5.4.2. Shade tree water consumption
5.4.3. Soil surface evaporation100
5.4.4. Total evapotranspiration101
5.5. CONCLUSIONS
5.6. REFERENCES
CHAPTER VI107
6. GENERAL DISCUSSION AND CONCLUSIONS107
6.1. Study site and experimental design107
6.2. Complementarity and competition for water in coffee shaded with <i>Tabebuia rosea</i> and <i>Simarouba glauca</i> in sub optimal environmental condition
6.2.1. Below-ground interactions108
6.2.2. Coffee water use110
6.2.3. Shade tree water use112
6.2.4. Shade tree effects on coffee production in sub optimal environmental condition for coffee cultivation
6.3. Water dynamics and use in coffee agroforestry in sub optimal condition115
6.3.1. Evapotranspiration partitioning115
6.3.2. Evapotranspiration by directly and indirect estimation compared118
6.4. GENERAL CONCLUSIONS120
6.4.1. TESTING THE RESEARCH HYPOTHESES121

6.5.	CONSIDERATIONS	AND	RECOMMENDATIONS	FOR	FUTURE	RESEARCH	ON
COFFEE WATER RELATIOS IN AGROFORESTRY123							
REFE	ERENCES (Chapters I	, II, VI	[)				125

LIST OF TABLES

Chapter IV

Chapter V

5.2. Total evapotranspiration calculated from transpiration plus soil evaporation in a plot basis (with the standard error of the mean) and as percentage of total

LIST OF FIGURES

Chapter I

Chapter III

3.2. Schematic representation of the experimental design of (one quarter of) the shade plot with coffee, *Simarouba glauca* and *Tabebuia rosea* spatial distribution pattern....28

Chapter IV

4.10. Spatial distribution of throughfall beneath *Tabebuia rosea* (TR) as a function of gross rainfall at different distances from the tree trunk (1.0 m, 2.5 m and 5.5 m)......58

4.11. Spatial distribution of throughfall beneath *Simarouba glauca* (SG) as a function of gross rainfall at different distances from the tree trunk (1.0 m, 2.5 m and 5.5 m)......58

4.12. Rainfall interception by coffee trees in FS and AFS over the period of study......59

4.14. Interception measured and modelled by coffee tree canopies in FS......62

4.15. Rainfall interception measured and modelled by Simarouba glauca canopy......63

4.16. Rainfall interception measured and modelled by Tabebuia rosea canopy...........63

Chapter V

5.12. Soil evaporation measured and modelled in FS coffee. Regression equation: $y=0.759x + 0.5385; R^2=0.44.....93$

5.13. Soil evaporation measured and modelled in AFS. Regression equation: y= 0.9804x + 0.3857; $R^2=0.62$94

5.14. Mean coffee leaf water potential at predawn and midday from three days consecutive measurements in the dry (February and July) and wet seasons (July and November) in 2012 and 2013. Bars represent the standard error of the mean.........95

Chapter VI

CHAPTER I

GENERAL INTRODUCTION

1.1. INTRODUCTION

World population is forecast to exceed nine billion people by 2050 (FAO 2009). The global agricultural production will have to increase by about 70% from its 2005 level and nearly 100% in developing countries to guarantee global food security (FAO 2009). Thirty percent of the earth's land surface is used for agriculture. The expansion of food production is the biggest factor responsible for increasing deforestation worldwide that in turn is considered to be one of the contributing factors to global climate change (FAO 2013). Fewer forests mean larger amount of greenhouse gases entering the atmosphere. Also, trees play an important part in the water cycle, absorbing the water via their roots and releasing it into the atmosphere. Without forests, the climate may become dryer in some regions.

Furthermore, 70% of the extracted freshwater is used for irrigation for food production. But irrigation systems dependent on groundwater are at risk once aquifers have been depleted by over pumping. The world's irrigated agriculture tripled from 1950 to 2000, has stabilized since then and begun to decline in some countries. Falling water tables are forcing a number of countries, that were once self-sufficient in grain, to become heavily dependent on imports (FAO 2013). In most countries where rain fed agriculture is a common practice, the effects of change in the patterns of temperature and precipitation increase the vulnerability of available water to match the agricultural requirements (Lin *et al.* 2008).

In such a context, there is a need to maximize food production with efficient land use and water consumption. The combined production of crops with tree species and animals on the same site is a traditional land-use practice which provides heterogeneous agricultural landscapes with great potential for ecosystem services maintenance such as natural pest management, carbon sequestration, biodiversity, water and soil conservation (Harvey *et al.* 2008). Greater complexity leads to greater stability of ecosystems in the context of the global climate change. Studies on the effects of the hurricane Mitch in Nicaragua and Honduras in 1998 demonstrated that intercropping and integrated farming systems may improve resistance of extreme climate events. Traditional, mixed farms retained more topsoil, higher soil moisture, more vegetation within the system and therefore lower economic losses when compared to intensively managed systems (Holt-Giménez 2002). Also, the lower vulnerability of agroforestry systems is closely related to lower external inputs such as water and nutrients that are cycled within the system (Tilman *et al.* 2002).

Water loss may be minimized in agroforestry by reducing runoff, soil evaporation, and the transpiration demand and by increasing efficiency in soil water use (Beer 1987). Researches have demonstrated improvements in soil physical properties and enhancing infiltration in agroforestry (Sklenicka et al. 2002; Udawatta et al. 2006; Kumar et al. 2008). Litter cover combined with dense root systems slowed runoff, reducing susceptibility to erosion and diminish nutrient and water loss (Udawatta et al. 2002). Shade trees minimized solar radiation and temperature near the soil surface and, consequently reduced soil evaporation (Ritchie 1971; Jackson and Wallace 1999). Coffee shade trees also reduced radiation incidence, enhancing air humidity and, thereby reduced the vapor pressure deficit and potential crop transpiration rates (Gutiérrez and Meinzer 1994). Studies in coffee agroforestry in Costa Rica showed that shade trees contributed to diminishing of solar radiation incidence on the coffee leaf surface. This reduced leaf temperature by up to 7°C under shade when compared to the open system, depending on time of the day and leaf position (Siles et al. 2009). That is very important since leaf high temperature favors excessive water loss in the gas exchange process. Moreover, trees may contribute to greater efficiency in soil water use by taking up water and nutrients that would otherwise be lost by leaching. This benefit from agroforestry, however, depends on root niche differentiation in which shallow rooting crops and deep tree root systems have a complementarity relationship in resource use, extracting water and nutrients from different layers of the soil profile (Sanchez 1995; Cannell et al. 1996; Schaller et al. 2003).

Traditional shaded coffee cultivation that imitated its understory native condition (Sylvain 1955) has been progressively changing to more intensive systems. The belief that more light would prevent fungal diseases led to a massive conversion of coffee cultivation towards full sun production in Brazil in the 1950s (Da Matta 2004). In Mexico, Colombia, the Caribbean and Central America, unshaded coffee was extended in the early 1990s motivated by increasing prices on the international market (Rice and Ward 1996). As pressure to grow coffee increased, it inevitably spread into less suitable environments for coffee production such as unfavorable temperatures and water shortages. However, the current and potential impacts of global warming,

especially in unsuitable environmental conditions for coffee cultivation, associated with external market forces that lower coffee prices due to increased production, and the increasing trend toward green consumerism, resulted in calls to reduce agrochemicals use and renewed the interest in shaded coffee production (Muschler and Bonnemann 1997; Haggar *et al.* 2001).

Moreover, despite coffee not being a food crop, by value it is the most important agricultural commodity in the international market and involves more than 60 million people in 80 countries on four continents. Family and smallholder farming dominate the vast majority of coffee growing worldwide. In El Salvador, 74% of coffee farms are smaller than 7 ha (Méndez 2008). In Guatemala and Costa Rica more than 50 and 92% respectively are smaller than 5 ha (Samper 1999; Fisher and Victor 2014) and in Mexico 99% of coffee farmers manage less than 10 ha (Lin 2007). Secondary incomes from coffee production such as firewood, fruit and timber, are very important for small-scale producers. It has been recognized that family farming comes closer to diversified agriculture and sustainable food production than intensive, industrialized agriculture. Nevertheless, investigations on coffee crop and shade timber tree interactions are remain few.

Competition for water between coffee and shade trees depend on tree species characteristics, soil water availability, site characteristics and management (Ong 1995). In optimal environmental conditions the risk to coffee cultivation was found to be minor and no competition for water between coffee and shade trees was reported (Siles et al. 2009). In sub optimal environmental conditions for coffee cultivation little information is available. It is known that all components of ecosystem water balance are affected by shade trees (Rutter 1963; Jaramillo and Chaves 1999). For example, water balance comparisons between coffee monoculture (FS) and coffee shaded with Inga densiflora (AFS) in optimal environmental conditions for coffee cultivation in Costa Rica were carried out by using rainfall interception, transpiration, runoff and soil water content measurements. It was found that total water use varied from 30 - 35% in full sun (FS) to 40 - 45% in agroforestry system (AFS) of the annual gross rainfall. Lower interception and more rainfall reaching the soil increased the runoff to 7.5% - 9.5% in FS compared to 3.5% - 5.5% in AFS. Soil water uptake was enhanced at soil water between 100 and 200 cm depth in AFS, which suggested complementarity in water use by coffee and shade trees (Cannavo et al. 2011).

Competition or a complementarity relationship between coffee and shade trees is influenced by water balance calculations based on changes in soil water content due to inputs (rainfall) versus outputs (evaporation, transpiration and runoff) over a particular period of time. However, studies that independently measure all components of the agroforestry water balance are limited. Studies on coffee water relations in different environmental conditions focused on shade timber tree effects will allow the development of multi-functional agriculture able to cope with the forecast lack of water for production, maintenance of current forest lands and increased incomes for farmers.

1.2. CONTEXT OF THE RESEARCH

The investigation was conducted in sub-optimal conditions for *Coffea arabica* cultivation which can be summarized as:

- Mean annual temperature of 27^oC. This is higher than normally recommended for *Coffea arabica*;
- 455 m altitude. This is considered to be rather low for *Coffea arabica* cultivation at this latitude (11⁰ 53' 54" N);
- Mean annual rainfall of 1470 mm with a six-month long dry season (which on average receives only 9% of the annual rainfall);
- Rooting depth was limited to 2 m depth, below which a pan constituted by coarse volcanic ejecta impeded root penetration;
- Additionally, a hardened layer (*talpetate*) crossed this accessible 2 m layer and further impeded root growth.

Originally in the Ethiopian tropical forests, *Coffea arabica* is adapted to altitudes of 1600 to 2800 m, air temperature about 20^oC and rainfall ranging from 1600 to more than 2000 mm per year. Differences in the rainfall pattern over the period of study, being 968 mm in 2012 and 1312 mm in 2013, provided the opportunity to compare water dynamics in relatively normal and abnormally dry conditions in FS and AFS (Fig 1.1).



Fig 1.1. Monthly rainfall distribution from the climate station (in black) and long term mean annual rainfall (in white) in Masatepe, Nicaragua, over the period of study.

In the long-term research area in which the study was carried out, the 12-year old agroforestry experiment allowed analysis and comparison of the performance of coffee unshaded and shaded by a mixture of evergreen *Simarouba glauca* DC. and deciduous *Tabebuia rosea* Bertol. under sub optimal environmental conditions. Both shade tree species are widely utilized in Central America region for timber and other products but are poorly studied, and never before in coffee agroforestry systems.

At the study site, soils are characterized by the presence of a compact soil layer at intermediate and variable depth locally known as *talpetate*, formed by the cementing action of calcium carbonate on volcanic ash. The *talpetate* layer occurs in 15% of the soils of the Pacific coastal plain in Nicaragua and is quite common in different soils of Central America and Mexico. Tests crops of maize and sugar cane indicated that the presence of *talpetate* affected rooting conditions, the nutrients and the moisture availability and the potential for mechanization of the soils (Vogel and Acuña Espinales 1995). However, there is a lack of information about the effects of the *talpetate* on coffee root growing.

Such environmental conditions are representative of one site class that has been utilized for coffee cultivation in Nicaragua, in Central America and other regions highly susceptible to climate changes impacts in which water resources are forecast to diminish. Location of studies in agriculturally sub-optimal environments is important, as these are the conditions under which many farmers have to operate. The results may help farmers and researchers to better understand how to maximize coffee production with efficient water use in regions with similarly unfavorable environments for coffee cultivation.

1.3. OBJECTIVES

The central aim of this investigation was to evaluate water relations and use in full sun coffee compared to coffee shaded with deciduous *Tabebuia rosea* Bertol. and evergreen *Simarouba glauca* DC. which have been utilized as shade trees in Central America but have never been studied before in coffee agroforestry. The study also aimed to evaluate below ground interactions between coffee and shade tree species in the presence of a hard compact soil layer on soil water uptake and root system distribution.

1.3.1. Specific objectives

- To measure and compare coffee root distribution in full sun coffee and agroforestry.

- To assess *Tabebuia rosea* and *Simarouba glauca* root system spatial distribution and its influence on coffee root system development in agroforestry.

- To evaluate the influence of shade tree root system on coffee water uptake facilitation in a compact soil layer present in the study site locally called talpetate.

- To evaluate *Tabebuia rosea* and *Simarouba glauca* shade tree effects on coffee transpiration and soil surface evaporation rate in agroforestry compared to full-sun coffee.

- To measure and compare water loss by rainfall interception in coffee shaded by *Tabebuia rosea* and *Simarouba glauca* and full-sun coffee.

- To evaluate coffee, *Tabebuia rosea* and *Simarouba glauca* influences on water loss by evapotranspiration in sub optimal environmental condition and possible competition between coffee and shade tree.

1.4. HYPOTHESES

1.4.1. Root distribution and water use in the full sun coffee compared to coffee agroforestry

We hypothesized that coffee and tree roots explore different spatial niches; and that evergreen *Simarouba glauca* and deciduous *Tabebuia rosea* tropical timber tree species exhibit different root system behaviors that differentially influence the coffee root distribution. Overall, tropical timber tree species behavior (as distinct from shade tree species, typically leguminous) as coffee shade trees has been poorly investigated. Moreover, in the study area there is a compact soil horizon, locally called *talpetate*, which is typical of the soil conditions in the coffee cultivation Carazo region in Nicaragua. We hypothesized that under restrictive soil conditions, tree roots penetrate through the hard-pan and facilitate coffee water uptake throughout the soil profile. We also hypothesized that soil water uptake by coffee in agroforestry system depends on the shade tree root system characteristics and the rainfall distribution and intensity over the year.

1.4.2. Rainfall interception, stemflow and throughfall in coffee shaded by *Tabebuia rosea* Bertol. and *Simarouba glauca* DC.

We hypothesized that tree canopies modify water availability for coffee cultivation in an agroforestry system; and that rainfall partitioning in interception, stemflow and throughfall affects coffee water use in shaded system. Little is known about the importance of rainfall partitioning in coffee water use in agroforestry and no information was found about the influence of *Tabebuia rosea* and *Simarouba glauca* as shade trees on water loss by rainfall interception in coffee agroforestry compared to full sun coffee.

1.4.3. Water loss by plant transpiration and soil evaporation in coffee shaded by *Tabebuia rosea* Bertol. and *Simarouba glauca* D.C. compared to unshaded coffee in sub optimal environmental conditions

We hypothesized that deciduous *Tabebuia rosea* and evergreen *Simarouba glauca* timber tree species modify the water budget in different ways and thus their suitability as coffee shade trees in the prevailing sub optimal environmental conditions. We also hypothesized that water loss by plant transpiration and soil surface evaporation may

lead to a competitive relationship between coffee and shade trees depending on tree species' characteristics, soil conditions and available water.

1.5. GENERAL OUTLINE OF THE THESIS

This dissertation was divided into six Chapters as follows: Chapter I which presents a general introduction including context of the research, objectives and hypotheses; Chapter II reviews the main topics related to water dynamics in agroforestry systems, water relations in coffee, shade tree performance and characteristics and the impacts of global climate change in coffee water use. Chapter III presents the topic Root distribution and water use in a coffee shaded with *Tabebuia rosea* Bertol. and *Simarouba glauca* DC. compared to full sun coffee in sub-optimal environmental conditions (published in Agroforestry Systems in 2015). Chapter IV discusses the partitioning of rainfall, in particular rainfall interception, stem flow and throughfall as affected by coffee, *Tabebuia rosea* and *Simarouba glauca* compared to full sun coffee while Chapter V discusses partitioning of soil water into plant transpiration and soil evaporation in unshaded coffee and coffee shaded by the timber tree species *Tabebuia rosea* and *Simarouba glauca*. The Chapter VI consists of a general discussion about the most important findings of the experiment and conclusions with recommendations for future research on the topic.

CHAPTER II

LITERATURE REVIEW

2.1. COFFEE WATER RELATIONS

Originally arabica coffee developed as an understorey shrub in Ethiopian tropical forests at altitudes of 1600–2800 m with an average temperature of around 25^oC and well distributed 2000 mm rainfall but with three or four consecutive dry months (Sylvain 1955). Only relatively recently has coffee been grown in the open condition (i.e. without tree shade), including regions closer to the equator in which two annual dry seasons may occur (Cannell 1985).

Most coffee growing areas (except countries near the equator) exhibit the same phenological cycle in which a period of water stress, induced either by dry soil or dry air, seems to be obligatory to prepare flower buds for blossoming. High concentration of endogenous abscisic acid (ABA) maintains the sexual structure of flower buds protected from dry season effects in a dormant state known as hydro-periodism (Alvim 1973; Crisosto *et al.* 1992). Studies have shown that buds need to experience a critical low level of water potential for flowering, approximately -1.2 MPa, which allows a rapid influx of water into the buds at the time of rainfall or irrigation. But too severe water stress and exposure to high temperature may cause abortion of flowers with impacts on coffee yields (Carr 2001).

The understanding of water relations in coffee is paramount for coffee cultivation since small changes in water availability at key periods of crop development may result in negative effects on coffee grown even without apparent signs of water deficit (Da Matta *et al.* 1993). Physiological aspects of coffee plants and their water requirements were first studied by Nutman (1937) and subsequently by many authors (Wormer 1965; Bierhuizen *et al.* 1969; Meinzer *et al.* 1990; Crisosto *et al.* 1992; Gutiérrez and Meinzer 1994; Tausend *et al.* 2000; Carr 2001; Da Matta *et al.* 2007) by using different techniques. Those findings were complemented by the contributions from Fanjul *et al.* (1985), Barradas and Fanjul (1986), Van Kanten and Vaast (2006), Gutierrez (2007), Siles (2007), Franck and Vaast (2009), Lin (2010) and Charbonnier (2013) among other authors who compared coffee water relations in shaded and unshaded environments. In the following paragraphs, I will try to order and resume this wealth of scientific information.

2.1.1. COFFEE TRANSPIRATION

Transpiration in a range of woody species has been determined by using heat as a tracer for sap movement. The methods commonly employed are the stem heat balance, trunk sector heat balance, thermal dissipation technique and heat-pulse method (Smith and Allen 1996). Gutiérrez *et al.* (1994a) successfully measured coffee sap flow by using stem heat balance in greenhouse and field experiments. Coffee transpiration was also measured by using the potometer method (Lin 2010). However, comparison between both techniques showed more accuracy in the measurements by the stem heat balance while the potometer method displayed operational difficulties and low precision in scaling up data for the plot level (Goulden and Field 1994).

A number of experiments have been conducted to determine which endogenous and exogenous factors govern transpiration. The results are somewhat contradictory. For example, field experiments showed that coffee was able to keep relatively high levels of transpiration when evaporative demand was high even under severe water deficit (Gutiérrez and Meinzer 1994). It was reported that transpiration rate of arabica coffee plants in greenhouse experiments remained at 80% of its maximum when soil water content was about 30%. Relative turgidity decreased from 92% to 80% when soil water declined from field capacity to wilting point (Bierhuizen *et al.* 1969). Other studies showed that transpiration started to decline only when soil moisture dropped to 20% (Nunes and Duarte 1969). The apparent independence of transpiration rate from the soil water content was also reinforced by other studies in which the low soil water availability over the dry season did not have any significant effect on reducing transpiration in coffee monoculture (Van Kanten and Vaast 2006).

The influence of the leaf area index, hydraulic architecture and properties of the waterconducting pathway in coffee transpiration has been demonstrated. A study in three coffee cultivars (Typica, San Ramon and Yellow Caturra) showed that the differences in daily transpiration rates seemed to be governed by the apparent hydraulic conductance of the soil-leaf pathway rather than the leaf area. Of the three cultivars studied Typica exhibited the highest rates of sapflow, the more open crown, the least vulnerability of hydraulic conductivity and the lowest value of the ratio of leaf area to sapwood area (Tausend *et al.* 2000). Conversely, another study on transpiration (T) as a component of evapotranspiration (ET) expressed as the ratio T/ET at different values of LAI showed that transpiration increased from 40% to 95% of ET while LAI increased from 1.4 to 6.7 in coffee grown in hedgerow configuration (Gutiérrez and Meinzer 1994).

Responsiveness to vapor pressure deficit (VPD) in coffee indicated that air humidity could play the major role in controlling stomatal aperture (Fanjul et al. 1985; Gutiérrez and Meinzer 1994). The influence of different micro environments with different temperatures and VPD on coffee transpiration depending on leaf position and location was demonstrated by simulations based on a three dimensional coffee canopy architecture by Dauzat et al. (2001). Typical values of stomatal conductance were found to be between 0.10 and 0.15 mol $m^2 s^{-1}$ and maximum values were usually attained prior to midday followed by a steady decline during the afternoon (Gutiérrez et al. 1994b). Another study demonstrated high values of stomatal conductance in the early morning hours that started to decline at around 9 – 10 h with increasing in VPD (Barros et al 1997). Nevertheless, values of air saturation deficit at around 1.5 kPa led to a reduction in stomatal conductance in arabica coffee even in soil at field capacity (Carr 2001). A similar threshold of saturation deficit of air was observed in the dry period when coffee transpiration decreased and ceased to follow the reference evapotranspiration pattern either in full sun or in the tree-shaded environment (Van Kanten and Vaast 2006). Nevertheless, coffee water use and performance in different environmental conditions is still considered to be poorly understood.

2.1.2. COFFEE LEAF WATER POTENTIAL

Coffee growth and yield are determined by plant water balance but, plant water status varies constantly over time. The leaf water potential (LWP) is the driving force for the liquid water movement through the plant. Conceptually LWP is a function of flux rate and internal resistances and represents the thermodynamic expression of tissues water status. It is assumed that during the night hydraulic equilibration takes place so that before sunrise the water in the plant is in equilibrium with the soil water potential and water consumption is minimal, and at solar noon it is expected that water consumption reaches its highest level (Elfving *et al.* 1972).

Da Matta *et al.* (1993) proposed that the ability of coffee to retain high leaf water potential under dehydrating conditions was related to osmotic adjustment which is defined as a net increased solute concentration that serves to maintain turgidity and therefore it is expected to favor drought tolerance. It is accepted that, in some species, accumulation of osmotically active solutes such as proline, for example, has been considered as an adaptation to water shortage (Hare *et al.* 1998). Nevertheless, proline concentration in cultivar Catuai did not change the capacity of coffee plants to withstand drought. This amino acid (proline and others) accumulation was considered to be an injury response imposed by water limitation rather than a mechanism against drought stress (Mazzafera and Teixeira 1989). Moreover, osmotic adjustment is not a general trait observed in coffee genotypes. A study using five arabica coffee cultivars (Catuai, Guatemalan, Mokka, San Ramon and Yellow Caturra) in the field in drying soils showed great variation in LWP between the different cultivars and in the rates at which gas exchange activity declined. The most severe leaf water deficit was observed in Mokka cultivar which reached LWP of -2.60 MPa (± 0.07) at midday and -1.65 MPa (± 0.20) at predawn in drought condition (Meinzer *et al.* 1990). The study suggested that coffee leaf water potential maintenance was related to water use rates and soil water extraction efficiency in the different cultivars.

Coffee leaf water status maintenance has also been attributed to the ability of coffee plants to lower cellular wall elasticity under water limiting condition (Meinzer *et al.* 1990). However, osmotic and elastic adjustments and efficiency in water absorption and use may be not enough to couple with a gradual diminishing in soil water availability without an efficient stomatal control of transpiration (Da Matta 2004).

The somewhat contradictory results demonstrate that more investigation is needed on coffee water relations and plant water balance. The current knowledge is clearly not enough in order to improve coffee growth and yield with maximum efficiency in water use especially with the present tendency towards low water availability conditions as coffee cultivation increasingly takes place in areas of marginal climatic suitability.

2.2. CLIMATE IMPACTS ON COFFEE GROWTH AND YIELD

Drought episodes are considered to be the major environmental stress affecting coffee production. Drought stress affects the growth and the development of the plant in different ways over the coffee crop's phenological phases. Overall, severe drought stress disturbs the plant's metabolism with disruption of osmotic and ionic homeostasis and damage to cell membranes and proteins. In such conditions, the time factor is paramount for coffee crop survival and yield maintenance as drought stress usually
develops slowly and increases in intensity the longer it lasts (Da Matta and Ramalho 2006; Camargo 2010).

Plant water deficit is accompanied by loss of turgor and wilting, cessation of cell enlargement, closure of stomata, reduction in photosynthesis and movement of water and nutrients to various parts of the plant with interference to many basic metabolic processes. The maintenance of cell turgidity is crucial for plants. Turgidity loss may indirectly affect photosynthesis through reduction of stomatal conductance.

The drought condition is often aggravated by both high solar radiation and high temperature. A decrease in CO₂ assimilation in *Coffea arabica* at temperatures above 24^oC was reported by Nutman (1937). Another study demonstrated that continuous exposure to temperatures above 23^oC accelerates fruit development and ripening often accompanied with loss of quality (Camargo 2010). Moreover, variations on the size of coffee beans within and between years depend on the plant water status during the period of fruit expansion occurring between 10-17 weeks after blossoming. Drought may reduce the size of beans, the number of fruiting nodes per tree and the number of fruits per node depending on its severity and timing. Some fruit shedding that occurs during the period of rapid fruit swelling may be aggravated by drought. Prolonged drought in the initial stages of bud development may result in the atrophy of set fruit (Carr 2001).

Also, prolonged dry environmental conditions reduce shoot extension and the area of individual leaves. Early leaf senescence is also stimulated in drought conditions. But, despite leaf abscission being a mechanism to avoid transpiration when water is limiting, coffee plants may delay or reduce leaf shedding as a strategy to save resources that otherwise would be used in restoration of leaf area in the rainy season (Carr 2001). The avoidance of leaf shedding is also important to maintain crop yield but, high solar radiation incidence and temperature may result in absorption of much more energy than that used in the photosynthesis process and cause overheating of leaves which can reach 40°C in extreme cases. The excess of energy may potentially lead to reactive oxygen species production that can detrimentally oxidize photosynthetic pigments and proteins (Da Matta and Ramalho 2006). In response to such conditions, leaf water potential tends to decrease and a hydraulic signalization of the plant water status induces the biosynthesis of the stress hormone abscisic acid

(ABA) as a strategy for water loss control in the short-term. Therefore, stomatal conductance is reduced in order to maintain the plant water status.

Moreover, the relationship of the roots with shoots is also strongly affected by conditions of prolonged shortage of water. Reduction of shoot growth is one way that plants use to keep their water status stable and to reduce water loss to the atmosphere. As water flows from the higher to the lower water potential, root water potential decreases as a mechanism to maintain plant water status and avoid water flux from the root xylem to the soil solution by diminishing root hydraulic conductivity. At the same time the plant may invest in expansion of the root system to increase the soil volume explored and increase water uptake. Root system distribution and growth plays an important role in maintaining a favorable internal water status under water deficit conditions. Plants able to develop deeper root systems usually are more tolerant to drought. In dry conditions coffee roots may extend deeper and extract a greater quantity of soil water from deep layers. Studies on coffee root system characteristics indicated that the most efficient strategy to face dry conditions seems to be associated with a deeper root system rather than larger dry mass (Da Matta and Ramalho 2006).

Genotype selection in order to improve coffee yield in water-limited environments was found to be only partially successful due to the variability of rainfall and the polygenic nature of drought tolerance. A complementary approach involves the identification of traits that contribute to withstanding drought condition. Potential important traits may include water extraction efficiency, water use efficiency, hydraulic conductance, osmotic and elastic adjustment and modulation of leaf area. However, most of these traits are complex and poorly understood (Da Matta 2004).

There are a wide range of factors that influence coffee responses to drought and there is a need to improve the knowledge of coffee performance in sub-optimal environmental conditions which are characteristic of most marginal lands that have been used for coffee cultivation more recently, especially regarding the effects of climate changes.

2.3. WATER DYNAMICS IN COFFEE AGROFORESTRY

In agroforestry, water dynamics and use is affected by shade trees in different ways. For example, the trees' presence may influence water fluxes by changes in the microclimate under shade. In a study on coffee agroforestry in Mexico, the effect of Inga trees on buffering the maximum and minimum temperature and reducing the VPD in agroforestry compared to an open system was demonstrated. Daily maximum temperature reduction of $4 - 5^{\circ}$ C in the shade was reported (Barradas and Fanjul 1986).

Reduction of leaf temperature by shade trees also favors water use by coffee. Depending on the time of the day, season and leaf position within the coffee canopy, in coffee shaded with *Inga densiflora* leaf temperature was reported to be between 1 and 7^oC lower (Siles *et al.* 2009). A similar study demonstrated that coffee leaves without shade experienced air temperatures higher than ambient in the wet and dry season while in a shaded environment they were often lower (Muschler 1997). This temperature reduction is very important particularly in the tropics and marginal regions for coffee cultivation. Leaf temperature has a direct effect on the photosynthesis process for which the optimal rate is between 18^oC and 24^oC for arabica coffee (Da Matta 2004).

Comparison between coffee agroforestry and open systems showed that stomatal conductance was greater in shaded plants (Gutiérrez *et al.* 1994a; Vaast *et al.* 2005). The close correlation between stomatal conductance and net photosynthesis may result in carbon gain by the sunfleck effect in the shade (Fanjul *et al.* 1985). The spatial and temporal variability of net photosynthesis and transpiration was simulated (by using the MAESTRA model) in coffee agroforestry at a plot scale and showed that above-ground net primary productivity was not influenced by shade tree presence. Simulations showed greater light use efficiency in the shade compared to coffee in the full sun despite the reduction on photosynthetically active radiation under shade tree canopies (Charbonnier 2013).

The effect of different shade tree species on coffee water consumption was demonstrated in sub-optimal conditions in Costa Rica by Van Kanten and Vaast (2006). Coffee shaded by *Terminalia ivorensis* exhibited lower water consumption (538 mm p.a.) compared to that under *Eucalyptus deglupta* (689 mm p.a.) and under *Erythrina*

poeppigiana (702 mm p.a.). Similar values of coffee water consumption (703 mm p.a.) were found in Mexico in coffee agroforestry with *Inga leptoloba* as shade tree (Jiménez and Goldberg 1982). Also, water use of coffee and shade tree combined resulted in greater whole system water consumption when compared to the full sun (Van Kanten and Vaast 2006).

Water loss by soil surface evaporation was measured in coffee agroforestry at different shade levels in Mexico by Lin (2007). It was demonstrated that the higher the shade cover the lower was the soil evaporation. Lower water loss by soil evaporation means there is more water available for plant water use. However, apart from this study no other soil evaporation measurements have been found in coffee agroforestry.

Another factor to consider is that in agroforestry, the water available for under-storey coffee may be limited by interception of rainfall by shade tree canopies. Partitioning of rainfall into throughfall, stemflow and interception was studied in coffee shaded with *Inga densiflora* in Costa Rica by Siles *et al.* (2010). It was demonstrated that total rainfall interception was slightly increased in the shade (11.4% of rainfall) compared to an open system (9.6% of rainfall). This result was related to a three-fold greater leaf area index of coffee than that of shade tree. The influence of leaf area index and pruning on rainfall interception was demonstrated in coffee associated with *Inga* spp in Mexico by Jaramillo and Chaves (1998) and with *Erythrina poeppigiana* in coffee agroforestry in Costa Rica by Imbach *et al.* (1989). Studies on interception by shade tree canopies are very important especially in dry environments in which rainfall water is a critical factor for coffee growth and yield but investigations on this rainfall interception in coffee agroforestry are few.

2.4. COFFEE AGROFORESTRY WATER BALANCE

The water balance is one of the most important tools to assess water partitioning and competition for water in agroforestry systems. The water balance by an ecosystem over a given period of time may be estimated by the equation:

 $P = (Et + \Delta S + I + R_{off}) - R_{on}$

Where P=precipitation; Et=evapotranspiration; Δ S=change in soil water storage; I= Interception; R_{off}=run off; R_{on} = run on. Evapotranspiration is the water transpired by plants plus the soil evaporation. Interception comprises the rainfall intercepted by the foliage and evaporated back to the atmosphere. ΔS is the variation in soil water content in the soil volume studied. Run off and run on may occur either over the soil surface or as sub-surface flow within permeable upper horizons, and percolation (drainage) down to a water table is also included in the run off term, as it is a loss to the system (Rutter 1963).

The water balance partitioning in evapotranspiration, throughfall, drainage and run off in a coffee agroforestry watershed in Costa Rica demonstrated that superficial run off was 50% lower than in the monoculture. The experimentation and modelling across hydrological and physiological approaches indicated the great potential of coffee agroforestry to reduce superficial displacement of sediments and to provide hydrological environmental services by the efficient mechanism of aquifer recharge and discharge (Gomez-Delgado *et al.* 2010). However, studies on hydrological performance in coffee agroforestry either at a plot or at a basin scale are few.

Coffee agroforestry water flux assessment by measuring rainfall interception, simulation of transpiration and percolation were determined in *Coffea arabica* associated with *Cordia alliodora* and *Erythrina poeppigiana* in Costa Rica by Imbach *et al.* (1989). Great variability in interception in *Coffea arabica* with *Erythrina poeppigiana* was demonstrated and significant differences between interception in both systems was found to be due to the effect of management, particularly shade tree pruning. Transpiration amounted to 42.2% of rainfall in *Coffea arabica* with *Erythrina poeppigiana* while with *Cordia alliodora* it was 36.5% probably due to the pruning effect. It was demonstrated that water loss by leaching in coffee agroforestry was comparable to that of a natural forest (Imbach *et al.* 1989).

A coffee agroforestry water balance by measuring rainfall interception, transpiration, runoff and change in soil water content allowed simulations of different climate scenarios. A potentially high risk of water stress in agroforestry was predicted when the dry season was extended by 4-6 weeks. It was reported that the length of the dry season appeared far more important than the annual volume of rainfall (Cannavo *et al.* 2011).

Very little information is available on all components of coffee agroforestry water balance by using reliable and comparable methods.

2.5. COFFEE SHADE TREE CHARACTERISTICS AND PERFORMANCE

There may be great shade tree diversity when coffee is planted in forest after clearing of the understorey but more commonly coffee agroforestry consists of one, two or more shade tree species, often specifically planted, and includes timber, leguminous or fruit trees. For instance, 77 different coffee shade tree species were reported in El Salvador (Escalante and Somarriba 2001), 24 in Nicaragua and 107 in Mexico (López-Gómez *et al.* 2008). In Costa Rica diversification diminished with the increasing of farm size and coffee management (Llanderal and Somarriba 1999).

Leguminous trees are by far the most prevalent shade tree in Mexican and Central American coffee agroforestry. Leguminous trees such as *Albizia, Inga, Leucaena, Erythrina* and *Gliricidia* spp. have been preferred by farmers as coffee shade tree in part to also produce firewood. Leguminous trees have the advantage of generally being tolerant of periodic pruning in order to avoid light competition. Also, legumes may enhance soil nutrition through their nitrogen fixation capacity (Muschler 1997; Beer *et al.* 1998). Studies in coffee agroforestry with *Erythrina poeppigiana* as shade tree showed an input of 34 kg ha⁻¹ year⁻¹ of nitrogen fixation compared to full sun coffee (Babbar and Zak 1994).

Despite the beneficial impact of these legume species, in sub optimal conditions for coffee cultivation in the south of Costa Rica it was reported that coffee was affected by the seasonal phenology of *Erythrina poeppigiana* when the shade trees completely lost their leaves during the dry seasons which coincided with the period when the buffering effect of shade was needed most (Vaast *et al.* 2007). This phenological characteristic led farmers to prefer evergreen to deciduous as coffee shade trees in the Guanacaste region in Costa Rica. On the other hand, deciduous trees may be useful in areas with sub-optimal rainfall to provide mulch to maintain soil moisture and minimize water use when they are leafless in the dry periods (Albertin and Nair 2004).

Concerning timber trees, *Eucalyptus, Terminalia, Cordia, Cedrela* and *Alnus* are among the most common genera utilized in Central America coffee zones. But, despite the high potential for diversification of production and increasing incomes for farmers by timber production, especially in periods of lower coffee prices, only a small number of species of timber trees are used in coffee plantations (Haggar *et al.* 2001; Somarriba *et al.* 2004). Studies on coffee agroforestry based on timber tree species showed that

Eucalyptus deglupta and *Terminalia ivorensis* were ecologically and economically viable options for sub optimal environmental conditions in Costa Rica (Vaast *et al.* 2007). In Latin America, particularly in Costa Rica, farmers have introduced *Terminalia amazonia* as a coffee shade tree instead of *Terminalia ivorensis* due to its more compact crown and better quality of wood (Dzib *et al.* 2006). In optimal environmental conditions in the Central Valley of Costa Rica, coffee associated with *Eucalyptus deglupta* experienced lower soil water content and suffered drought stress during the five months dry season in contrast to coffee in association with regularly pruned *Erythrina poeppigiana* or full sun coffee (Jimenez and Alfaro 1999). But overall there is a lack of knowledge about timber tree characteristics and performance in agroforestry in different environmental conditions, particularly plant behavior which varies with other associated plants, arrangement, management and environment (Huxley 1996).

Despite coffee accounting for the largest part of the system water use, shade tree water consumption may lead to a greater total water use in agroforestry when compared to monoculture. Studies on transpiration of *Inga densiflora* as a coffee shade tree demonstrated that it accounted for 40-50% of the total water use in the system (Siles 2007). Estimation of annual transpiration of evergreen *Eucalyptus deglupta* resulted in 210 mm p.a. while *Terminalia ivorensis* was 519 mm and *Erythrina poeppigiana* was 195 mm, which represented 23%, 49% and 21% respectively of the total system water use. Although *Eucalyptus deglupta* was considered a better shade tree species than *Terminalia ivorensis* due to its constant and lower shade level during the year, results indicated that in the medium term shade of both timber tree species improved coffee growth and increased productivity under the sub optimal conditions of the study site (Van Kanten and Vaast 2006).

Knowledge about shade tree characteristics and potential adaptation of environmental conditions is crucial to avoid competitive relationships with coffee in resources use in agroforestry. Management skills by farmers such as selection of the shade tree species, planting density, tree thinning and frequency of canopy pruning are very important to ensure that the level of shade is neither too high for adequate coffee growth and productivity nor too low for effective protection of coffee plants against adverse climatic conditions (Vaast *et al.* 2007). However, little information is available about tropical timber tree species in coffee agroforestry in sub optimal environmental conditions for coffee cultivation. Although widely used in agroforestry in Central

America no information was found about the characteristics of *Simarouba glauca* and *Tabebuia rosea* as coffee shade trees.

2.6. BELOW GROUND RELATIONSHIPS IN COFFEE AGROFORESTRY

The coffee root system distribution is characterized as being more concentrated in the upper first 30 cm of the soil profile and then to decrease with depth (Nutman 1933a; Guiscafre-Arrilaga and Gomez 1938). Also, a study using tracer techniques at different distances from the trunk showed greatest root activity near to the trunk to around 30 cm distance from the stem (Huxley *et al.* 1974). Coffee root growth was found to be highly variable depending on seasonal pattern, environmental and soil condition (Nutman 1933b; Franco and Inforzato 1946; Morales and Beer 1998). The lateral spread of surface roots was constrained by the influence of neighboring coffee trees. More than one coffee tree in the same station improved deep axial root growth as far as to 3.5 m while single trees did not extend deeper (Clowes and Logan 1985). In agroforestry coffee roots may be influenced by the shade tree root system but in turn may also affect tree root distribution. *Coffea arabica* associated with *Eucalyptus deglupta* in Costa Rica showed distinct root system spatial arrangement and contrary to expectations, coffee fine roots were sufficiently competitive to displace shade tree fine roots (Schaller *et al.* 2003).

Below ground interactions in agroforestry may define complementarity or competition in water use. Studies on fine roots in a sub optimal growing zone in Costa Rica in which *Coffea arabica* was shaded by *Eucalyptus deglupta* or *Terminalia ivorensis* demonstrated that *Terminalia ivorensis* was potentially a stronger competitor with coffee compared to *Eucalyptus deglupta* (Van Kanten 2003; Vaast et al. 2007) while in optimal environmental condition for coffee cultivation *Eucalyptus deglupta* as coffee shade tree presented no competitive relationship (Schaller *et al.* 2003). However, studies on coffee root behavior in agroforestry and shade tree root system performance in different environmental conditions are also few.

In restrictive soil conditions root distribution and water movement may be inhibited while in agroforestry tree roots can favor crop root penetration and water uptake. Studies demonstrated that in a rigid soil matrix, the channels created by old dead tree root are utilized by crops to acquire water from deeper soil layers (Schroth 1995; Udawatta et al. 2006). Studies in coffee root development in distinct soil types showed little effect by a hard pan on coffee root growth (Nutman 1933b). Apart from his work, the impacts of a restrictive soil layer on coffee root penetration have been poorly investigated.

2.7. GAPS IN THE KNOWLEDGE OF COFFEE AGROFORESTRY WATER RELATIONS

Robust information on coffee agroforestry is still limited, especially in water use and dynamics. Field experiments on coffee water use are relatively few, given the extensive cultivation and value of the crop. In agroforestry, considering the species diversity, management heterogeneity and variations in environmental conditions, assessment of coffee water performance is difficult to extrapolate. Modeling could provide a holistic approach by connecting inter-dependent process. However, the available information in coffee agroforestry water fluxes influenced by different shade tree species and environmental condition to integrate successful process-based models is limited. Very little information was found on tropical timber tree species as coffee shade trees despite the potential additional income for farmers. There is a lack of information on root system behavior of coffee and associated shade trees along climatic and soil gradients in order to assess potential competition for water under the effects of climate change. Very few studies were found on coffee agroforestry evapotranspiration. Most of them are incomplete and do not accounts with direct measurement of the components, as conducted in this study.

CHAPTER III

ROOT DISTRIBUTION AND WATER USE IN COFFEE SHADED WITH *TABEBUIA ROSEA* BERTOL. AND *SIMAROUBA GLAUCA* DC. COMPARED TO FULL SUN COFFEE IN SUB-OPTIMAL ENVIRONMENTAL CONDITIONS ^a

Padovan, M.P.^{1,2,3,*}, Cortez, V.J.², Navarrete, L.F.², Navarrete, E.D.², Deffner, A.C.⁵, Centeno, L.G.⁴, Munguía, R.⁴, Barrios, M.², Vílchez-Mendoza, J.S.², Vega-Jarquín, C.⁴, Costa, A.N.¹, Brook, R.M.³, Rapidel, B.^{2,5}

¹ Instituto Capixaba de Pesquisa, Assistência Técnica e Extensão Rural (INCAPER), Afonso Sarlo, 160, Vitoria, Espirito Santo, Brasil.

² The Tropical Agricultural Research and Higher Education Center (CATIE), 7170, Turrialba, Costa Rica.

³ School of Environment, Natural Resources and Geography, College of Natural Sciences, Bangor University, Bangor, Gwynedd LL57 2UW, UK.

⁴ Universidad Nacional Agraria (UNA), Managua, Nicaragua.

⁵ CIRAD UMR SYSTEM (CIRAD-INRA-SupAgro), 2 Place Viala, 34060 Montpellier, Cedex, France.

*Author for correspondence (e-mail: mpadovan@catie.ac.cr)

Padovan, Vega-Jarquín, Costa, Brook and Rapidel designed research; Padovan, Cortez, Navarrete, Navarrete, Deffner, Centeno, Munguía and Barrios performed research; Padovan and Vílchez-Mendoza analyzed the data; Padovan, Brook and Rapidel wrote the paper.

ABSTRACT

Root niche differentiation for optimal exploitation of resources was found in an arabica coffee agroforestry system in Nicaragua. Rooting behavior was compared in both unshaded (FS) and shaded (AFS) coffee combined with two previously untested tropical timber species (deciduous *Tabebuia rosea* Bertol. and evergreen *Simarouba glauca* DC.). The predominant andisol presents a compacted soil layer (talpetate). Study was developed in sub-optimal environmental conditions for coffee cultivation

^a Published in Agroforestry Systems 89 (5): 857-868. 2015.

(455 m.a.s.l., mean 27°C/year, 1300 mm rainfall/year, six months dry season) in Nicaragua.

Twelve and five trenches 200 cm deep were dug in AFS and FS respectively. Roots per unit area were counted on two perpendicular soil faces. Volumetric water was measured continuously over two years by using 45 reflectometers in different soil layers. The talpetate varied greatly in depth, thickness and physical structure.

Coffee fine roots were more abundant than tree roots and were concentrated in the shallower strata (0 – 80 cm) whilst tree roots proliferated more below 100 cm. The *Simarouba glauca* root system was denser below 100 cm than *Tabebuia rosea* root system. There was no meaningful difference in coffee root counts in FS and under *Tabebuia rosea*, but coffee root counts were higher near *Simarouba glauca trees*.

2012 and 2014 had mild dry seasons and whole profile soil water content was similar in FS and AFS, but in the 2013 severe dry period volumetric water and water uptake were lower in AFS than in FS. This indicates that the normal advantage of greater soil exploration in AFS was cancelled presumably due to continued water uptake by deep rooting trees whereas the FS still had available water.

Key words: coffee agroforestry; niche differentiation; talpetate; compact soil layer; soil water content.

3.1. INTRODUCTION

Roots are the connection between shoot requirements and soil resources in terrestrial plants. While structural roots form the anchorage for the plant and storage of carbohydrates, fine roots perform the function of resource acquisition. Water and nutrients uptake are related to the root system distribution pattern which is in turn influenced by genotype, environmental interactions and management (Nair 1984; Fitter 1996).

In agroforestry systems where crop and tree growth occur simultaneously, the root system distribution pattern of the different components is a determinant of complementarity or competition in resource use. An ideotype of a shade tree would be that it should possess a deep root system in order to take up water and nutrients that are not available for crops (Sanchez 1995; Cannell *et al.* 1996). In reality most tree species have dense root systems in the topsoil where nutrients are concentrated

(Schroth 1995). A global study of tropical tree rooting distributions showed that on average 26% of roots are in the upper 10 cm of the soil profile, 60% down to 30 cm and 78% in the top 50 cm (Jackson *et al.* 1996).

Comparison between pure tree stands and agroforestry showed that in the monocrop stands of hybrid walnut (*Juglans nigra* x *regia*) and poplar (*Populus euramericana*) fine root density decreased with depth while in the intercropped stand tree fine root density was displaced to deep soil layers (Mulia and Dupraz 2006). Other study demonstrated that maize confined the root system of apple tree (*Malus domesticus*) laterally and induced a greater vertical root development in intercropping compared to monoculture apple tree growth in Nebrasca (Yocum 1937).

Below-ground competition may be avoided by both selection of shade trees with appropriate rooting architecture and tree root management. Root pruning of trees provides a powerful management tool for manipulating competition although the interruption of bi-directional flow of water from deep soil horizons to lateral roots at the top layer and vice-versa may be a disadvantage (Ong *et al.* 2007). Root bi-directional flow may benefit shallow-rooted crops growing in drying soils if adjacent trees can provide water from depth by accessing groundwater and therefore, the coexistence between tree and crops may be facilitated (Roupsard *et al.* 1999; Smith *et al.* 2004).

Although the coffee root system is characterized by its concentration in the upper soil layers (Nutman 1934; Bull 1963; Huxley *et al.* 1974), coffee root extension and distribution are also highly influenced by environmental and soil conditions. In dry sites, coffee root distribution is skewed towards deeper soil layers, and this distribution is related to higher drought tolerance (Ramos and Carvalho 1997; Da Matta and Ramalho 2006).

In optimal conditions for coffee cultivation in Costa Rica, no competition for water was found when shaded with *Eucalyptus deglupta* (Schaller *et al.* 2003) or *Inga densiflora* (Siles *et al.* 2009). However, reports on coffee and tropical timber tree interactions in sub-optimal environmental conditions are few. This paper discusses soil exploration by coffee and shade tree roots in a mature agroforestry experiment comparing unshaded monoculture coffee (FS, "full sun") with adjacent agroforestry (AFS) plots, comprising coffee cultivated under mixed shade of two tree species.

The research described in this paper took place in a coffee agroforestry system in a location with a compacted soil horizon. Other work has demonstrated that compacted soil layers may limit water movement, root penetration and therefore, plant water uptake (Bennie 1996; Bengough *et al.* 2011). In such soil conditions, tree roots play an important role in facilitating water flow and crop root penetration (Nair 1984; Van Noordwijk *et al.* 1996). Voids left by partially decomposed tree roots may be utilized by crops to acquire water from deeper soil layers (Van Noordwijk *et al.* 1991). Moreover, root channels with live or decaying roots can increase saturated hydraulic conductivity by serving as conduits for preferential flow (Johnson-Maynard *et al.* 2002; Benegas *et al.* 2014).

For this study the hypotheses tested were that under soil conditions characterized by a compact layer, trees roots penetrate through the hard-pan and so facilitate coffee water uptake throughout the soil profile; that coffee and tree roots explore different spatial niches; and that evergreen and deciduous tropical timber tree species exhibit different root system distributions.

The study was conducted in sub-optimal conditions for arabica coffee cultivation, typified by a dry season with six months with rainfall < 50 mm, the presence of a compacted soil layer at intermediate and variable depth, and root exploration ultimately limited by a mineral compact layer at approximately 2 m depth. This experiment site is representative of the whole Carazo coffee growing region in Nicaragua. Location of such studies in agriculturally difficult environments is important, as these are the conditions under which many farmers have to operate.

3.2. MATERIALS AND METHODS

3.2.1. Site description

This study was carried out from November 2011 to May 2014 in an experiment located at Jardin Botánico, Masatepe, Department of Masaya, southern Nicaragua (11[°] 53' 54" N, 86[°] 08' 56" W) at a long term research site managed by the Centro Agronómico Tropical de Investigación y Enseñanza (CATIE), jointly with the Universidad Nacional Agraria (UNA), Federación Cooperativas de Ahorro y Crédito (CENECOOP-FEDECARUNA) and Instituto Nicaraguense de Tecnologia Agropecuaria (INTA). The experiment was established in 2000 and was described by Haggar *et al.* (2011).

The site is located at 455 m.a.s.l. which is considered to be rather low for arabica coffee cultivation. The mean annual temperature is 27^oC with only slight seasonal variation and mean annual rainfall is 1470 mm. From 85% to 97% of the total annual precipitation falls over the wet season (May-November) while a pronounced seasonal drought occurs from late November to mid-May (Vogel and Acuña Espinales 1995). Annual rainfall recorded was 968 mm and 1312 mm in 2012 and 2013 respectively.

In the study area the topography is flat and soils are predominantly characterized as andisols, which are derived from volcanic ejecta. These soils are typically deep, well drained and have high organic matter content, low bulk density, high allophane content and consequently a high phosphorus fixation capacity, high amorphous mineral content and high water retention capacity (FAO, 2001).

On this particular study site, however, soils are characterized by the presence of a hardened layer locally known as talpetate (Fig 3.1). Such layers occur in about 15% of the Nicaragua Pacific region. Its properties reflect both geologic and soil-forming processes and can be extremely variable. It ranges from soft, weathered material containing some harder rock fragments to a fairly continuous hard layer with rock-like properties. The texture varies from fine to sandy, with the latter often appearing stratified (Vogel and Acuña Espinales 1995).



Fig 3.1. Talpetate layer in the soil profile with roots inside the fractures

At Masatepe, the talpetate layer is characterized by high aluminium and silicon contents which are retained in allophones after irreversible dehydrating to become a cemented material. This is usually associated with limitations to water movement and root growth. However, weaknesses or fractures that occur in the talpetate layer allow roots to access the deeper soil horizons (Vogel and Acuña Espinales 1995).

Besides the talpetate, in Masatepe, the soil profile consists of three other main layers distinguished by color: brown (uppermost layer), reddish (usually above the talpetate) and a yellowish, granular layer, under the talpetate. At 1.60 - 2.0 m depth, there is a dark granular compact layer, without organic content, where neither roots nor breaks were observed, so we did not explore below 2.10 m.

The bulk density varied from 0.62 g cm⁻³ (S.E.= 0.026) to 0.65 g cm⁻³ (S.E.= 0.039) on average in the shade and full sun plots, respectively, while in the talpetate layer it reached 0.72 g cm⁻³ in both systems. The mean organic matter content of the soil profile was 3.38% (S.E.=1.05) and 3.82% (S.E.=2.0) in the shade and full sun plots respectively. In the uppermost, brown layer organic matter content was found to be 8.75% (S.E.=0.10) in the shade system and 8.22% (S.E.=0.25) in the full sun, and decreased with depth. The pH (in H₂O) did not differ in both systems and ranged between 5.70 and 5.80 which is considered adequate for coffee cultivation. Management includes fertilization with 37.3 kg ha⁻¹ of N, 48.8 kg ha⁻¹ of P and 27.6 kg ha⁻¹ of K as NPK per year. In addition 34.4 kg ha⁻¹ of N as urea and 12 kg ha⁻¹ of K as KCl are applied each year.

3.2.2. Experimental design

The experimental design consists of a full sun monocrop coffee (FS) plot (50.5 m x 28.5 m) as a pseudo replication and two adjacents coffee agroforestry system (AFS) plots (80.0 m x 40.0 m and 60.0 m x 23.7 m) in which data were collected in 2011 and 2013. For the current study, pseudo-replication occurred within the full sun treatment plot due to the poor set of coffee plants and problems with small floods in the real replication. However, it was considered appropriate for a process-based study such as this one. The *Coffea arabica*, variety "Pacas" was planted in 2000, at a density of 4000 plants ha⁻¹. Coffee spacing was 2 m between rows and 1.25 m between plants. Coffee plants were pruned selectively in accordance with standard agronomic practice. In the coffee agroforestry system plot *Coffea arabica* is associated with *Simarouba glauca* DC. (Simaroubaceae) and *Tabebuia rosea* (Bertol.) DC. (Bignoniaceae) planted as shade

trees. The initial density of the mixed tree cover was 667 trees per ha with alternating rows of both species, but tree density was reduced over time, since 2000, in order to achieve an agronomically appropriate shade level. The mean density of *Tabebuia rosea* was 113 trees ha⁻¹ and that of *Simarouba glauca* was 75 trees ha⁻¹ over the period of the study (Fig 3.2). The experimental design was based on three hypothesized levels of competition for water in AFS compared to full sun coffee. The split-split plot design consisted of five incomplete blocks (whole plot) with four split plots (treatments): coffee monoculture, and in the coffee agroforestry system, coffee near *Simarouba glauca* trees, coffee near *Tabebuia rosea* trees and shaded coffee as far as possible from both tree species (around 4 m); and two split-split plots (soil profiles) for each treatment.



Fig 3.2. Schematic representation of the experimental design of (one quarter of) the shade plot with coffee, *Simarouba glauca* and *Tabebuia rosea* spatial distribution pattern over the period of the experiment.

3.2.3. Tree species studied

Simarouba glauca DC. is native to tropical and sub-tropical regions of Central America, Mexico and the Caribbean. It is an evergreen timber tree that grow successfully over a range of soil types and their deep root system enables efficient access to water year round (Brodribb *et al.* 2003). In the study area *Simarouba glauca* trees at age of 13 years reached in the average 17 m height with 25.5 cm DBH.

Tabebuia rosea Bertol. is native to Mexico, Central America, Venezuela and coastal Ecuador. Trees grow in a great variety of habitats, but prefer regions with high annual rainfall (between 1200 and 2500 mm). It has a deep root system and tolerates occasional waterlogging. It is a deciduous tree that sheds its leaves in the late dry season (CABI, 2013). In the study area *Tabebuia rosea* trees stood bare for about two months (April and May) in 2012 and for four months (from the end of February to the end of June) in the more severe dry season of 2013. By the end of the experiment trees reached 15.5 m height and 28.7 cm DBH on average in the experimental area.

3.2.4. Fine root distribution

Fine root distribution of trees and coffee plants was recorded by the root impact counting method (Tardieu 1988; Van Noordwijk et al. 2000). This method is based on the traditional profile wall method proposed by Thiel in 1892 that has been used with modifications by several authors (Bohm 1979). A root impact was considered to be any intersection of a root with the exposed vertical soil profile under study (Laclau et al. 2001). Fine roots were defined as root with diameters ≤ 2 mm. Three and six trenches were dug in the full sun and shade coffee respectively in November 2011 while another two and six trenches were dug in unshaded and shaded coffee in November 2013. A total of five trenches in FS and twelve trenches in AFS were analyzed. Both trench faces were used for recording. No significant effects of time of recording were found on root distribution (p>0.05), so a mature, stable root system was assumed for analytical purposes. Ten profiles were analyzed in the full sun coffee plot and twenty four profiles in coffee agroforestry plots: eight profiles were 150 cm distant from Tabebuia rosea stems; eight profiles 150 cm distant from Simarouba glauca stems and the other eight as far as possible from any tree, but within the agroforestry plots. Trenches were located between the coffee rows and about 50 cm away and perpendicular from the coffee stems. Each trench was 200 cm deep and 60 cm wide. The soil profile was divided into 10 x 10 cm grid cells and the number of roots in each

grid cell was counted. We used a small knife to gently remove the surrounding soil and expose the root ends. Roots of *Coffea arabica, Tabebuia rosea* and *Simarouba glauca* were distinguished from each other on the basis of color, smell and rigidity characteristics. The *Tabebuia rosea* roots were flexible and whitish while *Simarouba glauca* roots were rigid, easily broken and yellowish with more strong smell compared to *Tabebuia rosea* roots. The *Coffea arabica* roots had a clearer color in comparison to the tree roots, they were brittle and with a smell very characteristic.

3.2.5. Soil water content

We measured the changes in the soil water content during the two and a half years of our experiment: when the first trenches were dug, a total of 45 Time Domain Reflectometers (Campbell Scientific Inc. CS616, subsequently called TDR probes) were inserted into the soil of one of the trench walls, horizontally in the middle of each of the four soil layers reported above; thus, depths of insertion were variable. Once the probes were inserted the trenches were fulfilled. Horizontal location of each TDR soil profile was established so that it was representative of a reproducible cell (mid distance from the center of the coffee row and the center of the interrow). TDR probes were connected to dataloggers (CR 1000 with multiplexer, Campbell Scientific Inc.), scanned every minute and data were stored every 30 minutes during the whole experiment. Minimal disturbance soil samples of sufficient dimensions to insert a TDR probe were also extracted from each soil layer (two samples per layer). Progressive drying was measured simultaneously by TDR and by weighing, so that a calibration equation was available for each soil layer, following a protocol adapted from Udawatta et al. (2011). The volumetric soil water contents were then multiplied by the thickness of the soil layers in which each TDR probe was inserted to calculate the soil water content of each layer. All soil layer water contents were added to get the soil water content of the whole profile at each time step.

We then calculated water uptake for each trench per period of time. As a first step, the deepest TDR in each trench was examined. For the purposes of this paper we only selected periods when these deep TDR showed a consistent decrease or a stable signal throughout the period, indicating that water was not diffusing down from upper layers. We were thus able to discard periods when drainage out of the observed soil profile could have occurred. All periods with heavy rains had thus to be discarded (whole wet season in 2013, and almost all wet season in 2012, with the exception of a one month

period in July-August when rainfall events were very low). The data retained covered 53% of the two-year experiment, in nine continuous periods ranging 25-60 days. Soil surface run-off and run-on was also discounted since no run-off was observed in the study area because the soil surface was horizontal. Moreover, andisols are renowned for their high infiltration rates.

We could then estimate evapotranspiration from the coffee systems from soil water content change: providing the assumptions of lack of drainage were accepted the water balance equation could be simplified to:

S1 - S2 + R = Et

Where: S_1 is the soil water stock in the soil profile at the beginning of the period (mm); S_2 is the soil water stock in the soil profile at the end of the period (mm); R is the accumulated rainfall during the period (mm); E_t is the evapotranspiration by the soil-plant system accumulated during the period (mm). Final data were expressed in mm.day⁻¹, calculated as the ratio between total evapotranspiration and the total number of days of the period.

Data analysis

General and mixed linear models were performed taking into account the hierarchical model in the split-split plot design (Pinheiro and Bates 2009). Roots were analyzed by variance analysis to compare root impacts between both data collection periods. Root spatial distribution was analyzed by co-variance analysis regarding the depth as co-variable. As residuals did not conform to a normal distribution, data were transformed into natural log (1+root impact). The model was also applied to analyze the effect of the talpetate on coffee root distribution by contrast analysis. The model took into account the logarithm of coffee root impacts in the treatments and the presence or absence of talpetate in replications by depth. Differences in soil water content between both the full sun and agroforestry system in the three dry periods studied were tested. The temporal series for both treatments were assumed in the model as a first order autoregressive function. Model assumptions were evaluated by residuals and predictors plots. Heterogeneous variances were modelled for treatments and years. All analyses were performed by using InfoStat software 2014 (Di Rienzo *et al.* 2014).

3.3. RESULTS AND DISCUSSION

3.3.1. Coffee and tree fine root distribution

Coffee root growth in deep layers was not significantly enhanced in the presence of tree roots (p>0.05), fine root growth being similar in full sun and in shaded coffee systems. Coffee roots reached 150 cm and 170 cm depths in the full sun and in agroforestry system respectively which suggest only a weak influence of tree roots on coffee root depth penetration (Fig 3.3).



Fig 3.3. Coffee root impacts dm⁻² in the soil profile in full sun coffee (FS) and coffee agroforestry (AFS) from the mean of ten soil profiles in FS plot and 24 in AFS. Errors bars represent the standard error of the mean.

Most roots were concentrated in the upper 30 cm of the soil profile where 56.9% coffee fine root impacts in the full sun and 50.6% in the shade occurred. However, coffee root distribution did differ between shade tree species. There were more coffee root impacts on average in the soil profile near *Simarouba glauca* than near *Tabebuia rosea* trees (p=0.001) (Fig 3.4). In the uppermost 30 cm, coffee root counts were approximately an order of magnitude greater than maximum tree root counts. In contrast to the coffee, shade tree root density was greater in deep soil layers than close to the soil surface, exhibiting niche differentiation between them and the shallower coffee roots (Fig 3.5).



Fig 3.4. Coffee root impacts dm⁻² in soil profiles near *Tabebuia rosea* and *Simarouba glauca* trees from the mean of eight soil profiles near each tree. Errors bars represent the standard error of the mean.



Fig 3.5. Niche differentiation of fine coffee roots and fine tree roots in the agroforestry system from the mean of 24 soil profiles in shade plots. Pictures represent the whole grids where roots were counted in 10cm x 10cm cells.

Roots of both tree species reached the dark, granular compacted layer at around 200 cm depth but penetrated no further. Tree root distribution pattern varied with species. In the deeper layers although both tree species showed higher root density below 110 cm depth, the *Simarouba glauca* root system was denser and more concentrated while *Tabebuia rosea* roots did not display such a distinct zonation in the soil profile (p<0.0001). Maximum fine root density of *Simarouba glauca* occurred at 165 cm depth and *Tabebuia rosea* exhibited a less distinct peak at 115 cm depth (Fig 3.6).



Fig 3.6. *Simarouba glauca* and *Tabebuia rosea* fine root distribution from the mean of 24 soil profiles in shade plots. Errors bars represent the standard error of the mean.

3.3.2. The effect of talpetate on coffee roots distribution

Field observations demonstrated that the depth of the talpetate layer varied between 20-170 cm and 20-150 cm in the full sun and in the shade plots respectively. The talpetate layer width was also highly variable ranging from 20 to 120 cm in the full sun and 30 to 80 cm in the shaded plots, although these differences were not significant between both systems. Also, there were sites where the tapetate layer was dense and compact and others where it was fractured at different levels regardless of the treatment. The bulk density and the organic matter content of the talpetate layer were not significantly different between the full sun and agroforestry systems or between the treatments (p>0.05). The bulk density ranged from 0.54 g cm⁻³ to 0.85 g cm⁻³

while the organic matter varied from 0.5 % to 1.66 %. The presence of the talpetate layer had a significant effect (F=6.32; p=0.02) on coffee root growth. The contrast analysis of presence and absence of talpete on coffee root distribution by depth showed that coffee root impacts diminished in the presence of talpetate (0.55; S.E=0.03) compared to in the absence of talpetate (0.67; S.E.=0.04) regardless of the treatment. Vogel and Acuña Espinales (1995) also found limitations on root growth and agriculture development in soils with talpetate in comparison to soils without talpetate layer in Nicaragua.

3.3.3. Soil water content

Soil water content was a consequence of the rainfall inputs and its interactions with water uptake by plants and movement within the soil. Rainfall in 2011 was normal for the region, with relatively late rains in November (data not recorded locally). Total rainfall in the 2012 dry season was 57.2 mm with a maximum daily rainfall event of 16.8 mm. In 2012 (Fig 3.7), the rainfall in September to November was low, and the following dry season lasted almost six months, with only 23 mm rainfall overall.



Fig 3.7. Monthly rainfall distribution from climate station, Masatepe, Nicaragua; February 2012 to May 2014.

The corresponding rainfall in late 2013 was much more abundant. Comparison between soil water content of both systems showed significant differences (F = 5.98; p = 0.0148) on the soil water content in the three periods studied (Fig 3.8).



Fig 3.8. A comparison of soil water content in the whole soil profile (2000 mm) from the mean of three trenches in the full sun and six trenches in agroforestry in the (a) 2012, (b) 2013 and (c) 2014 dry periods.

The full sun coffee showed greater mean water content (561 mm; S.E.=20.2) over the whole soil profile than the agroforestry system (484 mm; S.E.=26.3) in the period of study. Comparison of temporal differences in soil water content of both systems showed that in 2012 soil water content was greater (584 mm; S.E.= 27.1) compared to 2013 (468 mm; S.E.=23.2). In 2014 slopes were similar to 2012 (516 mm; S.E.=33.9) indicating the same rates of water uptake (Fig 3.8a and 3.8c). However, in 2013 in AFS the slope was close to zero due to lower available water. The lowest values of water content during the observation period were registered at the end of the 2013 dry season in AFS when it reached 453 mm while in 2012 it was 572 mm and in 2014 506 mm (Fig 3.8b).

3.3.4. Soil water uptake

The comparison of daily water uptakes by both coffee systems from the soil profile is presented in Fig 3.9. We used only periods when drainage could be discounted. Soil water content data came from a total of nine soil profiles: three in the full sun coffee and six in the coffee agroforestry system. In the agroforestry system we found greater water consumption than in full sun coffee for most of the periods presented. However, at the end of the severe 2013 dry season (p=0.07) this condition seemed to be reversed. Both systems showed a very low rate of water uptake at the end, but data indicated that the coffee agroforestry system had lower soil available water than the full sun coffee (Fig 3.9). This indicates that in 'normal' dry seasons, water extraction rate in AFS was similar to FS and due to deeper tree root exploration, water content was lower. However, in the 2013 dry period, roots in AFS had extracted water to the point that soil water content was approaching a steady state, while there was still enough water in the FS soil to permit water uptake, albeit at reduced rate, to continue. Thus, only in very dry periods did the presence of tree roots exert competition to the point where coffee was no longer able to extract water.



Fig 3.9. Comparison of soil water uptake by the two coffee cropping systems from the whole soil profile 2012 to early 2014. Horizontal bars represent the period lengths under consideration. Errors bars represent standard error of the mean. Probabilities for identical values figure above each comparison (T-test).

3.4. CONCLUSIONS

Root distribution may define competition or complementarity in resources use in agroforestry systems. In this investigation shallower coffee roots and deeper tree root distribution patterns suggest complementarity in soil water use throughout the soil profile. This root niche differentiation is desirable in agroforestry but is not often demonstrated.

However, coffee and tree root system distribution varied with tree species. *Simarouba glauca* presented denser root system in deep soil layers compared to *Tabebuia rosea* suggesting a facilitative relationship upon coffee root growth. Coffee and *Simarouba glauca* exhibited a clear niche differentiation whereby soil water and nutrients are likely to be extracted from different strata. The results reinforced the ecological hypothesis (Cannell *et al.* 1996) in which tree deep root system may improve soil resources use that are not available for crops.

We demonstrated that although coffee roots crossed the talpetate zone, there was some influence of this hard layer on restricting coffee root growth regardless of the treatment. Soil water content was higher in the full sun coffee rather than the agroforestry in the whole soil profile in the dry seasons of the investigation period. At the beginning of both 2013 and 2014 dry seasons, the agroforestry coffee system was able to take up water at a greater rate from the soil profile, explained by a greater requirement by trees and a better exploration of the whole profile by roots.

But by the end of a severe dry season, as observed in 2013, the advantage of the better soil exploration was cancelled: it seems that soil water became almost exhausted during the dry season. By comparison, full sun coffee, which used less water at the beginning of the 2013 dry season, still had soil water left in the deeper layer when the dry season extended. This deep soil water was progressively used by the system, allowing a greater water use when compared to the agroforestry coffee.

Although more data would be required to distinguish the water uptake from coffee, both species of trees or the soil evaporation, we can hypothesize that there was a competition between trees and coffee for soil water uptake in atypically dry periods.

3.5. REFERENCES

- Benegas, L.; Ilstedt, U.; Roupsard, O.; Jones, J.; Malmer, A. 2014. Effects of trees on infiltrability and preferential flow in two contrasting agroecosystems in Central America. Agriculture, Ecosystems & Environment 183: 185-196.
- Bengough, A.G.; McKenzie, B.M.; Hallett, P.D.; Valentine, T.A. 2011. Root elongation, water stress, and mechanical impedance: a review of limiting stresses and beneficial root tip traits. Journal of Experimental Botany 62(1): 59-68.
- Bennie, A.T.P. 1996. Growth and mechanical impedance. 2nd ed. Waisel, Y.; Eshel, A.;Kafkafi, U. eds. New York. Basel. Hong Kong, Marcel Dekker, Inc. 453-470 p.(Plant roots. The hidden half)

Bohm, W. 1979. Methods of studying root systems. Billings, W.D.; Golley, F.; Lange, O.L.; Olson, J.S. eds. Berlin Heidelberg New York, Springer-Verlag. 188 p. (Ecological Studies)

- Brodribb, T.J.; Holbrook, N.M.; Edwards, E.J.; Gutiérrez, M.V. 2003. Relations between stomatal closure, leaf turgor and xylem vulnerability in eight tropical dry forest trees. Plant, Cell and Environment (26): 443-450.
- Bull, R.A. 1963. Studies on the effect of mulch and irrigation on root and stem development in Coffea arabica L. Turrialba 13(2): 96-115.
- Cannell, M.G.R.; Van Noordwijk, M.; Ong, C.K. 1996. The central agroforestry hypothesis: the trees must acquire resources that the crop would not otherwise acquire. Agroforestry Systems 34: 27-31.
- Da Matta, F.M.; Ramalho, I.D.C. 2006. Impacts of drought and temperature stress on coffee physiology and production: a review. Plant physiology 18(1): 55-81.
- Di Rienzo, J.; Casanoves, F.; Balzarini, M.; Gonzalez, L.; Tablada, M.; Robledo, C. 2014. InfoStat Manual. Argentina., Universidad Nacional de Córdoba. 336 p.
- Fitter, A. 1996. Characteristics and functions of root systems. In. 1996. Root plants. The hidden half. Tel Aviv. Israel, p. 1-20.
- Haggar, J.; Barrios, M.; Bolaños, M.; Merlo, M.; Moraga, P.; Munguia, R.; Ponce, A.; Romero, S.; Soto, G.; Staver, C.; de M. F. Virginio, E. 2011. Coffee agroecosystem performance under full sun, shade, conventional and organic management regimes in Central America. Agroforestry Systems 82(3): 285-301.
- Huxley, P.A.; Patel, R.Z.; Kabaara, A.M.; Michell, H.W. 1974. Tracer studies with P on the distribution of functional roots of arabica coffee in Kenya. Annals of Applied Biology 77: 159-180.
- Jackson, R.B.; Canadell, J.E., J.R.; Mooney, H.A.S., O.E.; Schulze, E.D. 1996. A global analysis of root distributions for terrestrial biomes Oecologia 108: 389-411.
- Johnson-Maynard, J.L.; Graham, R.C.; Wu, L.; Shouse, P.J. 2002. Modification of soil structural and hydraulic properties after 50 years of imposed chaparral and pine vegetation. Geoderma 110: 227-240.
- Laclau, J.P.; Arnaud, M.; Bouillet, J.P.; Ranger, J. 2001. Spatial distribution of Eucalyptus roots in a deep sand soil in the Congo: relationships with the ability of the stand to take up water and nutrients. Tree physiology 21: 129-136.

- Mulia, R.; Dupraz, C. 2006. Unusual fine root distributions of two deciduous tree species in southern France: What consequences for modelling of tree root dynamics? Plant and Soil (281): 71-85.
- Nair, P.K.R. 1984. Role of trees in soil productivity and conservation. In: Soil productivity aspects of agro-forestry. The International Council for Research in Agro-forestry. Nairobi. (Chapter 3): 29-49.
- Nutman, F. 1934. The root-system of Coffea arabica. Part III. The spacial distribution of the absorbing area of the root. Empire Journal of Experimental Agriculture 2: 293-302.
- Ong, C.K.; Anyango, S.; Muthuri, C.W.; Black, C.R. 2007. Water use and water productivity of agroforestry systems in the semi-arid tropics. Annals of Arid Zone 46(3-4): 255-284.
- Pinheiro, J.C.; Bates, D.M. 2009. Mixed-Effects Models in S and S-PLUS. 2nd edition ed. New York, Springer. 530 p. (Statistics and computing)
- Ramos, L.C.S.; Carvalho, A. 1997. Shoot and root evaluations on seedlings from Coffea genotypes. Bragantia 56(1): 59-68.
- Roupsard, O.; Ferhi, A.; Granier, A.; Pallo, F.; Deppomier, D.; Mallet, B.; Jolly, H.I.; Dreyer, E. 1999. Reverse Phenology and dry season water uptake by Faidherbia albida (Del.) A Chev. in agroforestry parkland of Sudanese West Africa. Functional Ecology 13: 460-472.
- Sanchez, P.A. 1995. Science in agroforestry. Agroforestry Systems 30: 5-55.
- Schaller, M.; Schroth, G.; Beer, J.; Jimenez, F. 2003. Species and sites characteristics that permit the association of fast-growing trees with crops: the case of Eucalyptus deglupta as coffee shade in Costa Rica. Forest Ecology and Management 175: 205-215.
- Schroth, G. 1995. Tree root system characteristics as a criteria for species selection and systems design in agroforestry. Agroforestry Systems 30: 125-143.
- Siles, P.; Harmand, J.M.; Vaast, P. 2009. Effects of Inga densiflora on the microclimate of coffee (Coffea arabica L.) and overall biomass under optimal growing conditions in Costa Rica. Agroforestry Systems 78(3): 269-286.

- Smith, M.; Burgess, S.S.O.; Suprayogo, D.; Lusiana, B.; Widianto, M. 2004. Uptake, partitioning and redistribution of water by roots in mixed species agroecosystems. In Van Noordwijk, M. ed. 2004. Below-ground interactions in tropical agroecosystems : concepts and models with multiple plant components. Cambridge, MA, USA, CABI Publishing. p. 157-170.
- Tardieu, F. 1988. Analysis of the spatial variability of maize root density I. Effect of wheel compaction on the spatial arrangement of roots. Plant and Soil (107): 259-266.
- Udawatta, R.P.; Anderson, S.H.; Motavalli, P.P.; Garrett, H.E. 2011. Calibration of a water content reflectometer and soil water dynamics for an agroforestry practice. Agroforestry Systems 82: 61-75.
- Van Noordwijk, M.; Widianto, M.; Heinen, M.; Hairah, K. 1991. Old tree root channels on acid soils in the humid tropics: Important for crop root penetration, water infiltration and nitrogen management. Plant and Soil 134: 37-44.
- Van Noordwijk, M.; Lawson, G.; Soumaré, A.; Groot, J.J.R.; Hairiah, K. 1996. Root distribution of trees and crops: competition and/or complementarity. In Huxley, C.K.O.a.P. ed. 1996. Tree-crop interactions. CAB International. p. 319-359.
- Van Noordwijk, M.; Brouwer, G.; Meijboom, F.; Do Rosario, G.; Oliveira, M.; Bengough,
 A.G. 2000. Trench profile techniques and core break methods. In Smit, A.L.;
 Bengough, A.G.; Engels, C.; Van Noordwijk, M.; Pellerin, S.; Van der Geijn, S.C. eds.
 2000. Root Methods, a Handbook. Berlin, Springer Verlag. p. 211-233.
- Vogel, A.W.; Acuña Espinales, E. 1995. Reference soils of the Pacific Coastal Plain with a hardpan (Talpetate). International Soil Reference and Information Centre and Universidad Nacional Agraria. 21 p. (Soil brief Nicaragua) (3).
- Yocum, W.W. 1937. Root development of young delicious apple trees as affected by soils and by cultural treatments. University of Nebraska Research Bulletin (95): 1-55.

CHAPTER IV

RAINFALL INTERCEPTION, STEMFLOW AND THROUGHFALL IN COFFEE SHADED WITH *TABEBUIA ROSEA* BERTOL. AND *SIMAROUBA GLAUCA* DC. COMPARED TO FULL SUN COFFEE

4.1. INTRODUCTION

Trees associated with crops may improve water use by diminishing water loss from soil evaporation (Jackson and Wallace 1999), by enhancing soil infiltration capacity and reducing runoff (Rao *et al.* 1998; Lunka and Patil 2016), by increasing retention of water in soil via higher organic matter content due to tree litter inputs (Beer 1987) and also by increasing water uptake from deep soil layers that otherwise could be lost by leakage (Cannell *et al.* 1996). However, trees may use a significant proportion of rainfall (Ong and Black 1994) and additionally may intercept and modify raindrop trajectories reducing the water that reaches the ground with impacts on crop growth and yield (Calder 1998).

Rainfall interception is recognized as an important component of water balance in agroforestry systems. The water inputs through gross rainfall may pass through the foliage and crop branches (throughfall) and add to soil water content; or flow down the leaves, branches and trunk to reach the ground (stemflow) or water may be kept on the leaf surface until it evaporates (interception) and so never reaches the soil. Because of the difficulties of directly measuring water loss by canopy interception it is estimated from the difference between the incident gross rainfall and the sum of throughfall and stemflow (Horton 1919; Rutter 1963). The rainfall interception process is influenced by both the climatic condition (rainfall intensity, wind speed, air temperature, solar radiation) and tree canopy parameters such as architecture and leaf area as well as the leaf and trunk surface characteristics (Crockford and Richardson 1990).

Throughfall has been shown to be the most important component of water input. Studies on rainfall interception by five tropical timber tree species in Panamá demonstrated that the average throughfall varied from 73.1% to 87.7% of the gross rainfall (Park and Cameron 2008). The influence of tree species on throughfall was also demonstrated in the Central Valley of California where values for pear and oak trees were 77% and 58% of the gross precipitation, respectively (Xiao *et al.* 2000). Also, it

43

was found that throughfall rates increased with the distance from the tree trunks (Eschner 1967). This same general pattern was reported by different authors and is likely to be due to the typically greater canopy density near the stem. It could also be due to crown architecture whereby branches nearer the stem facilitate greater stem flow as a percentage of incident rainfall. Throughfall rate has been documented to closely match the rainfall pattern of the precipitation event (Rutter *et al.* 1971). Intense precipitation events may further increase the probability of branch drip and decrease stemflow volumes by overloading preferential flow paths on tree trunks, therefore enhancing throughfall (Crockford and Richardson 1990).

Overall, stemflow contributes little to precipitation reaching the ground when compared with throughfall. In the tropical environment, studies showed stemflow to be around 1% of gross rainfall and consequently it has been considered to be negligible (Marin *et al.* 2000). Nevertheless, stemflow has provided an important localized source of soil moisture and chemical inputs through the rainwater funneling process in which a spatial concentration of water and solutes occurs particularly at the base of tree trunks (Levia and Frost 2003). Stemflow input quantities into forest and agricultural soils are highly variable between and within types of vegetation cover.

In agroforestry systems both the tree and crop canopies affect rainfall distribution and the input to the soil surface. In coffee agroforestry great variability in rainfall interception from canopy surfaces has been demonstrated. Comparisons between coffee agroforestry with low and high shade density in India demonstrated that interception increased with shade density and varied from 9% to 16% of the gross rainfall for low and high shade levels, respectively (Gurav et al. 2012). High rates of interception were found in coffee agroforestry in optimal environmental conditions in Colombia with around 47% of gross rainfall in full sun (FS) and 54.5% in agroforestry (AFS) averaged for different shade tree species (Jaramillo 2003). A study in Coffea arabica shaded with Erythrina poeppigiana which was pruned periodically showed only 3.5% of the total rainfall was intercepted by the pruned shade tree (Calder 1998). The effect of rainfall amount on rainfall interception was demonstrated in coffee shaded with Inga densiflora in Costa Rica. It was found that interception loss reached values close to 75% for 1mm rainfall and 50% for rainfall around 5 mm while for rainfall greater than 30 mm interception decreased to 10% (Siles et al. 2010). Such variability in rainfall interception estimation in coffee agroforestry clearly indicates the need for further studies. To assess the water loss by interception from the system is of considerable interest, particularly in dry climates.

Although the process of retention and subsequent evaporation of water droplets on vegetation has been extensively studied in forests, it is still poorly understood and hardly taken into account in agroforestry systems. Those studies cited above undertaken on coffee agroforestry were insufficient to show consistent results for throughfall distribution or a pattern of stemflow in coffee and shade trees. The interaction between rainfall and vegetation characteristics is relevant for understanding water relations in coffee agroforestry taking into account the different environmental conditions. This chapter presents a comparison between rainfall interception in unshaded coffee and coffee agroforestry shaded with *Tabebuia rosea* and *Simarouba glauca* timber tree species in a coffee agroforestry system in Nicaragua. The objective is to contribute to better understanding of the effects of rainfall partitioning on the water balance of the coffee agroforestry system and the effects of the two tropical timber tree species on rainfall distribution.

4.2. MATERIALS AND METHODS

4.2.1. Site description and experimental design

The study was carried out in an experiment located at Jardin Botánico, Masatepe, Department of Masaya, southern Nicaragua (11^0 53' 54" N, 86⁰ 08' 56" W), a long term research site managed by the Centro Agronómico Tropical de Investigación y Enseñanza (CATIE), jointly with the Universidad Nacional Agraria (UNA), Federación Cooperativas de Ahorro y Crédito (CENECOOP-FEDECARUNA) and Instituto Nicaraguense de Tecnologia Agropecuaria (INTA). The experiment was established in 2000 and is described in Haggar *et al.* (2011). The site is located at 455 m.a.s.l. where mean annual temperature is 27°C and mean annual rainfall is 1470 mm. In 2013 the annual rainfall recorded was 1312 mm. Of the total annual precipitation 85% to 97% falls during the wet season (May-November) while a pronounced seasonal drought occurs from late November to mid-May. In this study measurements were carried out after 28 rainfall events from 26 May to 11 October 2013 during three main periods: at the beginning (May-June), in the middle (July-August) and at the end (September-October) of the wet season.

The experimental design consisted of a full sun monocrop coffee (FS) plot (1440 m²) and an adjacent coffee agroforestry system (AFS) plot (3200 m²). In both systems *Coffea arabica* density was 4000 plants ha⁻¹ throughout the experiment, 2 m between rows and 1.25 m between plants. Coffee was selectively pruned yearly in February after harvest.

In the coffee agroforestry system plot, *Coffea arabica* was associated with *Simarouba glauca* DC. (Simaroubaceae) and *Tabebuia rosea* (Bertol.) DC. (Bignoniaceae) planted as shade trees regularly mixed in the plantation. The mean density of *Tabebuia rosea* was 113 trees ha⁻¹ and that of *Simarouba glauca* was 75 trees ha⁻¹ over the period of the study. Three sub plots were established in the agroforestry system as pseudo-replicates, each one bounded by two *Simarouba glauca* trees and two *Tabebuia rosea* trees in the plot corners and the coffee plants within (mean area, 44.93 m² each). Tree specimens were representative of the average tree size in the plot. In the full sun plot coffee plants were randomly selected but stratified according to the range of coffee tree sizes.

4.2.2. Species studied

Coffea arabica L. (Rubiaceae) variety "Pacas" is a natural mutation of Bourbon discovered in 1949 in El Salvador on a farm belonging to a family named Pacas. This variety is known to be adapted to hot and dry environments. Its agro-ecological aspects were analyzed and compared to the other traditional varieties at the study site and it displayed good branching capacity, branch length and leaf production and a good yield potential in sub optimal environmental conditions (Blanco *et al.* 2003).

Simarouba glauca DC. is native to tropical and sub-tropical regions of Central America, Mexico and the Caribbean. In the study area this evergreen timber tree species reached about 17 m height, 25.5 cm DBH and canopy area of 41 m² with an average of 7.2 m canopy diameter. The stem has finely cracked and grey colored outer bark while the inner bark is creamy in color. Mean bark thickness was 0.92 cm in the study area over the experiment. This species is successful over a range of soil types and their deep root system enables effective access to water year round. It has large and waxy compound leaves approximately 20 cm in length comprising 12–16 oblong pinnae, each approximately 5 cm in length, in a dense rounded crown (Brodribb *et al.* 2003).

Tabebuia rosea Bertol. is native to Mexico, Central America, Venezuela and coastal Ecuador. Trees grow in a great variety of habitats, but thrive better in regions with annual rainfall between 1200 and 2500 mm. Trees require a dry period to initiate and enhance flowering and fruiting. It has a deep root system and tolerates occasional waterlogging (CABI 2013). It is a fairly fast growing tree. At the study site 12 year old *Tabebuia rosea* trees reached 15.5 m height and 28.7 cm DBH. The tree crown was wide with an average of 44 m² with irregular, stratified ramifications and only a few thick branches. Tree canopy diameter was about 7.4 m by the end of the experiment. Leaves are compound, digitate and long petiolate. Each leaf has five leaflets of variable size, the middle one being the largest. It is a deciduous tree that sheds its leaves in the late dry season (Gentry 2002). The bark can be grey to brown and may be vertically fissured. Mean bark thickness was 1.53 cm in the study area.

4.2.3. Rainfall

Rainfall was measured by an automatic weather station installed in the full sun plot. Sensors installed at 2.50 m height were connected to a dataloggers (CR1000 Campbell Scientific Instruments). Data were continuously collected every 30 minutes over the period of study. The weather station measured relative humidity and temperature (HMP50 Campbell Scientific Instruments) and also solar radiation (CS300 Campbell Scientific Instruments), wind speed (03101 Campbell Scientific Instruments) and rainfall with tipping bucket rain gauges (TE525MM/TE525M Campbell Scientific Instruments). Rainfall measurements were calibrated by using five manual rain gauges installed in the full sun system and as close as possible to the edge of the agroforestry system plot without being affected by the tree stratum. Throughfall and stemflow were measured for 28 rainfall events during the period of study, which summed 444.5 mm, corresponding to 33.8 % of the total annual rainfall in 2013 (1312mm). Measurements were done during 8 rainfall events from 26 May to 13 June; 11 rainfall events from 18 July to 6 August and over 9 events from 15 September to 11 October (Fig 4.1).



Fig 4.1. Daily rainfall events over the period of study and in the three periods measured: May-June, July-August and September-October 2013.

4.2.4. Leaf area index

Tree leaf area index (LAI) was calculated from hemispherical photographs taken by a Nikon Coolpix 4500 camera with a fish-eye lens oriented vertically upwards towards the tree canopy. Hemispherical photographs were analyzed by using the Gap Light Analyzer software (Frazer *et al.* 1999). Leaf area indices calculated from the hemispherical photographs were calibrated by destructive direct measurements that combined gravimetric and planimetric techniques (Jonckheere *et al.* 2004). Leaf area of the trees in the plots was calculated using tree density.

Coffee leaf area was measured on a sample of 18 and 30 coffee shoots in the full sun and AFS, respectively, over the period of study. The leaf area was calculated by using the equation established in the laboratory:

Leaf area $(cm^2) = 0.7243 * length (cm) * width (cm)$

The shoot leaf area (m^2) was obtained by multiplying mean leaf area of every 20th leaf by the total number of leaves per shoot. LAI of coffee in both systems was calculated by multiplying the average of shoot leaf area by shoot density per m^2 .
The coffee and tree LAI data interpolation from measurements carried out at the beginning, the middle and at the end of the experiment allowed estimation of LAI for each rainfall event.

4.2.5. Stemflow

Coffee and tree stemflows were measured by using collars fitted and sealed on to the stem in an upward spiral in order to collect the flowing water. Collars were installed around six tree trunks each of *Tabebuia rosea* and *Simarouba glauca*. Collars were also installed around nine and 15 coffee stems in the full sun and agroforestry plots, respectively. Water collected by collars was conducted by a tube into a bucket with lid placed on the ground and measured daily. Daily stemflow per tree species calculated at the plot scale was based on tree species density. Coffee plant density was also used for stemflow estimation at the plot scale.

4.2.6. Throughfall

Throughfall was monitored under the canopy of nine and 15 coffee plants in the full sun and agroforestry respectively. Three rain gauges (made of PVC cylinders with edges which had been beveled) were placed at intervals from the coffee trunk (0.20 m, 0.60 m and 1.0 m) in a diagonal line at ground level (Fig 4.2).



Fig 4.2. Rain gauge distribution for coffee throughfall monitoring in the full sun and shade plots.

The rainfall collected in each rain gauge was divided by the rain gauge section (0.0117 m^2) and throughfall was weighed for each distance by multiplying this throughfall by the area sampled by each gauge (m^2) divided by the area of the unitary cell. For coffee, this weighing was calculated geometrically and the result of the multiplication was divided by the total of a quarter of the area occupied by coffee tree (Fig 4.2). A total of 72 rain gauges were used for coffee throughfall measurements: 24 coffee plants equipped with three gauges. Throughfall was also monitored under six Simarouba glauca and in six Tabebuia rosea trees in the coffee agroforestry being two trees of each species within each one of the three sub plots in the shade. We used a total of 78 rain gauges regularly spaced in circles at 1.0, 2.5 and 5.5 m distant from the tree stems and above the coffee plant canopies (Fig 4.3). The sampling area of each rain gauge was calculated by using the classical Thiessen polygon technique (Tabios and Salas 1985). Placing rain gauges above coffee canopy height was necessary so that interception and throughfall could be partitioned into tree and coffee canopy components. The average volume of water measured in the morning after each rain event divided by the rain gauge section per unit area was used to calculate throughfall per tree. Throughfall was extrapolated for the whole plot based on tree density. Throughfall as a percentage of gross rainfall was calculated for a range of five rainfall event magnitudes (< 5 mm; 5-10 mm; 10-20 mm; 20-40 mm; >40 mm) occurring during the study periods.



Fig 4.3. Rain gauge (circles) distribution for tree throughfall monitoring and spatial weighing for each gauge (m²) calculated by the Thiessen polygon method in one out of three replications in the shaded plot.

4.2.7. Interception loss

Loss by canopy interception by coffee and trees was calculated by subtracting the measured daily throughfall and stemflow from gross rainfall based on the follow equation:

 $\mathsf{L} = \mathsf{R} - (\mathsf{S} + \mathsf{T})$

where: L= interception loss; R= Rainfall; S= Stemflow; T= throughfall (Rutter 1963; Gurav *et al.* 2012). Interception dynamics over the two-year period of experiment was obtained by modeling coffee and shade tree interception measurements as a function of rainfall (Gash 1979).

4.2.8. Statistical analysis

Data analysis was conducted by using InfoStat software (Di Rienzo *et al.* 2014). A general linear mixed model allowed determination of significant differences on throughfall, stemflow and interception between systems regarding LAI as a fixed effect. The influence of the distance from the coffee and tree trunks on throughfall was tested. Relationships between gross rainfall versus throughfall and versus stemflow were derived from curvilinear and linear regression analyses, and these are presented towards the end of the results section.

4.3. RESULTS

4.3.1. Leaf area index

Coffee LAI was greater in shaded coffee compared to FS over the monitoring period (p<0.001). As measurements were carried out during the growing season LAI tended to increase over the time. Changes in LAI of coffee in FS, in AFS and of *Tabebuia rosea* and *Simarouba glauca* over the experiment were presented in Table 4.1.

LAI of trees was significantly different between both tree species (p<0.0001) with mean values of 0.52 (S.E.=0.01) for *Tabebuia rosea* compared to 0.37 (S.E.=0.01) for *Simarouba glauca*. Deciduous *Tabebuia rosea* showed greater variation in LAI which is probably related to refoliation in the period studied while LAI of evergreen *Simarouba glauca* was more stable (Table 4.1).

Table 4.1. LAI of coffee in FS, coffee in AFS, *Tabebuia rosea* and *Simarouba glauca* in the three consecutive periods studied: May-June, July-August and September-October 2013. (Mean ± standard error).

	May-June	July-August	Sep-October
Coffee FS	1.95 ± 0.41	2.60 ± 0.48	3.41 ± 0.01
Coffee AFS	3.05 ± 0.15	3.73 ± 0.46	4.76 ± 0.01
Tabebuia rosea	0.25 ± 0.25	0.50 ± 0.18	0.78 ± 0.31
Simarouba glauca	0.33 ± 0.02	0.35 ± 0.04	0.41 ± 0.06

In AFS LAI of coffee and trees combined was 37% greater than LAI in FS coffee with mean total LAI of 2.66 (SE=0.03) and 4.26 (SE=0.02) over the monitoring period in FS and AFS, respectively.

4.3.2. Stemflow

Stemflow increased gradually over the three periods studied reaching greater values in the last period when most heavy storms occurred. Coffee stemflow was linearly related to rainfall in FS (R^2 =0.99) and to tree throughfall (R^2 =0.98) in AFS (Fig 4.4).

Coffee stemflow as a percentage of incident rainfall was greater in AFS compared to FS (p=0.0001). Stemflow varied between 1.3% and 1.6% of rainfall in FS and of tree throughfall in AFS respectively. Coffee stemflow as a percentage of gross rainfall in FS and tree throughfall in AFS for the five magnitudes of rainfall studied are presented in Table 4.2.

Table 4.2. Stemflow as a percentage of gross rainfall (or throughfall) in coffee FS, coffee AFS, *Tabebuia rosea* (TR) and *Simarouba glauca* (SG) for five magnitudes of rainfall over the period of study. SE is the standard error of the mean.

	Coffee FS		Coffee AFS		TR		SG	
	%	SE	%	SE	%	SE	%	SE
< 5 mm	0.8	±0.14	1.9	±0.37	0.9	±0.25	0.6	±0.16
5 - 10 mm	1.0	± 0.09	1.3	±0.07	1.8	± 0.17	1.1	± 0.08
10 - 20 mm	1.4	± 0.07	1.6	± 0.05	2.4	±0.07	1.4	± 0.08
20 - 40 mm	1.5	± 0.05	1.6	± 0.05	2.2	±0.02	1.3	± 0.03
> 40 mm	1.6	± 0.02	1.8	± 0.02	2.1	±0.02	1.3	±0.02



Fig 4.4. Coffee stemflow as a function of rainfall in FS and tree throughfall in AFS over the period of study. Vertical bars represent the standard error of the mean.

Tree stemflow varied with tree species and was greater in *Tabebuia rosea* compared to *Simarouba glauca* (p=0.0001) (Fig 4.5). Tree stemflow represented an average of 1.9% and 1.1% of gross rainfall for *Tabebuia rosea* and *Simarouba glauca*, respectively (Table 4.2). Differences may be related to variations on bark width and texture. The *Tabebuia rosea* bark was found to be thick (1.53 cm; SE=0.27), coarse and fissured while *Simarouba glauca* bark was thin (0.92 cm; SE=0.04) and smooth. Relationship between stemflow and rainfall was strong and positive for both shade tree species (R^2 =0.98).







4.3.3. Throughfall

Linear regression analysis showed strong and positive relationships between throughfall below coffee trees with gross rainfall in FS (R^2 =0.99) and with tree throughfall below shade trees in AFS (R^2 =0.98) (Fig 4.6).



Fig 4.6. Throughfall below coffee trees in FS and AFS as a function of rainfall in FS and the tree throughfall in AFS over the period of study. Vertical bars represent standard error of the mean.

Differences between systems were not significant (p=0.08) for throughfall expressed as a percentage of gross rainfall being 65% and 69% in AFS and FS, respectively. In FS coffee throughfall as a fraction of rainfall tended to increase gradually with rainfall amount in unshaded coffee and ranged between 61% and 84% of gross rainfall for rainfall events up to 5mm and above 40 mm, respectively. In AFS coffee throughfall reached its maximum at 84% of rainfall when rainfall events were between 20-40 mm (Table 4.3).

	Co	ffee FS	Со	ffee AFS		SG		TR
	%	SE	%	SE	%	SE	%	SE
< 5 mm	61	± 2.0	53	± 2.0	31	± 1.0	27	± 1.0
5 - 10 mm	62	± 2.0	65	± 3.0	53	± 1.0	47	± 1.0
10 - 20 mm	71	± 2.0	78	± 3.0	72	± 1.0	66	± 1.0
20 - 40 mm	79	± 3.0	84	± 4.0	81	± 1.0	68	± 1.0
> 40 mm	84	± 5.0	75	± 4.0	80	± 1.0	68	± 2.0

Table 4.3. Throughfall as a percentage of gross rainfall (or throughfall) by coffee FS, coffee AFS, *Tabebuia rosea* (TR) and *Simarouba glauca* (SG) for five magnitudes of rainfall. SE is the standard error of the mean.

The effect of the distance from the trunk on coffee throughfall was tested and this effect varied between systems (p<0.001) being greater in FS compared to AFS. The large differences in such effect on coffee throughfall between systems may be explained by greater foliage in the shade with 31% greater LAI compared to FS over the period of the experiment. Throughfall measured beneath coffee trees in FS was lowest at 0.20 m from the coffee stem with 35% of rainfall while at 0.60 m and 1.0 m distance from the coffee trunk was similar with 89% and 79% of incident rainfall, respectively (Fig 4.7). In AFS coffee throughfall expressed as a percentage of tree throughfall increased gradually with the distance from the coffee trunk and differences between distances were significant (p=0.01). Coffee throughfall represented 57%, 66% and 73% of tree throughfall at 0.20 m, 0.60 m and 1.0 m from the coffee trunk, respectively (Fig 4.8).



Fig 4.7. Throughfall below coffee canopies as a function of rainfall in FS at different distances from the coffee trunk (0.20, 0.60 and 1.0 m).



Fig 4.8. Throughfall below coffee canopies as a function of tree throughfall in AFS at different distances from the coffee trunk (0.20, 0.60 and 1.0 m).

Throughfall measured beneath shade tree canopies varied with tree species (p<0.001) being greater beneath *Simarouba glauca* trees compared to *Tabebuia rosea*. The ratio of tree throughfall to gross rainfall was 0.61 and 0.53 beneath *Simarouba glauca* and *Tabebuia rosea*, respectively. A positive and strong relationship was found between throughfall below both tree species and rainfall (R^2 =0.99) (Fig 4.9). A general tendency of increasing in *Tabebuia rosea* and *Simarouba glauca* throughfall expressed as a percentage of gross rainfall was observed (Table 4.3).



Throughfall (mm)

Fig 4.9. Mean tree throughfall beneath six *Tabebuia rosea* and six *Simarouba glauca* as a function of gross rainfall over the period of study. Vertical bars represent the standard error of the mean.

Throughfall beneath *Tabebuia rosea* was similar at distances from the tree trunk (p=0.08) (Fig 4.10) while beneath *Simarouba glauca* was similar at 2.5 m and 5.5 m and differed at 1.0 m distance from tree trunk (Fig 4.11). Tree throughfall as a percentage of gross rainfall represented 50%, 56% and 64% of rainfall at 1.0 m, 2.5 m and 5.5 m distance from *Tabebuia rosea* stem, respectively, while beneath *Simarouba glauca* throughfall was 55%, 65% and 69% of rainfall, respectively, for the same distances.



Fig 4.10. Spatial distribution of throughfall beneath *Tabebuia rosea* (TR) as a function of gross rainfall at different distances from the tree trunk (1.0 m, 2.5 m and 5.5 m).



Fig 4.11. Spatial distribution of throughfall beneath *Simarouba glauca* (SG) as a function of gross rainfall at different distances from the tree trunk (1.0 m, 2.5 m and 5.5 m).

4.3.4. Rainfall interception

Water loss by interception by coffee plants as a proportion of rainfall was similar in both systems (p=0.17). Interception expressed as a percentage of rainfall represented 33% and 30% of gross rainfall in AFS and FS, respectively (Fig 4.12). However, coffee interception performed differently in both systems depending on rainfall size. Rainfall interception expressed as a percentage of rainfall tended to diminish gradually with increasing in rainfall amount in both systems (Table 4.4). Greater variability in rainfall interception in AFS compared to FS was likely due to the inherent variability in tree throughfall as a component of rainfall interception estimation in the shaded environment.



Fig 4.12. Rainfall interception by coffee trees in FS and AFS over the period of study.

Overall interception by shade trees varied with tree species (p<0.0001). Interception expressed as a fraction of rainfall represented 44% and 38% of incident rainfall by *Tabebuia rosea* and *Simarouba glauca* canopies, respectively, over the monitoring period (Fig 4.13). Interception by shade trees expressed as a percentage of gross rainfall was greater for small rainfall events below 5mm when reached 69% and 72% of rainfall in *Simarouba glauca* and *Tabebuia rosea* respectively (Table 4.4). The percentage of canopy interception loss diminished gradually with increasing in rainfall

size and reached 18% and 30% of gross rainfall for big rainfall events above 40 mm, in *Simarouba glauca* and *Tabebuia rosea,* respectively (Table 4.4).



Fig 4.13. Interception by *Tabebuia rosea* (TR) and *Simarouba glauca* (SG) canopies as a function of rainfall. Vertical bars represent the standard error of the mean.

Table 4.4. Rainfall interception expressed as a fraction of rainfall or tree throughfall in coffee FS, coffee AFS, *Simarouba glauca* and *Tabebuia rosea* over the period of experiment.

	Coffee FS		Coffee AFS		SG		TR	
	%	SE	%	SE	%	SE	%	SE
< 5 mm	38	± 2.0	44	± 2.0	69	± 1.0	72	± 1.0
5 - 10 mm	37	± 2.0	34	± 3.0	46	± 1.0	51	± 1.0
10 - 20 mm	28	± 2.0	21	± 3.0	26	± 1.0	30	± 1.0
20 - 40 mm	20	± 2.0	15	± 4.0	18	± 1.0	30	± 1.0
> 40 mm	15	± 5.0	24	± 4.0	18	± 2.0	30	± 2.0

Globally rainfall partitioning in throughfall, stemflow and interception expressed as a percentage of gross rainfall (Table 4.5) showed that stemflow accounted for the minor proportion of rainfall while throughfall was the most important contribution for interception estimation. In agroforestry most rainfall interception was due to shade trees being greater by *Tabebuia rosea* (Table 4.5).

Table 4.5. Rainfall partitioning in throughfall, stemflow and interception expressed as a fraction of rainfall or tree throughfall in coffee FS, coffee AFS, *Simarouba glauca* and *Tabebuia rosea* over the period of experiment.

	Throughfall		Stemflow]	nterceptic	n
	%	SE	%	SE	%	SE
Coffee FS	69	± 3.0	1.3	±0.1	30	± 3.0
Coffee AFS	65	±3.0	1.6	±0.2	33	± 5.0
Tabebuia rosea	53	±1.0	1.8	±0.2	44	±1.0
Simarouba glauca	61	±1.0	1.0	±0.1	38	± 1.0

4.3.5. Rainfall interception modelling

Rainfall interception measurement and modelling were used in order to extrapolate the results obtained for the whole period of study. In agroforestry, rainfall interception modelling may consist in an important tool to predict the effects of water loss by coffee and shade tree canopies, particularly in water limiting condition for coffee cultivation. Coffee and tree daily interception values derived from throughfall and stemflow measurements were modelled as a function of incident rainfall or tree throughfall by using Michaelis-Menten equation:

$$I = \frac{Imax * R}{Km + R}$$

Where:

I is the interception $I_{max} \text{ is the maximum interception}$ R is the rainfall k_m is the Michaelis-Menten constant

 K_m is equal to the rainfall amount in which the interception is half of its maximum value. The model yields an asymptotic equation in which for low rainfall rates interception is proportional to rainfall and for rainfall rates above k_m values interception is equal to the interception maximum and is independent of rainfall rates. Thus, the

model is based on two main variables: the rainfall and the maximum interception. Coffee interception modelled in FS showed that maximum interception was attained at 2.0 mm with k_m of 2.4 mm rainfall threshold (Fig 4.14) while in AFS maximum interception of 5.7 mm was obtained with K_m of 13 mm. Regarding the inherent difficult to accurately estimate interception in AFS due to the heterogeneous rainfall produced by the shade tree canopies effect we choose to adjust maximum interception by coffee in the shade. We used coffee LAI as a reliable reference rather than the rainfall obtained from tree throughfall. Thus, mean coffee LAI in AFS related to coffee LAI in FS in the period of measurements allowed obtaining 3.0 mm maximum interception.



Fig 4.14. Interception measured and modelled by coffee tree canopies in FS.

The model fitted well for *Simarouba glauca* trees and reached maximum interception at around 4.6 mm with k_m of 4.5 mm rainfall (Fig 4.15). However, for *Tabebuia rosea* maximum interception was estimated at 25 mm with k_m of 64 mm, which is clearly unrealistic (Fig 4.16).



Fig 4.15. Rainfall interception measured and modelled by *Simarouba glauca* canopy.



Fig 4.16. Rainfall interception measured and modelled by *Tabebuia rosea* canopy.

Coefficients from Michaelis-Menten equation (Table 4.6) were applied for coffee and tree interception modeling. In AFS coffee interception modeling took into account tree throughfall calculated from the difference between rainfall and the sum of tree interception and tree stemflow. Coefficients for tree stemflow modeling were obtained from stemflow measured in each tree species versus gross rainfall regression (Table 4.7). Tree interception and stemflow were up scaled to the agroforestry plot level by using tree density. Interception by trees and shaded coffee summed allowed the whole agroforestry system interception calculation.

	I _{max}	K _m
Coffee FS	2.0	2.4
Coffee AFS	3.0	13.2
TR	25.0	63.8
SG	4.6	4.5

Table 4.6. Coefficients for coffee FS, coffee AFS, *Tabebuia rosea* (TR) and *Simarouba glauca* (SG) interception (RI) by the equation: I=a*b/a+b

Table 4.7. Regression coefficients for *Tabebuia rosea* (TR) and *Simarouba glauca* (SG) stemflow (Sf).

	а	b	е
TR		0.022	-0.0118
SG		0.013	

The model applied to the whole set of rainfall events over the two-year period of observation allowed estimating changes over time of interception in both systems. Pearson correlation coefficients between interception measured and modelled was 0.76 and 0.90 in FS and AFS, respectively. Simulations showed that globally contrasting precipitation pattern observed between both years of study did not have an effect on rainfall interception (p=0.22). Differences between years were found when comparing interception in the wet seasons (p=0.01) rather than in the dry periods (p=0.15). Irrespective of the system, rainfall interception expressed as a percentage of rainfall represented 28% and 33% of rainfall in 2012 and 2013 wet seasons while in the dry represented 10% and 11% of gross rainfall in 2012 and 2013, respectively. Comparison between systems showed greater interception in AFS (p<0.0001) with 18% and 24% of gross rainfall in FS and AFS, respectively, over the period of the experiment. Rainfall interception as a percentage of rainfall in both systems by year and season is presented in Table 4.8.

Year	System	Season	Interception (%)	SE
	AEC	Dry	11	± 2.0
2012	AFS	Wet	33	± 2.0
2012	FS	Dry	10	± 2.0
	15	Wet	24	± 2.0
	AFS	Dry	12	± 2.0
2013	AI 5	Wet	38	± 2.0
2015	FS	Dry	11	± 2.0
	15	Wet	27	± 2.0

Table 4.8. Interception as a percentage of gross rainfall over the dry and wet seasons in 2012 and 2013 in shaded (AFS) and unshaded (FS) systems.

4.4. DISCUSSION

4.4.1. Stemflow

This study demonstrated that the average coffee stemflow was quite low. Mean coffee stemflow was between 1.6% and 1.3% of gross rainfall or tree throughfall. Mean coffee stemflow was similar to previous studies in coffee shaded with *Eucalyptus deglupta* in Costa Rica in which stemflow was found to be 2% and 3% of rainfall in unshaded coffee and in coffee shaded with *Eucalyptus deglupta*, respectively (Harmand *et al.* 2007). On the other hand, our results did not agree with another study on coffee shaded with *Inga densiflora* growing in higher rainfall and optimal environmental conditions for coffee cultivation in Costa Rica, which reported values for coffee stemflow of 10.6% and 7.2% of gross rainfall (Siles *et al.* 2010). Variations in precipitation pattern, on stem area and stand density could contribute to explaining contrasting results. However, few studies are available on rainfall partitioning in coffee agroforestry and, in most of them, stemflow measurements are considered negligible thereby precluding comparisons.

Stemflow measured in shade trees was often greater down *Tabebuia rosea* tree trunks when compared to *Simarouba glauca* trees which could be attributed to the inherent species-specific differences in tree architecture, including bark texture and bark water storage capacity. However, flaky-barked species such as *Tabebuia rosea* was expected to display greater water-holding capacity and therefore, lower stemflow generation

than smooth-barked Simarouba glauca (Crockford and Richardson 1990). This contradiction may be explained by the inherent difficulty of fitting collectors on a rough trunk with a watertight seal, and possible meteorological influences on water loss from collectors under great rainfall intensity. Our results for stemflow of 1.8% and 1.1% of rainfall for Tabebuia rosea and Simarouba glauca respectively were similar of that reported for Inga densiflora of 1.2% of gross rainfall in coffee agroforestry (Siles et al. 2010). Results were also similar to stemflow of 1.28% of gross rainfall found for Inga spp in coffee agroforestry in Colombia (Castaño and Jaramillo 1999). Our results in tree stemflow are also within the ranges reported for five tropical timber tree species in a Panamanian plantation which varied from 0.9% to 2.7% of rainfall (Park and Cameron 2008). Stemflow was found to be of 1.2% of gross rainfall in maize and Grevillea robusta in agroforestry system in semi-arid Kenya (Jackson 2000) while in a perennial polyculture in central Amazonia stemflow varied from 0.1% to 1.4% of rainfall for different tree species being similar of 1.8% of precipitation measured in a primary rainforest nearby the study site (Schroth et al. 1999). There is a large consensus that this component is relatively small. Further experiments and improvements on techniques for stemflow measurement are needed in order to get better accuracy of results. However, investing more resources on this component might not be the most efficient strategy to improve our understanding of water flows in agroforestry systems.

4.4.2. Throughfall

In this study, throughfall beneath coffee canopies as a percentage of incident rainfall varied from 69% to 65% in FS and AFS, respectively (p=0.08). This result may be influenced by lower LAI in unshaded coffee plants which could favor raindrops to pass through the coffee canopies in FS. Reduced LAI and canopy extension in coffee plants in FS compared to coffee bushes in AFS could also explain the great variability in coffee throughfall with distance from the coffee stem in the unshaded environment. In contrast greater canopy cover in the shaded coffee led to a lower influence of the distance from coffee trunk (especially near the trunk at 0.20 m and 0.60 m) on throughfall rates. Comparison with Harmand *et al.* (2007) findings showed greater throughfall of 82% in coffee throughfall under the wider canopy cover of *Simarouba glauca* and *Tabebuia rosea* coffee, throughfall was 65% of incident rainfall. Despite the differences in coffee density and management, this may indicate the potential effects

of different shade tree species on rainfall beneath tree canopies with some influence on drop characteristics that reach the crops (Calder *et al.* 1996; Jaramillo and Chaves 1998).

However, coffee throughfall in AFS was calculated in relation to throughfall across tree canopies. This throughfall, as contrasted with incident rainfall, is highly heterogeneous. In our calculation, we assumed that coffee bushes were representative of the whole plot, and received, as a mean, the average throughfall. If this assumption is not correct, and the coffee bushes used for measuring coffee throughfall were located closer to tree trunks than the average situation, then those particular bushes could receive less throughfall than the average, and throughfall percentage might be underestimated.

In this study tree throughfall showed greater values beneath *Simarouba glauca* shade tree which had lower LAI and smaller leaves compared to *Tabebuia rosea*. This result is in agreement to the tendency of lower canopy cover and small leaves to produce bigger throughfall as previously reported by Geisler *et al.* (2013) and by Goebes *et al.* (2015). Moreover, the waxed leaf surface characteristic of *Simarouba glauca* seemed to facilitate water passing through the canopy compared to the rough leaf surface of *Tabebuia rosea* which indicated that leaf traits may have some influence on throughfall and interception rates.

4.4.3. Interception loss

We demonstrated that in the agroforestry system, tree and coffee canopies combined led to a greater reduction of water availability for plant growth compared to FS likely due to 37% greater total LAI in the shade. However, most interception in AFS was found to be by shade tree canopy rather than by the coffee canopy despite greater coffee LAI which may suggest some influence of plant architecture and canopy cover extent on interception rates. Moreover, leaf traits such as broader leaves and roughness influenced greater water retention on the leaf surface of *Tabebuia rosea* when compared to smaller leaves of coffee and *Simarouba glauca*. A study by Germer *et al.* (2006) showed that interception can vary broadly due to different leaf traits for canopies with the same LAI.

Water loss by interception that is indirectly estimated from throughfall and stemflow measurements is highly dependent from how accurately both terms are measured. In

this study bias on interception estimation, especially on *Tabebuia rosea* tree canopy and on coffee canopy in AFS, may be attributed to underestimation of both the stemflow and throughfall measurements. Bias was found especially during high rainfall events when rainfall water could have been splashed from the collectors which were not detected during the measurements. It is expected the most potential influence would be derived from throughfall measurements as this is the largest component in rainfall partitioning. Although the Thissen polygon technique applied in this study contributed to capturing the spatial variability of throughfall on a micro scale, adjustments towards a more regular pattern in the spatial arrangement of collectors rather than the orthogonal collector distribution could be recommended in future in order to minimize this potential source of error.

In this study simulations showed 24% of incident rainfall intercepted in AFS which was greater than interception of 3.5% of rainfall in coffee with *Erythrina poeppigiana* (periodically very severely pruned, virtually to stumps) and of 13.5% in coffee with *Cordia alliodora* (Imbach *et al.* 1989). Nevertheless, these figures come from a climate that is altogether very different from that of our study site, with many more rainfall events and a wetter climate. These differences could account, at least partly, for the differences in the percentage of intercepted rainfall. Our result was also greater than interception of 8.9% and 15.7% of gross rainfall found in coffee agroforestry with low (230 tree ha⁻¹) and high shade (390 tree ha⁻¹) levels reported by Gurav *et al.* (2012). The lower tree density (188 tree ha⁻¹) and greater interception in this study can be attributed to different shade tree characteristics, coffee agroforestry management and environmental condition.

This study showed daily interception of 0.61 mm (SE=0.05) in FS and 1.06 mm (SE=0.05) in AFS which was similar to daily interception of 0.63 mm and 0.66 mm in FS and 1.16 mm and 0.95 mm in AFS found by Cannavo *et al.* (2011). We demonstrated that in both systems most water loss by interception occurred in the wet seasons rather than the dry, when more rainfall water was available. Cannavo *et al.* (2011) showed similar values for interception as a percentage of gross rainfall in the dry season of 11.9% and 17.4% of rainfall while our study showed 10% and 11% of rainfall in FS and AFS, respectively. However, in the wet season our results were quite greater with 26% and 35% of total rainfall compared to 7.89% and 11.37% of rainfall in FS and AFS, respectively. Variations AFS may be related to uncertainties in

throughfall measured in both the coffee and shade trees with effects on interception rate estimation.

We demonstrated that contrasting precipitation pattern over the both years of study did not have an effect on interception in the whole. Differences were observed in the wet seasons (p=0.01) and not in the dry periods (p=0.15). In 2013 wet season interception was greater than in 2012 wet season in both systems following the greater rainfall amount in the second year. Interception represented 24% and 27% of gross rainfall and 33% and 38% of rainfall in FS and AFS in 2012 and 2013, respectively. In the dry seasons simulations showed similar interception between years and systems which represented 10% and 11% of rainfall and 11% and 12% of rainfall in FS and AFS in 2012 and 2013, respectively.

4.5. CONCLUSIONS

This study demonstrated the substantial effect of *Tabebuia rosea* and *Simarouba glauca* shade trees on the reduction of amount of rainfall reaching the soil surface in a coffee agroforestry system. Interception of rainfall by the canopy was significant in the agroforestry system and could result in greater soil moisture deficits in a sub-optimal climatic environment such as experienced at this site. However, the proportion of rainfall intercepted was notably less during the dry season which is when soil moisture deficits had their most serious effects, compared to the wet season. The results reinforced the effect of shade tree species characteristics, canopy cover and rainfall variations on rainfall intercepted over the period of study.

4.6. REFERENCES

- Beer, J. 1987. Advantages, disadvantages and desirable characteristics of shade trees for coffee, cacao and tea. Agroforestry Systems 5: 3-13.
- Blanco, M.; Haggar, J.; Moraga, P.; Madriz, J.C.; Pavón, G. 2003. Morfología del café (*Coffea arabica* L.) en lotes comerciales. Nicaragua. Agronomia mesoamericana 14(1): 97-103.
- Brodribb, T.J.; Holbrook, N.M.; Edwards, E.J.; Gutiérrez, M.V. 2003. Relations between stomatal closure, leaf turgor and xylem vulnerability in eight tropical dry forest trees. Plant, Cell and Environment (26): 443-450.
- CABI. 2013. The CABI Encyclopedia of forest trees, CAB International. 536 p.

- Calder, I.R.; Hall, R.L.; Rosier, P.T.W.; H.G., B.; Prasanna, K.T. 1996. Dependence of rainfall interception on drop size: 2. Experimental determination of the wetting functions and two-layer stochastic model parameters for five tropical tree species. Journal of Hydrology 185(1-4): 379-388.
- Calder, I.R. 1998. Water use by forests, limits and controls. Tree physiology 18: 625-631.
- Cannavo, P.; Sansoulet, J.; Harmand, J.M.; Siles, P.; Dreyer, E.; Vaast, P. 2011. Agroforestry associating coffee and *Inga densiflora* results in complementarity for water uptake and decreases deep drainage in Costa Rica. Agriculture, Ecosystems & Environment 140(1-2): 1-13.
- Cannell, M.G.R.; Van Noordwijk, M.; Ong, C.K. 1996. The central agroforestry hypothesis: the trees must acquire resources that the crop would not otherwise acquire. Agroforestry Systems 34: 27-31.
- Castaño, O.; Jaramillo, A. 1999. Escurrimiento de agua y nutrimentos por el tallo en árboles de guamo (*Inga* sp). Chinchiná, Cenicafé. 22 p.
- Crockford, R.H.; Richardson, D.P. 1990. Partitioning of rainfall in a eucalypt forest and pine plantation in southeastern Australia: II. Stemflow and factors affecting stemflow in a dry sclerophyll eucalypt forest and a Pinus radiata plantation. Hydrological Processes 4: 145-155.
- Di Rienzo, J.; Casanoves, F.; Balzarini, M.; Gonzalez, L.; Tablada, M.; Robledo, C. 2014. InfoStat Manual. Argentina., Universidad Nacional de Córdoba. 336 p.
- Int. Symp. For. Hydrol. 1967. Interception and soil moisture distribution. Sopper, W.E., Hull, H.W. (Eds.). ed. Toronto, Pergamon Press. 191-200. p.
- Gash, J.H.C. 1979. An analytical model of rainfall interception by forests. Journal of the Royal Meteorological Society 105: 43-55.
- Geisler, C.; Nadrowski, K.; Kühn, P.; Baruffol, M.; Bruelheide, H.; Schmid, B.; Scholten,
 T. 2013. Kinetic Energy of Throughfall in Subtropical Forests of SE China –
 Effects of Tree Canopy Structure, Functional Traits, and Biodiversity.
 10.1371/journal.pone.0049618
- Gentry, A.H. 2002. Tropical forest biodiversity: distributional patterns and their conservational significance. OIKOS 63(1): 19-28.
- Germer, S.; Elsenbeer, H.; Moraes, J.M. 2006. Throughfall and temporal trends of rainfall redistribution in an open tropical rainforest, south-western Amazonia (Rondonia, Brazil). European Geosciences Union 10(3): 383-393.

- Goebes, P.; Bruelheide, H.; Härdtle, W.; Kröber, W.; Kühn, P. 2015. Species-specific effects on throughfall kinetic energy in subtropical forest plantations are related to leaf traits and tree architecture. 10.1371/journal.pone.0128084
- Gurav, M.; Kumar, S.; G., K.C.; Vaast, P. 2012. Throughfall and Interception Loss in Relation to Different Canopy Levels of Coffee Agroforestry Systems. International Journal of Environmental Sciences 1(No.3): 145-149.
- Haggar, J.; Barrios, M.; Bolaños, M.; Merlo, M.; Moraga, P.; Munguia, R.; Ponce, A.; Romero, S.; Soto, G.; Staver, C.; de M. F. Virginio, E. 2011. Coffee agroecosystem performance under full sun, shade, conventional and organic management regimes in Central America. Agroforestry Systems 82(3): 285-301.
- Harmand, J.M.; Avila, H.; Dambrine, E.; Skiba, U.; De Miguel, S.; Renderos Duran, R.V.; Oliver, R.; Jimenez, F.; Beer, J. 2007. Nitrogen dynamics and soil nitrate retention in a Coffea arabica-Eucalyptus deglupta agroforestry system in Southern Costa Rica. Biogeochemistry 85: 125-139.
- Horton, R.E. 1919. Rainfall interception. *In* Brooks, C.F. ed. 1919. Monthly Weather Review. N.Y., p. 603-623.
- Imbach, A.C.; Fassbender, H.W.; Beer, J.; Borel, R.; Bonneman, A. 1989. Sistemas agroforestales de café (Coffea arabica) con laurel (Cordia alliodora) y café con poro (Erythrina poeppigiana) en Turrialba, Costa Rica. *In.* 1989. Acta Agronómica - Universidad Nacional de Colombia. Turrialba, p. 400-414.
- Jackson, N.A.; Wallace, J.S. 1999. Soil evaporation measurements in an agroforestry system in Kenya. Agricultural and Forest Meteorology 94: 203-215.
- Jaramillo, A.; Chaves, B. 1998. Intercepcion de la lluvia en un bosque y en plantaciones de *Coffea arabica* L. Cenicafe 42: 129-135.
- Jaramillo, A. 2003. La lluvia y el transporte de nutrimentos dentro de ecosistemas de bosque y cafetales. Cenicafe 54: 134-144.
- Jonckheere, I.; Fleck, S.; Nackaerts, K.; Muysa, B.; Coppin, P.; Weiss, M.; Baret, F. 2004. Review of methods for in situ leaf area index determination Part I. Theories, sensors and hemispherical photography. Agricultural and Forest Meteorology 121: 19-35.
- Levia, D.F.; Frost, E.E. 2003. A review and evaluation of stemflow literature in the hydrological and biogeochemical cycles of forested and agricultural ecosystems. Journal of Hydrology 274: 1-29.

- Lunka, P.; Patil, S.D. 2016. Impact of tree planting configuration and grazing restriction on canopy interception and soil hydrological properties: implications for flood mitigation in silvopastoral systems. Hydrological Processes 30(6): 945-958.
- Marin, C.T.; Boutena, W.; Sevink, J. 2000. Gross rainfall and its partitioning into throughfall, stemflow and evaporation of intercepted water in four forest ecosystems in western Amazonia. Journal of Hydrology 237: 40-57.
- Ong, C.K.; Black, C.R. 1994. Complementarity of resource use in intercropping and agroforestry systems. *In* Monteith, J.L.; Scott, R.K.; Unsworth, M.H. eds. 1994. Resource Capture by Crops. Loughborough, Nothinghan University Press. p. 255-278.
- Park, A.; Cameron, J.L. 2008. The influence of canopy traits on throughfall and stemflow in five tropical trees growing in a Panamanian plantation. Forest Ecology and Management 255: 1915-1925.
- Rao, M.R.; Nair, P.K.R.; Ong, C.K. 1998. Biophysical interactions in tropical agroforestry system. Agroforestry Systems 38: 3-50.
- Rutter, A.J. 1963. Studies in the Water Relations of *Pinus Sylvestris* in Plantation Conditions I. Measurements of Rainfall and Interception. Journal of Ecology 51(1): 191-203.
- Rutter, A.J.; Kershaw, K.A.; Robins, P.C.; Morton, A.J. 1971. A predictive model of rainfall interception in forests, 1. Derivation of the model from observations in a plantation of *Corsican pine*. Agricultural Meteorology 9: 367-384.
- Rutter, A.J. 1975. The hydrological cycle in vegetation. Vegetation and the atmosphere. 1: 111-154.
- Siles, P.; Vaast, P.; Dreyer, E.; Harmand, J.M. 2010. Rainfall partitioning into throughfall, stemflow and interception loss in a coffee (*Coffea arabica* L.) monoculture compared to an agroforestry system with *Inga densiflora.* Journal of Hydrology 395: 39-48.
- Tabios, G.Q.; Salas, J.D. 1985. A comparative analysis of techniques for spatial interpolation of precipitation. Journal of the American Water Resources Association 21: 365-380.
- Xiao, Q.; McPherson, E.G.; Ustin, S.L.; Grismer, M.E.; Simpson, J.R. 2000. Winter rainfall interception by two mature open-grown trees in Davis, California Hydrological Processes (14): 763-784

CHAPTER V

WATER LOSS BY PLANT TRANSPIRATION AND SOIL EVAPORATION IN COFFEE SHADED BY *TABEBUIA ROSEA* BERTOL. AND *SIMAROUBA GLAUCA* DC. COMPARED TO UNSHADED COFFEE IN SUB-OPTIMAL ENVIRONMENTAL CONDITIONS

5.1. INTRODUCTION

Coffee production is being threatened by increasing climate variability. Coffee shade has been suggested as a promising strategy to cope with the decline of available water and the increase of temperature in the context of global climate changes. Shade trees may buffer the effects of high temperature (Barradas and Fanjul 1986; Muschler 1997; Siles *et al.* 2009) and may increase water availability for plants use by reduction of soil erosion and runoff (Beer 1995; Gomez-Delgado *et al.* 2010). However, shade trees transpire and may thus increase the whole system water use depending on the shade tree species, management, soil and environmental conditions. Potential competition for water between coffee and shade tree is therefore, one of the main disadvantages of coffee agroforestry (Beer 1987).

The assessment of competition or complementarity in water use in agroforestry may be facilitated by evapotranspiration partitioning. Evapotranspiration comprises the processes by which water changes phase from a liquid to a gas: evaporation from the soil, transpiration through the stomata of plants and evaporation of water intercepted by plant canopies (Wilcox *et al.* 2003; Kool *et al.* 2014). These processes can be highly variable in space and time (Williams *et al.* 2004). Transpiration is considered as a productive flux because it is related to plant growth while soil evaporation and interception are regarded as being unproductive once it is lost to the atmosphere and is not used for plant biomass production (Liu *et al.* 2002).

The soil evaporation component may have a very important role in water conservation in agroforestry. Evaporation from the soil is principally from top soil where most fine roots are found. Shade trees may reduce radiation and temperature of the soil surface with decrease of water loss by soil evaporation (Ritchie 1971). The effects of increasing shade tree density on the gradual diminishing of soil evaporation was reported in coffee agroforestry by Lin (2007). Reduction of the soil evaporation component in agroforestry systems may increase water retained in the soil and thus the overall proportion of rainfall used by crop and trees through transpiration (Wallace *et al.* 1999).

Transpiration, as the dominant component of evapotranspiration (Lawrence *et al.* 2006; Xu *et al.* 2008) has been assessed and compared in coffee agroforestry and full sun coffee. Van Kanten and Vaast (2006) demonstrated that coffee transpiration was often greater in the full sun while the whole system water use was greater in the shade. Cannavo *et al.* (2011) showed that the higher water use by coffee and shade trees through transpiration plus water loss by interception resulted in lower drainage when compared to full sun coffee. However, despite water dynamics and use being significantly affected by shade trees little is known about whole system water use in coffee agroforestry since most studies are addressed to one or another evapotranspiration component. Studies that integrate soil surface evaporation and plant transpiration in coffee agroforestry with appropriate techniques for both components are few.

Here we studied the contribution of coffee and shade tree transpiration and soil evaporation to the total evapotranspiration in a coffee agroforestry system by measuring each component directly. We also compared the water consumption by deciduous *Tabebuia rosea* and evergreen *Simarouba glauca* grown as coffee shade trees. The results contribute to a better understanding of water allocation within the agroforestry system and coffee responses to moisture variability. This is important in order to identify shade trees ideotypes and possible management interventions which are more suitable for coffee agroforestry in the context of scarce water resources.

5.2. MATERIALS AND METHODS

5.2.1. Site description and experimental design

The study was carried out during 2012 and 2013 in an experiment located at Jardín Botánico, Masatepe, Department of Masaya, southern Nicaragua (11° 53' 54" N, 86° 08' 56" W) at a long term research site managed by the Centro Agronómico Tropical de Investigación y Enseñanza (CATIE), jointly with the Universidad Nacional Agraria

(UNA), Federación Cooperativas de Ahorro y Crédito (CENECOOP-FEDECARUNA) and Instituto Nicaraguense de Tecnologia Agropecuaria (INTA). The experiment was established in 2000, as described by Haggar *et al.* (2011).

The site is located at 455 m.a.s.l. which is considered to be rather a low altitude for arabica coffee (*Coffea arabica* L.) at this latitude. Mean annual temperature is 27° C – above *C. arabica* comfort zone - and mean annual rainfall is 1470 mm. From 85% to 97% of the total annual precipitation falls over the wet season that lasts from May to November while a pronounced seasonal drought occurs from late November to mid-May (Vogel and Acuña Espinales 1995). Annual rainfall recorded during the experimental period was 968 mm and 1312 mm in 2012 and 2013 respectively.

Soils in the area are predominantly characterized as Andisols, which are derived from volcanic ejecta. These soils are typically deep, well drained and have high organic matter content, low bulk density, high allophane content and consequently a high phosphorus fixation capacity, high amorphous mineral content and high water retention capacity (FAO, 2001). On this particular study site, however, soils are characterized by the presence of a hardened layer locally known as talpetate. Such layers occur in about 15% of the Nicaragua Pacific region. Its properties reflect both geologic and soil-forming processes and can be extremely variable (Vogel and Acuña Espinales 1995).

Soil bulk density of the whole soil profile varied from 0.62 g cm⁻³ (S.E.=0.026) to 0.65 g cm⁻³ (S.E.=0.039) on average in the shade and full sun plots, respectively, while in the talpetate layer it reached 0.72 g cm⁻³ in both systems. The mean organic matter content of the soil profile was 3.8% (S.E.=2.0) and 3.4% (S.E.=1.05)in the shade and full sun plots respectively. In the uppermost, brown layer organic matter content was found to be 8.7% (S.E.=0.10) in the shade system and 8.2% (S.E.=0.25) in the full sun, and this decreased with depth. The pH (in H₂O) did not differ in both systems and ranged between 5.70 and 5.80, which is considered acceptable for coffee cultivation. Management includes fertilization with 37.3 kg ha⁻¹ of N, 48.8 kg ha⁻¹ of N as urea and 12 kg ha⁻¹ of K as KCl are applied each year.

As a consequence of the original layout of the experiment within which the experimental plots used in this study are located, the design consists of a full sun monocrop coffee (FS) plot (1440 m^2) and an adjacent coffee agroforestry system (AFS)

plot (3200 m²). Sub-plots for sampling were established within these main plots, as pseudo-replicates. In the coffee agroforestry system plot *Coffea arabica* is associated with a mix of *Simarouba glauca* DC. (Simaroubaceae) and *Tabebuia rosea* (Bertol.) (Bignoniaceae) planted as shade trees. Tree spacing was originally 4 m x 4 m, alternating both species (Haggar *et al.* 2011), but tree density has been reduced over the time by thinning to achieve a shade level appropriate for coffee production. The mean density of *Tabebuia rosea* was 113 trees ha⁻¹ and that of *Simarouba glauca* was 75 trees ha⁻¹ over the period of the study. Coffee density throughout of the experiment was 4000 plants ha⁻¹, spacing being 2 m between rows and 1.25 m between plants in both the agroforestry and full sun coffee. Coffee plants were pruned periodically in accordance with standard agronomic practice.

5.2.2. Species studied

The species *Coffea arabica* L., *Tabebuia rosea* Bertol. and *Simarouba glauca* DC. studied in this Chapter had been already described in the previous Chapters. The main characteristics of both timber tree species are presented in the Table 5.1.

Table 5.1. Main characteristics of the two shade tree species in the study site: *Tabebuia rosea* and *Simarouba glauca*, over the period of the experiment.

	Tabebuia rosea Bertol.	Simarouba glauca DC.
Distribution	Central America, Mexico, Venezuela and coastal Ecuador	Tropical and sub-tropical regions of Centra America, Mexico and the Caribbean
Density (tree ha ⁻¹)	113	75
Phenology	deciduous	evergreen
Leaf morfology	compound leaves, digitate and long petiolate. Each leaf has five leaflets of variable size	compound leaves 20 cm in length comprising 12–16 oblong pinnae, each approximately 5 cm in length
Leaf texture	rough	waxy
Leaf area (m ²)	55 ± 10.6	62 ± 21.3
LAI	0.62 ± 0.12	0.46 ± 0.2
Bark depth (cm)	1.5 ± 0.27	0.92 ± 0.04
Bark texture	fissured	smooth
DBH (cm)	28.7 ± 0.41	25.5 ± 0.23
Heigh (m)	15.5 ± 0.20	17 ± 2.43
Crown diameter (m)	7.4 ± 0.65	7.2 ± 0.3
Crown area (m ²)	44.4 ± 7.8	41.2 ± 3.24
Root distribution	randomly distributed in the soil profile	more concentrated in deep soil layers

5.2.3. Climate

Two automatic weather stations were installed in the full sun and shade plots. Sensors installed at 2.50 m height were connected to dataloggers (CR1000, Campbell Scientific Inc.). Data were collected every 30 minutes from February 2012 to December 2013. Both weather stations measured relative humidity and temperature (HMP50, Campbell Scientific Inc.) and the full sun plot weather station additionally measured solar radiation (CS300, Campbell Scientific Inc.), wind speed (03101, Campbell Scientific Inc.) and rainfall (TE525MM/TE525M, Campbell Scientific Inc.). Reference evapotranspiration was calculated based on the FAO Penman-Monteith equation (Allen *et al.* 1998) using data from the automatic weather station installed in the full sun plot. Rainfall measurements from the automatic station were calibrated by monitoring five manual rain gauges in the study area.

5.2.4. Soil water content

Changes in the soil water content were continuously measured during 2012 and 2013 by using time domain reflectometer (TDR) probes (Campbell Scientific Inc. CS616) that were connected to dataloggers (CR 1000 with AM 16/32B multiplexer, Campbell Scientific Inc.). These were installed horizontally, being inserted from 15 cm to 190 cm into the walls of 200 cm deep pits, which were then back-filled. Distance between TDR probes depended on the characteristic soil layer depths which were quite variable in the study area. A total of nine pits (three in the full sun plot and six in the AFS plot) were equipped with these TDR probes, located one in each soil layer down to 200 cm depth (four to six TDR per pit). Data were scanned every minute and stored every 30 minutes. Minimal disturbance soil samples of sufficient dimensions to insert a TDR probe were also extracted from each soil layer (two samples per layer). In the laboratory, the progressive drying was measured simultaneously by TDR and by weighing, so that a calibration equation was available for each soil layer, following a protocol adapted from Udawatta et al. (2011). The volumetric soil water contents of the layers in which each TDR probe was inserted were then multiplied by the thickness of each layer to calculate the soil water content at each time step.

5.2.5. Coffee and tree leaf area index

Leaf area of coffee plants were measured in the dry (February and April) and wet seasons (July and November) over 2012 and 2013. To calculate the leaf area of the

coffee plants we measured an average of 30 typical coffee shoots in the full sun plot and 35 in AFS. Shoots were purposively selected by stratifying the whole shoot population using their height and diameter in both stands. In order to estimate total leaf area of each sampled shoot, we counted the total number of leaves, and measured length and width of every 20th leaf. The area of measured leaves was calculated by using an equation established by direct measurement of leaf area in the laboratory:

Leaf area = 0.7243 * length * width

Leaf area of each shoot was then calculated by multiplying the number of leaves by the mean leaf area. LAI of the coffee plots were estimated by multiplying the mean leaf area of the shoots by coffee population density and by the mean number of shoots per coffee plant. We measured coffee leaf area four times per year in 2012 and 2013 in the wet (July and November) and the dry (February and April) seasons.

Tree leaf area was calculated by using the hemispherical photograph technique on four trees of each species four times per year (February, April, July and November) in 2012 and 2013. Hemispherical photographs were taken by using a Nikon Coolpix 4500 digital camera with a fisheye lens which permitted the taking of vertically upward photographs of the tree canopy. Images were analyzed by the Gap Light Analyzer software (Frazer *et al.* 1999) and adjusted to the crown size (681 pixels for *Tabebuia rosea* and 1000 for *Simarouba glauca*). In order to correct for the effect of branch traces in the images, hemispherical photographs of leafless *Tabebuia rosea* canopy in the dry season were used. The effect of the distance between lens and tree crown was corrected by multiplying the number of the pixels of the image by the square of the distance between lens and crown.

To calibrate this indirect method, we selected four individuals of each species and applied the procedure previously described to take hemispherical photographs and calculate the resulting black pixels. We then felled those trees and harvested all their leaves. Three samples of 100 leaves were randomly taken from the bulk leaves of each tree, and their area was measured by using a scanner (planimeter LI 3000) close to the field. This planimeter was calibrated by measuring the area of a subsample of leaves using a calibrated planimeter LI 3100 (*Simarouba glauca* R^2 =0.9785; *Tabebuia rosea* R^2 =0.9031). The leaf area and dry weight of each sample allowed the calculation of the average of specific leaf area of sample by the equation:

 $SLA = LA_s / DW_s$

Where: SLA=specific leaf area; LA_s =leaf area of sample; DW_s =total dry weight of sample.

The fresh weight of all leaves in the tree the remaining leaves of each tree was measured, its moisture content was measured on a sub sample (around 20%) and total leaf area of each felled tree was then derived from total dry weight and SLA. Leaf area of each tree (LA_t) was calculated by the equation:

 $LA_t = DW_t * SLA$

Where: $LA_t = tree leaf area; DW_t = tree leaf dry weight; SLA = specific leaf area$

To ensure the exploration of a wide range of total leaf area for each species, we felled the trees at different times. We adjusted the linear regression between direct and indirect leaf areas for *Tabebuia rosea* ($R^2 = 0.899$) and *Simarouba glauca* ($R^2 = 0.724$), and used the regression equation to calculate leaf areas of the trees of our experiment (which we did not fell) during the whole two year period of our experiment. Tree leaf area for each species was then calculated based on the mean of each tree measurement date and tree density.

5.2.6. Coffee and tree transpiration

Coffee sap flow was measured in the dry (February and April) and wet seasons (July and November) in 2012 and 2013 by using the stem heat balance method (Dynagage/Dynamax, Inc.) in four coffee shoots at a time in each plot (but moving them from bush to bush to avoid heat damage). This method was successfully tested previously in the laboratory against direct measurement of water loss in potted coffee plants by Rapidel and Roupsard (2009). Coffee shoots were selected from the average of the stem diameters in both stands which was 2.94 cm and 3.07 cm in the full and shade coffee respectively. SGB 19, 25 and 35 gauges were connected to a CR 10 X datalogger (Campbell Scientific Inc.) and coffee shoots were monitored over an average period of six consecutive days, four times per year in 2012 and 2013. Coffee stems were protected against external heat and water ingress by thermal shields. The heat source was turned off at night in order to protect the stems from overheating. Data were collected every 15 minutes. Leaf specific transpiration for each shoot was calculated by dividing the water flow (L.d⁻¹) per shoot leaf area that was measured

simultaneously. Coffee transpiration was up scaled to the plot level (mm.d⁻¹) by using LAI.

Tree sap flow rates were continuously measured over 2012 and 2013 by using the thermal dissipation technique (Granier 1985; Granier 1987) in four trees of each species. Trees were selected taking into account the average of diameters in the plot, which were 25.8 cm and 23.5 cm for Tabebuia rosea and Simarouba glauca respectively. The set of probes (one continuously heated by a constant electrical source and another as a non-heated reference probe) were inserted horizontally into tree stems (22 mm deep at 2.5 m height above the ground) with a vertical separation between probes of 15 cm. The heated probe was connected to a 137 mA potentiometer (continuous current). Trunks were insulated 1 m above and below the probes. The natural thermal gradient between the probes were measured when sensors were run with the heaters off for 10 days in March 2012. The temperature gradient between the probes was recorded on a datalogger CR 800 (Campbell Scientific Inc.) every 30 minutes from February 2012 to December 2013. The sap flow was calculated by multiplying the flow density by the conducting section area (Smith and Allen 1996). Regression analysis by using measurements of conductive cross sectional sap wood area and the DBH from the same four trees of each species in the plot allowed the calculation of coefficients to estimate the conducting section for Tabebuia rosea (R^2 =0.69) and Simarouba glauca (R^2 =0.89) from the total trunk section. As the probe length was close to the sapwood thickness it was assumed that flow rate was constant over the whole sap wood cross section (Vertessy et al. 1995).

Calibration of thermal dissipation probes was undertaken by measuring the sap flow of the same trunk using the stem heat balance method (Dynagage/Dynamax, Inc.) over eight days in different periods in 2012 and 2013. Gauges (SGA 150) were connected to a datalogger CR 800 (Campbell Scientific Inc.) and data were recorded every 15 minutes. For each species the coefficient a for the Granier equation was adjusted by optimization to reduce the sum of squares of the differences between the thermal dissipation and the stem heat balance measurements from different periods. Mean tree transpiration of each species was multiplied per *Tabebuia rosea* and *Simarouba glauca* density to obtain transpiration in the AFS plot.

5.2.7. Leaf water potential

Coffee Leaf Water Potential (LWP) at predawn - PLWP and at midday – MLWP was measured and compared in FS and AFS by using a portable pressure chamber (Scholander *et al.* 1965). The measurements were taken over a three consecutive day period, four times per year, being two during the dry season (February and April) and two in the wet season (July and November) in 2012 and 2013. Four mature and fully expanded leaves were selected at random in the upper third of the bushes of three coffee plants in each plot. The measurements were performed in the field immediately after cutting the leaves, before sunrise for PLWP and between 12:00 and 12:30 for MLWP.

5.2.8. Soil evaporation

Measurements were carried out by using seven and eight weighing lysimeters in the FS and AFS respectively, over the 2012 (April to June) and 2013 (June to November) rainy seasons and continuing into the 2014 dry season (February to April). Lysimeters were made from PVC tubes (15.7 cm internal diameter and either 20 or 30 cm length) adapted from Jackson and Wallace (1999). These were filled by soil with minimal disturbance and replaced into the holes (Daamen et al. 1995). A mesh was attached at the bottom of the tubes in order to allow excess water to drain. We used a barrier made by zinc foil (28 mm) all around the threshold of the lysimeter and the internal soil wall to avoid soil falling inside the hole when the lysimeters were removed for the weighing process. Lysimeters were located in the row between coffee plants at a distance of 0.62 m from the trunks and in the inter row 1.0 m from the coffee trunks in both plots. Lysimeters were weighed every morning before 07.00h using a portable electronic balance (0.1 g resolution). Periods for analysis were selected taking into account an interval of at least 24h after a rainfall event even if relatively small in order to avoid potential errors in measurements and the on-going drainage process that could be mistaken for evaporation. The effect of the lysimeters tube length was also analyzed.

The Ritchie model (Ritchie 1972) was used to extrapolate these measurements of soil evaporation rate over the whole period of study in FS and AFS. The model considers soil evaporation to occur in two phases: 1) a constant rate stage which depends on the radiative energy that reaches the soil surface; 2) a falling rate stage in which soil evaporation depends on upward water movement in the soil profile dependent on soil

hydraulic properties. The first phase calculation was determined by potential evaporation estimated by the Penman equation (Penman 1963) with inputs from a FS weather station and assumed to be the same over the adjacent AFS plot. Soil evaporation calculation was based on the equation:

 $Es1 = Rns * \Delta / (\gamma + \Delta) * 0.408$

Where:

Rns=net radiation at the soil surface (MJ.m⁻²);

 Δ =slope of the saturation vapor pressure curve at mean air temperature (kPa K⁻¹);

 γ = constant of the wet and dry bulb psychrometer equation (kPa K⁻¹).

Net radiation and LAI were used as inputs for net radiation at the soil surface calculation in FS and AFS at the first stage, following Ritchie (1972). The coefficients of the Ritchie model were then calibrated to minimize the sum of squares of errors between measured and calculated evaporation rates.

5.2.9. Data analysis

Variance analysis was performed to compare the influence of the systems, seasons and tree species on soil water content in the treatments by using a general linear mixedeffects model (R, Ime4 package). The same model was utilized to assess and compare coffee and tree transpiration as a function of years, systems and seasons as well as the effect of the interactions of variables. The model was also applied to compare statistical differences in LWP. Soil evaporation was also analyzed as a function of LAI, lysimeters location, systems and seasons. Lysimeter size effects on soil evaporation measurements were also tested. Analysis were carried out by using InfoStat software (Di Rienzo *et al.* 2014).

5.3. RESULTS

5.3.1. Climate

Total annual rainfall was 968 mm in 2012 being about 34% lower than the long-term mean annual rainfall of 1470 mm in that region. The 2012 dry season lasted from the

beginning of January until middle May with a maximum daily rainfall event of 16.8 mm and a total rainfall of 57.2 mm. The subsequent 2012 wet season provided lower than normal precipitation and was followed by the 2013 dry season which lasted almost six months, with only 23.5 mm rainfall overall. Total rainfall in 2013 was 1312 mm, 11% lower than the long term mean.

The reference evapotranspiration (ET₀) fluctuated with variations in meteorological factors which changed with the season pattern. Daily reference evapotranspiration, calculated with inputs from the automatic weather station installed in the FS plot, was similar between years (p=0.06) but differed between seasons (p<0.0001) with means 3.8 mm.d⁻¹ (S.E.=0.05) and 3.3 mm.d⁻¹ (S.E.=0.04) in the dry and wet seasons, respectively. Maximum reference evapotranspiration of 5.39 mm.d⁻¹ was achieved in 2012 dry season (Fig 5.1). Vapor pressure deficit (VPD) was found to be similar between FS and AFS (p=0.47). VPD did not vary between years (p=0.08) but differed with seasons (p<0.0001) (data not shown). VPD was 0.40 kPa (S.E.=0.01) and 0.78 kPa (S.E.=0.01) averaged in the wet and dry seasons, respectively. Minimum VPD was found in AFS in 2012 wet season with 0.01 kPa whilst maximum values reached 1.38 kPa in FS and 1.35 kPa in AFS in 2012 dry season.



Fig 5.1. Daily reference evapotranspiration in the dry and wet seasons over the period of the experiment.

5.3.2. Soil water content (SWC)

Mean SWC from three and six trenches in FS and AFS, respectively, averaged over the 2000 mm soil profile over the whole year, was greater in 2012 (731mm; S.E.=3.8) compared to 2013 (690mm; S.E.=3.9) (Fig 5.2). Comparing treatment effects, soil water content in the whole profile was almost always lower in AFS when compared to full sun coffee (p<0.05) except for a short period at the end of 2013 wet season when mean soil water content reached its maximum (1000 mm). The minimum value of SWC was recorded at the end of 2013 dry season when it reached 452 mm in AFS which represented 12% lower soil water than in FS during the same period (Fig 5.2).



Fig 5.2. Mean soil water content (from three profiles in FS and six profiles in AFS) in the whole soil profile (0-2.0 m) and rainfall over the period of the experiment.

5.3.3. Coffee and tree leaf area index

LAI of coffee plants was always greater in AFS compared to FS (p<0.001), being 2.39 (S.E.=0.10) and 3.57 (S.E.=0.10) in FS and AFS respectively. Coffee LAI seasonal patterns showed a strong decrease during the whole length of the dry periods and afterwards increasing in the wet seasons in both systems, although this increase was much delayed after the harsh dry period of 2013. Mean coffee LAI ranged from 2.88
(S.E.=0.05) to 5.01 (S.E.=0.07) in the dry and wet seasons respectively in shaded coffee while in FS it varied from 1.68 (S.E.=0.02) in the dry to 3.73 (S.E.=0.12) in the wet seasons (Fig 5.3). LAI was little influenced by plant defoliation due to the coffee rust epidemics caused by *Hemileia vastatrix* in Central America over the period of the experiment. Mean annual incidence of coffee rust measured by CATIE in the study area over the experiment was 1% and 8% in FS and AFS respectively in 2012 while in 2013 was 2% in both systems (data not published), thus the effect was considered negligible.



Fig 5.3. Coffee leaf area index in the dry (February – April) and wet seasons (July – November) in AFS and FS over 2012 and 2013. Vertical bars represent the standard error of the mean.

Tree LAI varied with shade tree species, ranging on average from 0.46 (S.E.=0.003) for *Simarouba glauca* to 0.70 (S.E.=0.02) for *Tabebuia rosea*. Tree LAI of both species also varied with the seasonal dynamics (p<0.001). In deciduous *Tabebuia rosea* LAI dropped to zero in April with mean LAI ranging between 0.13 (S.E.=0.10) in the dry to 1.12 (S.E.=0.09) in the wet seasons (Fig 5.4). In evergreen *Simarouba glauca* LAI remained more stable with mean of 0.44 (S.E.=0.004) in the dry while in the wet it was 0.48 (S.E.=0.01).



Fig 5.4. *Tabebuia rosea* and *Simarouba glauca* leaf area index in the dry (February – April) and wet seasons (July - November) in 2012 and 2013. Vertical bars represent the standard error of the mean.

5.3.4. Coffee and tree transpiration

Coffee transpiration on a leaf area basis differed between systems (p<0.001) and was often greater in FS (0.78 L.d⁻¹m⁻² S.E.=0.02) compared to AFS (0.60 L.d⁻¹m⁻² S.E.=0.02) averaged over the period of study. Seasonal pattern influences on coffee transpiration in FS and AFS were compared by using a generalized linear mixed model. Coffee transpiration was typically greater (p<0.001) in the dry periods (February and April) compared to the wet periods (July and November) in both systems likely due to greater evaporative demand in dry conditions. In AFS mean coffee transpiration rate varied from 0.44 L.d⁻¹m⁻² (S.E.=0.02) to 0.59 L.d⁻¹m⁻² (S.E.=0.02) and in FS from 0.56 L.d⁻¹m⁻² (S.E.=0.03) to 0.81 L.d⁻¹m⁻² (S.E.=0.03) in the wet and dry seasons respectively. Coffee transpiration was influenced by changes in water availability in the two years studied (p<0.001). Lower water supply had the effect of reducing coffee transpiration per unit leaf area to 0.53 L.d⁻¹m⁻² (S.E.= 0.02) in 2013 compared to 0.80 L.d⁻¹m⁻² (S.E.=0.02) in 2012 .

Over the time course of a day a comparison of coffee transpiration in both systems showed a tendency to an earlier and longer peak in AFS in dry conditions. In 2012 dry season in AFS coffee transpiration reached a peak at 10:00 that was then constant until 12:00 when it started to decline while in FS the peak was reached at 11:00 and

declined at 11:30 in 2012 dry season (Fig 5.5 a). In 2013 dry season coffee transpiration in AFS was stabilized around 11:30 until 13:30 while in FS transpiration declined rapidly after the peak (Fig 5.5 c). Mean coffee transpiration reached maximum values of 0.31 L.h⁻¹.m⁻² in FS and of 0.24 L.h⁻¹.m⁻² in AFS in 2012 dry season with VPD of 1.7 kPa and 1.3 kPa respectively. In the 2013 hard dry season although the highest values of 2.9 VPD were recorded, the maximum transpiration rate was reduced to 0.21 L.h⁻¹.m⁻² and 0.13 L.h⁻¹.m⁻² in FS and AFS respectively (Fig 5.5 c), probably due to the effect of low soil water availability. In the wet season coffee transpiration rate tended to diminish in both systems. In 2012 transpiration followed the trend of evaporative demand (1.23 kPa) (Fig 5.5 b) while in 2013 despite the similar VPD (1.21 kPa) transpiration was reduced probably by the lower soil water availability (Fig 5.5 d). Coffee transpiration maximum values reached 0.13 L.h⁻¹.m⁻² in FS and AFS in the 2012 wet season while in 2013 it was around 0.05 L.h⁻¹.m⁻² in both systems.









Fig 5.5. Typical diurnal trends in coffee transpiration on a leaf area basis from mean of four coffee trees each in FS and AFS over five consecutive days in the 2012 dry (a) and wet season (b) and 2013 dry (c) and wet season (d).

When up scaled to the plot level, coffee transpiration was greater in AFS likely due to the influence of 33% greater coffee LAI in the shaded environment (Fig 5.6). Mean coffee transpiration varied between 1.43 mm (S.E.=0.24) and 2.74 mm (S.E.=0.13) and between 1.32 mm (S.E.=0.25) and 1.34 mm (S.E.=0.14) in the dry and wet seasons in 2012 and 2013, respectively, in the FS plot while in AFS mean coffee transpiration ranged between 1.81 mm (S.E.=0.42) and 2.32 mm (S.E.=0.30) and between 1.65 mm (S.E.=0.15) and 1.80 mm (S.E.=0.24) in the dry and wet seasons in 2012 and 2013, respectively.



Fig 5.6. Plot level mean daily coffee transpiration (mm d⁻¹) in FS and AFS and reference evapotranspiration in the dry (Feb-April) and wet seasons (July-Nov) in 2012 and 2013. Bars represent the standard error of the mean.

Tree transpiration varied with shade tree species, seasonal pattern and environmental conditions. Tree transpiration rate differed between years (p<0.0001) with mean transpiration per tree of 45 L.d⁻¹ (S.E.=2.33) in 2012 and reduced to 26 L.d⁻¹ (S.E.=3.14) in 2013 as a probable effect of lower soil water content. Deciduous *Tabebuia rosea* transpiration was highly influenced by seasonal pattern when compared to evergreen *Simarouba glauca*. *Tabebuia rosea* daily transpiration ranged from 86 L.d⁻¹ (S.E.=5.08) in the wet season (July-November) to 31 L.d⁻¹ (S.E.=3.59) in the dry season (February-April) while *Simarouba glauca* displayed more constant water

consumption that varied little, from 25 L.d⁻¹ (S.E.=1.59) to 29 L.d⁻¹ (S.E.=1.26) in the wet and dry season respectively. Tree transpiration on a plot basis followed the reference evapotranspiration in the wet but not in the dry seasons likely due to the water availability influence (Fig 5.7).



Fig 5.7. Plot level mean tree transpiration and reference evapotranspiration in the dry (Feb-April) and wet seasons (July-Nov) in 2012 and 2013. Bars represent the standard error of the mean

Comparison between daily *Tabebuia rosea* and *Simarouba glauca* water consumption from mean transpiration of five consecutive days measured each 30 minutes showed that in the wet seasons tree transpiration tended to follow the atmospheric demand (Fig 5.8 b and 5.8 d). *Tabebuia rosea* reached its maximum transpiration rate with 12.9 L.h⁻¹ and 11.7 L.h⁻¹ while *Simarouba glauca* reached maximum 4.3 L.h⁻¹ and 3.2 L.h⁻¹ in 2012 and 2013 wet seasons when VPD ranged from 1.6 kPa and 0.9 kPa respectively. However, in the dry seasons despite of the greater VPD which reached between 2.1 kPa and 2.8 kPa, transpiration fell to 0.90 L.h⁻¹ and 0.60 L.h⁻¹ in *Tabebuia rosea* and to 3.84 L.h⁻¹ and 1.49 L.h⁻¹ in *Simarouba glauca* in 2012 and 2013 respectively (Fig 5.8 a and 5.8 c). As a deciduous tree species *Tabebuia rosea* daily transpiration showed great difference between both years and seasons while evergreen *Simarouba glauca* did not. At the plot level *Tabebuia rosea* mean daily transpiration rate varied between 0.30 mm.d⁻¹ (S.E.=0.05) and 1.03 mm.d⁻¹ (S.E.=0.05) in the dry and wet season, respectively, while *Simarouba glauca* transpiration did not change between seasons with an average of 0.20 mm.d⁻¹ (S.E.=0.01) (Fig 5.8).



Fig 5.8. Diurnal patterns of transpiration (L.h⁻¹) by *Tabebuia rosea* and *Simarouba glauca* trees and VPD (kPa) from mean of five consecutive days in the 2012 dry (a) and wet seasons (b) and in the 2013 dry (c) and wet seasons (d).

5.3.5. Soil evaporation

Evaporation from the soil surface was greater in the wet compared to the dry seasons with mean of 2.22 mm.d-1 (S.E.=0.06) and 0.58 mm.d-1 (S.E.=0.05) in the wet and dry seasons respectively (Fig 5.9).

In the dry seasons (April 2012 and March-April 2014) rainfall events were light and sparse and soil water content in the uppermost soil layer showed little variation in both

systems with mean of 0.38 m³.m⁻³ (S.E.=0.004). Soil surface evaporation varied from 0.53 mm.d⁻¹ (S.E.=0.04) to 0.45 mm.d⁻¹ (S.E.=0.04) in FS and AFS respectively, but, differences between systems were not significant (p=0.55). The influence of rainfall events on soil evaporation in the dry season was demonstrated on the slope variation comparing April 2012 with March-April 2014 drying events (Fig 5.9). Evaporation in April 2012 came after a large storm (19.6 mm) in the previous four days while by the middle of March-April 2014 scattered and small rains that occurred previously to the measurements did not have the same effect on soil evaporation rates. During the dry seasons great atmospheric demand had no influence on soil evaporation rate and no difference was found related to location of lysimeters in the row or interrow (p=0.15) (Fig 5.10). The ratio of actual to reference evaporation was 0.10 on average in the dry while in the wet periods it was 0.67. In wet conditions such as May-June in 2012 and May-November 2013 soil evaporation showed greater rates in FS compared to AFS (p=0.001). Mean soil evaporation was 2.50 mm d^{-1} and 1.98 mm d^{-1} in FS and AFS, respectively. In contrast to the dry monitoring periods, lysimeters location at the row or the interrow had a significant effect on soil evaporation in the humid period (p=0.01) (Fig 5.10).



Fig 5.9. Soil evaporation (mm.d⁻¹) from lysimeters in FS and AFS from April 2012 to April 2014. Vertical bars represent the standard error of the mean. Daily rainfall (right axis) and ETO (left axis) are indicated as references.

Irrespective of the systems mean daily evaporation ranged from 2.41 mm.d⁻¹ (S.E.= 0.11) to 2.02 mm.d⁻¹ (S.E.=0.12) in the interrow and row location, respectively (Fig 5.10). Soil evaporation over the whole two-year measurement period was calculated by fitting the Ritchie soil evaporation model (Ritchie 1972) to our measured data (Fig 5.11). The relationship between soil evaporation measured and modelled was linear with R^2 =0.62 and R^2 =0.44 in AFS and FS respectively (Fig 5.12 and 5.13). Simulations showed that water loss by soil evaporation was far from negligible and represented 44% and 12% of incident rainfall in wet and dry season, respectively.



Fig 5.10. Daily soil evaporation from lysimeters located in the row or interrow over the dry monitoring periods in 2012 and 2014 and over the 2013 wet period.



Fig 5.11. Daily soil evaporation rate measured and modelled by Ritchie soil evaporation model, in FS and AFS from April 2012 to April 2014. LAI data are required to compute Ritchie model; therefore, we could not perform the calculation between December 2013 and March 2014.



Fig 5.12. Soil evaporation measured and modelled in FS coffee. Regression equation: y=0.759x + 0.5385; $R^2=0.44$



Fig 5.13. Soil evaporation measured and modelled in AFS. Regression equation: y = 0.9804x + 0.3857; $R^2 = 0.62$

5.3.6. Coffee leaf water potential (LWP)

Predawn leaf water potential (PLWP) differed between years (p=0.0002) being greater in 2012 following the greater soil water availability. Mean PLWP was -0.20 MPa (S.E.=0.07) in 2012 while in 2013 reached an average of -0.57 MPa (S.E.=0.07). The influence of the available water on PLWP was demonstrated by the high correlation coefficient of 0.94 between PLWP and soil water content (SWC). In the dry periods mean PLWP was -0.74 MPa (S.E.=0.06) while in the wet seasons it was -0.12 MPa (S.E.=0.06) (Fig 5.14).

Midday leaf water potential (MLWP) was also influenced by the seasonal pattern (p<0.0001) with mean MLWP of -1.60 MPa (S.E.=0.06) and -0.64 MPa (S.E.=0.05) in the dry and wet seasons, respectively. By the end of 2013 severe dry season MLWP reached its lowest values with -2.04 MPa in FS and -2.33 MPa in AFS. MLWP tended to be more negative when vapor pressure deficit (VPD) was larger and the SWC was limited, which corresponded to high correlation coefficients of -0.90 and 0.97 between MLWP and VPD and between MLWP and SWC, respectively.



Fig 5.14. Mean coffee leaf water potential at predawn and midday from three days consecutive measurements in the dry (February and July) and wet seasons (July and November) in 2012 and 2013. Bars represent the standard error of the mean.

5.3.7. Total evapotranspiration

On average total evapotranspiration was greater in coffee agroforestry compared to full sun coffee (p=0.004) (Table 5.2). Mean evapotranspiration rate was 3.89 mm.d-1 (S.E.=0.28) and 2.80 mm.d-1 (S.E.=0.28) in AFS and FS respectively. With respect to seasonal effects, evapotranspiration was estimated for both systems in February (middle of dry season), April (end of the dry season), July (middle of wet season) and November (end of the wet season) in 2012 and 2013. Due to missing data on coffee transpiration in November 2012 we decided to exclude the total evapotranspiration estimation in that period (Table 5.2).

Table 5.2. Total evapotranspiration calculated from transpiration plus soil evaporation on a plot basis (with the standard error of the mean) and as percentage of total evapotranspiration in FS and AFS in February, April and July in 2012 and in February, April, July and November in 2013.

	Transpiration				Soil Evaporation				Evapotranspiration	
	FS		AFS		FS		AFS		FS	AFS
	(mm.d-1)	%	(mm.d-1)	%	(mm.d-1)	%	(mm.d-1)	%	(mm.d-1)	(mm.d-1)
Feb 2012	1.40 ± 0.22	68	1.38 ± 0.19	69	0.65 ± 0.22	32	0.63 ± 0.21	31	2.05 ± 0.44	2.01 ± 0.40
April 2012	1.46 ±0.26	88	2.55 ± 0.66	92	0.21 ± 0.003	12	0.22 ± 0.005	8	1.67 ± 0.27	2.77 ± 0.66
July 2012	2.74 ±0.13	58	3.76 ± 0.37	70	1.96 ± 0.27	42	1.63 ± 0.09	30	4.69 ± 0.39	5.39 ± 0.47
Feb 2013	1.68 ± 0.36	91	2.59 ± 0.23	94	0.18 ± 0.002	9	0.18 ± 0.002	6	1.86 ± 0.36	2.77 ± 0.23
April 2013	0.95 ± 0.14	89	1.22 ± 0.10	91	0.12 ± 0.0004	11	0.12 ± 0.0004	9	1.07 ± 0.14	1.34 ± 0.10
July 2013	0.75 ±0.10	25	1.65 ± 0.25	46	2.24 ± 0.22	75	1.96 ± 0.20	54	2.99 ± 0.31	3.61 ± 0.46
Nov 2013	1.93 ±0.18	49	3.73 ± 0.40	68	2.00 ± 0.18	51	1.79 ± 0.22	32	3.93 ± 0.36	5.52 ± 0.62

In the partitioning of evapotranspiration, transpiration was the most important contributor to water loss compared to soil evaporation in both systems. Transpiration accounted for 83% and 69% of evapotranspiration while soil evaporation represented 17% and 30% in AFS and FS respectively. Evaporation from the soil surface represented 50% and 33% of total evapotranspiration in the wet season while in the dry season it was reduced to 20% and 12% in FS and AFS respectively. Transpiration varied from 67% to 50% and from 88% to 80% of evapotranspiration in the wet and dry seasons in AFS and FS respectively.

Transpiration partitioning in AFS demonstrated that coffee transpiration was typically the greatest fraction compared to tree transpiration (Table 5.3). On average coffee transpiration comprised 75% of the total transpiration in AFS while *Tabebuia rosea* and *Simarouba glauca* each represented 17% and 8% of the total.

	Coffee		Tabebuia rosea		Simarouba glauca		AFS Transpiration
	mm.d ⁻¹	%	mm.d⁻¹	%	mm.d ⁻¹	%	mm.d ⁻¹
Feb 2012	1.38 ± 0.19	59	0.71 ± 0.03	30	0.24 ± 0.006	10	2.33 ± 0.22
April 2012	2.22 ± 0.65	87	0.08 ± 0.001	3	0.25 ± 0.005	10	2.55 ± 0.66
July 2012	2.32 ± 0.30	62	1.22 ± 0.06	32	0.22 ± 0.01	6	3.76 ± 0.37
Feb 2013	2.29 ± 0.21	88	0.12 ± 0.02	5	0.18 ± 0.003	7	2.59 ± 0.23
April 2013	1.01 ± 0.09	83	0.06 ± 0.004	5	0.14 ± 0.01	12	1.22 ± 0.10
July 2013	0.92 ± 0.17	56	0.57 ± 0.06	35	0.15 ± 0.02	9	1.65 ± 0.25
Nov 2013	2.67 ± 0.31	72	0.88 ± 0.08	23	0.17 ± 0.01	5	3.73 ± 0.40

Table 5.3. Transpiration partitioning in AFS with coffee, *Tabebuia rosea* and *Simarouba glauca* transpiration rate in mm.d⁻¹ and in percentage of the total transpiration in the system. Standard errors of the means are in brackets.

Also, coffee transpiration as a proportion of the total transpiration of the system tended to great values when *Tabebuia rosea* water requirements were low in the dry periods (April 2012 and February-April 2013) which characterized a complementarity relationship over time.

5.4. DISCUSSION

5.4.1. Coffee water use

Coffee water consumption on a leaf area basis was 23% greater in FS compared to AFS. On the other hand, at a plot scale we found coffee transpiration 15% greater in AFS likely due to 33% greater leaf area index in shaded coffee. Also, in the study area measurements by CATIE showed shade level between 53% and 67% of full irradiance over the period of experiment. Mean 60% of shade level had an effect on attenuation of stomatal limitation with greater coffee transpiration rates related to FS. This is consistent with other study in which comparison of different shade levels in optimal coffee growing condition in Costa Rica, in lower climatic stress condition than this study site, demonstrated that shade level of up to 55% was beneficial for coffee gas exchange maintenance at the leaf scale (Franck and Vaast 2009). Irrespective of shade level, the same trend of greater coffee transpiration rate in AFS was previously reported by Van Kanten and Vaast (2006) for coffee associated with timber tree species *Eucalyptus deglupta* or *Terminalia ivorensis* or with leguminous *Erythrina poeppigiana* when compared to FS system. Our results showing greater coffee water use on a leaf area basis in the open system was found to be similar to another study

on coffee shaded with *Inga densiflora* in Costa Rica in which FS coffee transpiration was about 20-45% greater than in AFS (Cannavo *et al.* 2011).

In this study coffee transpiration was driven by both the atmospheric demand and soil water availability. Contrasting precipitation in the two consecutive years of study allowed comparison of coffee transpiration behavior in both years. In 2012, when soil water was not so limiting, coffee transpiration tended to follow air saturation deficit in the wet and dry season as demonstrated in Figure 5.5. Restriction of coffee transpiration was observed under VPD values of 1.7 kPa in FS in the dry season, probably, as a mechanism to prevent water stress. The close relationship between transpiration and atmospheric parameters has been previously demonstrated (Fanjul *et al.* 1985; Gutiérrez and Meinzer 1994) and coffee transpiration inhibition at a similar threshold of VPD = 1.5 - 1.6 kPa was reported by Van Kanten and Vaast (2006) and by Gutiérrez and Meinzer (1994). The general independence of coffee leaf transpiration from soil moisture was demonstrated when a decrease in transpiration rate was recorded only when 80% of the soil water in the rooting zone had been depleted (Nunes and Duarte 1969).

We demonstrated that in the second year, over the severe dry season, lower transpiration rate occurred despite the high solar radiation (1015 W.m⁻²) and high vapor pressure deficit (2.9 kPa) which suggest a response to low soil moisture that seems to have become the predominant limiting factor of transpiration. We also demonstrated that in the 2013 hard dry season, coffee leaf water potential declined to its lowest level when reached -1.94 MPa and -2.33 MPa at midday in FS and AFS, respectively. The same order of magnitude of -2.55 MPa was found for zero turgor in *Coffea arabica* cultivar reported by Meinzer *et al.* (1990) in drought conditions. The lowest levels of leaf water potential and decline of coffee water use found in AFS during the restrictive soil water condition in 2013 dry season was associated with high level of water stress in coffee plants. This result indicated competition for water between coffee and shade trees in such environmental conditions.

5.4.2. Shade tree water consumption

In the agroforestry system most water use was due to coffee plants rather than shade trees. This was probably a consequence of the greater coffee LAI and coffee density compared to the trees. Coffee water use represented 75% of the whole water

transpired in AFS while deciduous Tabebuia rosea shade trees accounted for 15% and evergreen Simarouba glauca for 8%. The Tabebuia rosea water consumption pattern was determined by leaf phenology, soil water availability and environmental conditions. The positive and strong correlation between LAI and transpiration rate ($R^2=0.80$) reinforced the effect of leaf phenology on *Tabebuia rosea* water consumption patterns. Despite of lower transpiration in the dry periods (February-April) Tabebuia rosea transpiration on a plot basis reached mean 0.30 mm.d⁻¹ being even greater (p=0.02) than *Simarouba glauca* transpiration that was 0.19 mm.d⁻¹ averaged in the period. Very low rates were observed in April when most of trees were leafless. But, Tabebuia rosea remained completely leafless for only short periods of about 2-3 weeks in April 2012 and 5-8 weeks during the 2013 severe dry season. Potential reduction in water loss over the dry season was compensated for by greater water consumption in the late dry seasons to achieve full leaf expansion which characterized Tabebuia rosea as a water spender compared to Simarouba glauca tree. Tabebuia rosea daily transpiration rates varied from 1.22 mm.d⁻¹ (S.E.=0.06) to 0.45 mm.d⁻¹ (S.E.=0.04) and from 0.67 mm.d⁻¹ (S.E.=0.10) to 0.09 mm.d⁻¹ (S.E.=0.01) in the wet and dry seasons in 2012 and 2013 respectively. Moreover, the deciduous Tabebuia rosea root system randomly distributed in the soil profile, as presented in Chapter III, indicated possible competition for water in the dry periods in the upper soil layer where most coffee roots are concentrated. This study demonstrated that such a competitive relationship was minimized during the "normal" dry periods because most Tabebuia rosea water requirements occurred in the wet seasons and did not coincide with the greatest periods of coffee water consumption that was in the dry. Simultaneous periods of great water requirements by deciduous Tabebuia rosea and of low water use by coffee plants suggested a complementarity in water use over time between coffee and shade tree. However, competition between coffee and shade tree was observed in periods of low water input supply as the 2013 severe dry season.

Conversely, evergreen *Simarouba glauca* may be considered as a water conserver with lower and more stable water consumption pattern over the course of the experiment compared to *Tabebuia rosea*. Exception was found in lower *Simarouba glauca* water use in 2013 severe dry season when maximum transpiration rate declined to 1.49 L h⁻¹ compared to 3.84 L h⁻¹ in 2012 dry season. Overall, mean *Simarouba glauca* daily transpiration rate ranged from 0.19 mm.d⁻¹ (S.E.=0.01) to 0.22 mm.d⁻¹ (S.E.=0.01) in the wet and dry season respectively. Although the seasonal differences in *Simarouba*

glauca transpiration rate were not statistically significant, previous studies showed the tendency of increasing transpiration rates as the dry season progressed in evergreen timber trees such as *Eucalyptus tetrodonta* and *Eucalyptus miniata* (Grady *et al.* 1999). Similar findings were reported for *Acacia mangium* in Panama likely as a response to the exploration of deep sources of soil water (Kunert *et al.* 2010). *Simarouba glauca* was characterized by denser and concentrated root system in deep soil layers (below 1.10 m depth) with a clear root niche differentiation to coffee roots as reported in Chapter III. This description of evergreen *Simarouba glauca* water use pattern and spatial below ground arrangements is supported by findings of Meinzer et al. (1999) in which species with small seasonal variability in leaf fall, was able to exploit deeper soil layers with increasing drought condition. Also it is worth to recall that in this investigation, *Tabebuia rosea* and *Simarouba glauca* water uptake and consumption must have been influenced by being limited to no more than 2.0 m soil depth exploration, below which point no roots were found.

5.4.3. Soil surface evaporation

This study demonstrated that shade trees had an effect on reducing water loss from soil surface evaporation. Shade trees were responsible for a decrease of 31% in soil evaporation compared to the open system. The same tendency of reduction on evaporation rate in the shade was demonstrated in a sub humid climate in Kenya in which soil evaporation in agroforestry was reduced by 35% when compared to bare soil (Wallace et al. 1999). Another study in a Grevillea robusta agroforestry system in Kenya showed that beneath shade tree soil evaporation was reduced to 39% of the rainfall compared to 55% without any canopy (Wallace et al. 1997). In coffee agroforestry, comparison between different canopy cover levels showed a reduction in soil surface evaporation with increasing canopy cover (Lin 2010). However, literature on water loss measurements by soil evaporation in coffee agroforestry system is sparse thereby any comparison with other studies was limited. Soil water evaporation takes place from where most coffee fine roots occur with a potential effect on coffee water use which is of considerable importance, especially in dry environments (Beer 1987). This study demonstrated that in dry periods despite the great potential evaporation of 4.7 mm low mean soil evaporation was observed (from 0.25 mm.d⁻¹ to 0.38 mm.d⁻¹). These rates were similar to findings of Wallace (1991) in arid lands where the evaporation rate of 0.5 mm.d⁻¹ was much less than potential evaporation of 3.8 mm. In wet condition greater evaporation from soil surface compared to the dry periods was also reported by Zhang et al 2008 and by Yunusa *et al.* (2004). We demonstrated that in the environmental condition studied soil water evaporation was far from negligible. Water loss by soil evaporation varied from 0.31 mm d⁻¹ (S.E.=0.02) to 1.76 mm d⁻¹ (S.E.=0.03) while coffee water use by transpiration ranged from 1.59 mm.d⁻¹ (S.E.=0.05) to 2.49 mm.d⁻¹ (S.E.=0.09) in the dry and wet seasons respectively. Similar orders of magnitude for coffee transpiration were reported by Van Kanten and Vaast (2006) however simultaneous measurements of soil evaporation are rare.

5.4.4. Total evapotranspiration

Total evapotranspiration was greater in AFS compared to FS as reported previously in coffee with Inga densiflora in optimal condition for coffee cultivation in Costa Rica (Cannavo et al. 2011). Evapotranspiration partitioning into plant transpiration and soil surface evaporation demonstrated that transpiration comprised the most important component in both systems. Transpiration represented 69% and 83% of the total evapotranspiration in FS and AFS respectively. Studies in a coffee orchard in Brazil showed similar proportions of transpiration to evapotranspiration of about 65% (Flumignan et al. 2011). Plant transpiration as a dominant source for total evapotranspiration was reported in other studies (Yepez et al. 2003; Williams et al. 2004; Xu et al. 2008) however in coffee agroforestry information is lacking and comparisons are limited. Our results demonstrated that in agroforestry most water consumption was due to coffee and not by shade trees. The effect of shade trees water use did not result in water stress for coffee cultivation until the 2013 severe dry season occurred. In the 2013 water limiting conditions coffee transpiration was reduced to 54% and 35% in AFS and FS respectively compared to the dry season in the previous year. This study showed the effect of shade trees on diminishing water loss by evaporation, particularly during the wet periods. This may result in more water being available for transpiration that potentially could be used for coffee growth and to improve yield, which is quite important in a sub-optimal environment such as at Masatepe. In drought condition evaporation from the soil surface was at its minimal rate of 0.04% to the total evapotranspiration and similar in FS and AFS. The same tendency of small or negligible fraction of soil evaporation to the evapotranspiration in drought condition was similarly reported in arid and semiarid environments (Williams et al. 2004) due to breakdown of soil hydraulic conductivity at the surface.

5.5. CONCLUSIONS

This study demonstrated that in sub optimal conditions for coffee cultivation agroforestry was a better water user when compared to a non-shaded coffee system since most of soil water was used for coffee transpiration and not by shade trees or lost by evaporation from the soil surface.

Our results indicate that even in these sub optimal environmental conditions soil water was not usually a constraint for coffee water consumption in agroforestry. Complementarity in water use over time was demonstrated between coffee and *Tabebuia rosea* whilst complementarity in root system distribution and soil water uptake was observed between coffee and *Simarouba glauca* tree. This supports the idea of complementarity of coffee and shade trees over most of the period of study.

Competition in water use between coffee and shade trees was observed in a severe dry season when water input supply was not enough to avoid coffee water stress in agroforestry due to coffee plus shade tree water requirements.

Evergreen *Simarouba glauca* characteristics such as taking up water from deeper soil layers and the lower and more constant water consumption pattern seemed to be more suitable as coffee shade tree when compared to deciduous *Tabebuia rosea* in sub optimal conditions for coffee cultivation.

5.6. REFERENCES

- Allen, R.G.; Pereira, L.S.; Raes, D.; Smith, M. 1998. Crop evaporation: guidelines for computing crop water requirements. Rome, FAO. 300 p. (FAO Irrigation and Drainage)
- Barradas, V.L.; Fanjul, L. 1986. Microclimatic characterization of shaded and opengrown coffee (*Coffea arabica*) plantations in Mexico. Agricultural and Forest Meteorology 38: 101-112.
- Beer, J. 1987. Advantages, disadvantages and desirable characteristics of shade trees for coffee, cacao and tea. Agroforestry Systems 5: 3-13.
- Beer, J. 1995. Efectos de los arboles de sombra sobre la sostenibilidad del cafetal. Boletin Promecafe 68: 13-18.

- Cannavo, P.; Sansoulet, J.; Harmand, J.M.; Siles, P.; Dreyer, E.; Vaast, P. 2011. Agroforestry associating coffee and *Inga densiflora* results in complementarity for water uptake and decreases deep drainage in Costa Rica. Agriculture, Ecosystems & Environment 140(1-2): 1-13.
- Daamen, C.C.; Simmonds, L.P.; Sivakumar, M.V.K. 1995. The impact of sparse millet crops on evaporation from soil in semi- arid Niger. Agricultural Water Manage 27(3): 225-242.
- Di Rienzo, J.; Casanoves, F.; Balzarini, M.; Gonzalez, L.; Tablada, M.; Robledo, C. 2014. InfoStat Manual. Argentina., Universidad Nacional de Córdoba. 336 p.
- Fanjul, L.; Arreola-Rodriguez, R.; Mendez-Castrejon, M.P. 1985. Stomatal Responses to Environmental Variables in Shade and Sun Grown Coffee Plants in Mexico. Experimental Agriculture 21 (3): 249-258.
- Flumignan, D.L.; Faria, R.T.; Prete, C.E.C. 2011. Evapotranspiration components and dual crop coefficients of coffee trees during crop production. Agricultural Water Management 98: 791-800.
- Franck, N.; Vaast, P. 2009. Limitations of coffee leaf photosyntesis by stomatal conductance and light availability under different shade levels. Trees 23: 761-769.
- Gomez-Delgado, F.; Roupsard, O.; Moussa, R.; Le Maire, G.; Taugourdeau, S.; Bonnefond, J.M.; Perez, A.; Van Oijen, M.; Vaast, P.; Rapidel, B.; Voltz, M.; Imbach, P.; Harmand, J.M. 2010. Modelling the hydrological behaviour of a coffee agroforestry basin in Costa Rica. Hydrology and Earth System Sciences 7: 3015-3071.
- Grady, A.P.O.; Eamus, D.; Hutley, L.B. 1999. Transpiration increases during the dry season: patterns of tree water use in eucalypt open-forests of northern Australia. Tree physiology 19: 591-597.
- Granier, A. 1985. Une nouvelle méthode pour la mesure du flux de seve brute dans le tronc des arbres. Annales des Sciences Forestieres 42: 193-200.
- Granier, A. 1987. Evaluation of transpiration in a Douglas-fir stand by means of sap flow measurements. Tree physiology 3: 309-320.
- Gutiérrez, M.V.; Meinzer, F.C. 1994. Estimating water use and irrigation requirements of coffee in Hawaii. Journal of the American Society for Horticultural Science 119(3): 652-657.
- Haggar, J.; Barrios, M.; Bolaños, M.; Merlo, M.; Moraga, P.; Munguia, R.; Ponce, A.; Romero, S.; Soto, G.; Staver, C.; de M. F. Virginio, E. 2011. Coffee

agroecosystem performance under full sun, shade, conventional and organic management regimes in Central America. Agroforestry Systems 82(3): 285-301.

- Jackson, N.A.; Wallace, J.S. 1999. Soil evaporation measurements in an agroforestry system in Kenya. Agricultural and Forest Meteorology 94: 203-215.
- Kool, D.; Agam, N.; Lazarovitch, N.; Heitman, J.L.; Sauer, T.J.; Ben-Gal, A. 2014. A review of approaches for evapotranspiration partitioning. Agricultural and Forest Meteorology 184: 56- 70.
- Kunert, N.; Schwendenmann, L.; Ischer, D.H. 2010 Seasonal dynamics of tree sap flux and water use in nine species in Panamanian forest plantations. Agricultural and Forest Meteorology 150: 411-419.
- Lawrence, D.M.; Thornton, P.E.; Oleson, K.W.; Bonan, G.B. 2006. The Partitioning of Evapotranspiration into Transpiration, Soil Evaporation, and Canopy Evaporation in a GCM: Impacts on Land–Atmosphere Interaction. Journal of hydrometeorology 8: 862-880.
- Lin, B.B. 2007. Agroforestry management as an adaptive strategy against potential microclimate extremes in coffee agriculture. Agricultural and Forest Meteorology 144: 85-94.
- Lin, B.B. 2010. The role of agroforestry in reducing water loss through soil evaporation and crop transpiration in coffee agroecosystems. Agricultural and Forest Meteorology 150: 510-518.
- Liu, C.; Zhang, X.; Zhang, Y. 2002. Determination of daily evaporation and evapotranspiration of winter wheat and maize by large-scale weighing lysimeter and micro-lysimeter. Agricultural and Forest Meteorology 111: 109-120.
- Meinzer, F.C.; Grantz, D.A.; Goldstein, G.; Saliendra, N.Z. 1990. Leaf water relations and maintenance of gas exchange in coffee cultivars grown in drying soil. Plant physiology 94: 1781-1787.
- Muschler, R.G. 1997. Effectos de sombra de Erythrina poeppigiana sobre Coffea arabica vars. Caturra y Catimor. 18º Simposio Latinoamericano de Caficultura. San Jose, Costa Rica, 157-162 p.
- Nunes, M.A.; Duarte, N.T. 1969. Resistencia a secura em *Coffea arabica*. Estudo comparativo de duas seleções. Garcia de Orta, Lisboa 17(1): 83-96.

Penman, H.L. 1963. Vegetation and hydrology. Commonwealth Burg. of soils 53: 124.

Ritchie, J. T. 1971. Dryland Evaporative Flux in a Subhumid Climate: I. Micrometeorological Influences. Agronomy Journal 63(1): 51-55.

- Ritchie, J. T. 1972. Model for predicting evaporation from a row crop with incomplete cover. Water Resource Research 8(5): 1204-1213.
- Scholander, P.F.; Hammel, H.T.; Bradstreet, E.D.; Hemmingsen, E.A. 1965. Sap pressure in vascular plants. Science 148: 339-346.
- Siles, P.; Harmand, J.M.; Vaast, P. 2009. Effects of Inga densiflora on the microclimate of coffee (*Coffea arabica* L.) and overall biomass under optimal growing conditions in Costa Rica. Agroforestry Systems 78(3): 269-286.
- Smith, D.M.; Allen, S.J. 1996. Measurement of sap flow in plant stems. Journal of Experimental Botany 47(305): 1833-1844.
- Udawatta, R.P.; Anderson, S.H.; Motavalli, P.P.; Garrett, H.E. 2011. Calibration of a water content reflectometer and soil water dynamics for an agroforestry practice. Agroforestry Systems 82: 61-75.
- Van Kanten, R.; Vaast, P. 2006. Transpiration of arabica coffee and associated shade tree species in sub-optimal, low-altitude conditions of Costa Rica. Agroforestry Systems 67: 187-202.
- Vertessy, R.A.; Benyon, R.G.; O'Sullivan, S.K.; Gribben, P.R. 1995. Relationships between stem diameter, sapwood area, leaf area and transpiration in a young mountain ash forest. Tree physiology 15: 559-567.
- Vogel, A.W.; Acuña Espinales, E. 1995. Reference soils of the Pacific Coastal Plain with a hardpan (Talpetate). International Soil Reference and Information Centre and Universidad Nacional Agraria. 21 p. (Soil brief Nicaragua) (3).
- Wallace, J.S. 1991. The measurement and modelling of evaporation from semi-arid land. *In* Sivakumar, M.V.K.; Wallace, J.S.; Renard, C.; Giroux, C. eds. 1991. Soil Water balance in the Sudano-Sahelian one Wallingford, UK, IAHS. p. 131-148.
- Wallace, J.S.; A., J.N.; Ong, C.K. 1997. Soil evaporation in a Grevillea robusta agroforestry system. Modelling the effects of tree canopy shade. Agricultural and Forest Meteorology.
- Wallace, J.S.; Jackson, N.A.; Ong, C.K. 1999. Modelling soil evaporation in an agroforestry system in Kenya. Agricultural and Forest Meteorology 94(189-202).
- Wilcox, B.P.; Breshears, D.D.; Seyfried, M.S. 2003. Water balance on rangeland. *In* Howell, B.A.S.a.T.A. ed. 2003. Encyclopedia of water science. New York, Marcel Dekker, Inc. p. 791-794.
- Williams, D.G.; Cable, W.; Hultine, K.; Hoedjes, J.C.B.; Yepez, E.A.; Simonneaux, V.;
 Er-Raki, S.; Boulet, G.; de Bruin, H.A.R.; Chehbouni, A.; Hartogensis, O.K.;
 Timouk, F. 2004. Evapotranspiration components determined by stable isotope,

sap flow and eddy covariance techniques. Agricultural and Forest Meteorology 125: 241-258.

- Xu, Z.; Yang, H.; Liu, F.; An, S.; Cui, J.; Wang, Z.; Liu, S. 2008. Partitioning evapotranspiration flux components in a subalpine shrubland based on stable isotopic measurements. Botanical Studies 49: 351-361.
- Yepez, E.A.; Williams, D.G.; Scott, R.L.; G., L. 2003. Partitioning overstory and understory evapotranspiration in a semiarid savanna woodland from the isotopic composition of water vapor. Agricultural and Forest Meteorology 119(1-2): 53-68.
- Yunusa, I.A.M.; Walker, R.R.; Lu, P. 2004. Evapotranspiration components from energy balance, sapflow and microlysimetry techniques for an irrigated vineyard in Inland Australia. Agricultural and Forest Meteorology 127: 93-107.

CHAPTER VI

GENERAL DISCUSSION AND CONCLUSIONS

6.1. Study site and experimental design

We had the opportunity to study a mature coffee agroforestry experiment established in 2000 in sub optimal conditions for coffee cultivation in Masatepe, Nicaragua. Such sub optimal conditions consisted of:

- Mean annual temperature of 27°C that is above *Coffea arabica* comfort zone and 455 m.a.s.l. which is considered to be rather low for *Coffea arabica* cultivation at this latitude;
- Mean annual rainfall of 1470 mm with a 6 month long dry season (which on average receives only 9% of the annual rainfall). Over the two years of study the contrasting mean annual precipitations of 968 mm in 2012 and 1312 mm in 2013 provided the chance to assess water dynamics and use in a normal and an abnormally dry conditions;
- Soil accessible to roots was limited to 2 meters depth, after which a pan constituted by coarse volcanic ejecta impeded root penetration. Additionally, a hardened layer (talpatete) crossed this accessible two-meter layer and further impaired root growth.

The original experiment consisted of three replicates of all treatments, as described by Haggar *et al.* (2011). Nevertheless, for the measurement of hydrological variables in this study, we selected only one full sun coffee plot, due to the poor establishment of coffee plants and problems with small floods in the other two replications. Thus, our experimental design consisted of a full sun monocrop coffee (FS) plot (1440 m²) and an adjacent coffee agroforestry system (AFS) plot (3200 m²) in which pseudo replications were located. In the coffee agroforestry system plot *Coffea arabica* was associated with *Simarouba glauca* DC. (Simaroubaceae) and *Tabebuia rosea* (Bertol.) DC. (Bignoniaceae) planted as shade trees. Tree spacing was originally 4 m x 4 m with 625 trees ha⁻¹ alternating both species in all rows, but tree density had been reduced over time in order to achieve an agronomically appropriate level of shade under mature trees. Tree spacing was 8 m x 8 m over the period of the experiment. *Tabebuia rosea* mean density was 113 trees ha⁻¹ and that of *Simarouba glauca* was 75 trees ha⁻¹.

Coffee density throughout the experiment was 4000 plants ha⁻¹, spacing being 2 m between rows and 1.25 m between plants. Coffee plants were pruned periodically in accordance with standard agronomic practice.

The root study experimental design consisted of a full sun monocrop coffee (FS) plot and two adjacent coffee agroforestry system (AFS) plots in which data were collected in 2011 and 2013. Evapotranspiration and rainfall interception studies were only done in one agroforestry plot due to equipment limitations, and we utilized pseudoreplications in both the full sun and the agroforestry plots. This design was considered appropriate for this process-based study.

For the evapotranspiration study the experimental design allowed us to assess the three hypothesized levels of competition for water in AFS compared to full sun coffee: coffee near *Simarouba glauca* trees, coffee near *Tabebuia rosea* trees and shaded coffee as far as possible from both tree species. Interception losses in trees and coffee bushes were studied in three sub plots established in the agroforestry system each one bounded by two *Simarouba glauca* trees and two *Tabebuia rosea* trees and including the coffee plants within. They were compared to interception losses in coffee bushes in the full sun treatment plot.

6.2. Complementarity and competition for water in coffee shaded with *Tabebuia rosea* and *Simarouba glauca* in sub optimal environmental condition

6.2.1 Below-ground interactions

One of the main goals of coffee and tree combination in agroforestry is to make better use of available soil water and improve overall productivity. However, in water-limiting conditions, the potential competition for water may impair the effectiveness of the agroforestry system. Competition can be avoided when coffee and tree root systems are able to exploit water from distinct soil horizons (Cannell *et al.* 1996; Ong *et al.* 1996). In this study most of the coffee roots were concentrated in the upper 30 cm of the soil profile and a minority reached a maximum depth of 150 cm and 170 cm depth in FS and AFS respectively. Coffee root growth was similar between systems which indicated little effect of shade tree root system on enhancing coffee root growth, as hypothesized. Also, in this investigation shallower coffee roots and deeper *Simarouba glauca* root distribution patterns suggested potential for complementarity in soil water use throughout the soil profile. Although both shade tree species showed greater root

density below 110 cm depth Simarouba glauca and Tabebuia rosea root systems behaved differently. Tabebuia rosea roots were present in the whole soil profile while Simarouba glauca roots were denser and concentrated in the deeper layers. Coffee and Simarouba glauca root system exhibited a clear niche differentiation whereby soil water was likely to be extracted from different strata. However, deep rooted trees and spatial difference in coffee and tree roots distribution by themselves are not enough to guarantee complementarity of resources use. A deep source of water is required depending on water availability and soil condition. Moreover, species effects could have influenced coffee and tree root systems partitioning. In a mature system with such coffee and shade tree spatial distribution as at the study site, an effect of the root system of the dominant species over the other may occur. In a mixed shade tree species this domination relationship may be observed between both shade tree species and or between shade tree and coffee roots. Vertical distribution of coffee and eucalyptus root systems in agroforestry studied by Schaller et al. (2003) demonstrated that coffee roots were competitive enough to displace the shade tree roots. However, in order to check species-specific characteristics of the shade trees utilized in our study a pure stand of Simarouba glauca and also of Tabebuia rosea in the same environmental conditions would be necessary.

In this particular study site, soil is characterized by the presence of a hardened layer of volcanic ash locally known as talpetate. This layer was extremely variable ranging from soft, weathered material containing some harder rock fragments to a fairly continuous hard layer with rock-like properties. It also varied in depth below the soil surface and in thickness of the layer. Although there was some influence of this hardened layer in restricting water movement, fractures enabled water and roots to cross the talpetate and infiltration to the deeper strata occurred. However, root growth, and soil water uptake was limited to two meters depth due to a much denser layer of compacted volcanic ash at that depth. The water stored in the soil was affected by 34 % and 11% lower rainfall in 2012 and 2013 respectively compared to the long-term mean annual rainfall of 1470 mm in that region which is considered low for coffee cultivation. Extended dry seasons lasted up to six months with 57.2 mm and 23 mm rainfall overall in 2012 and 2013 respectively, during which coffee and trees depended on stored soil water only. Limitations in both the depth of rooting zone and the soil water availability had great effect on coffee and tree water use over these prolonged dry periods. In a deeper soil, tree roots would normally go further down in order to get water from deep layers. Conversely, the time course of soil water content showed that in the dry periods water uptake tended to decrease earlier in AFS when compared to FS owing to the effect of shade trees. In 2013, during the very dry season, the coffee and tree water requirements were greater than the soil water stock in the 2 m of soil profile, which was insufficient to avoid the high level of water stress suffered by coffee.

6.2.2 Coffee water use

In the sub optimal environmental conditions studied coffee water use on a leaf area basis was 25% greater in FS than in AFS. As a LAI-dependent process (Taugourdeau et al. 2014), transpiration was found to be influenced by the seasonal variation being smaller when coffee was fully leaved and greater when leaf shedding occurring, over the dry periods. Other studies have shown that transpiration on a leaf area basis tend to increase in response to defoliation (Meinzer and Grantz 1991; Tausend et al. 2000). In the dry periods, fewer leaves distributed in the coffee plant crown provided low shade cast by neighbouring leaves and therefore, high solar radiation and temperature with humidity alteration at the leaf surface may occur, which favoured greater leaf transpiration compared to the wet periods. In the shaded environment high radiation and temperature were minimized which resulted in a lower leaf to air water vapour pressure deficit and therefore, lower leaf transpiration rates compared to FS. Greater coffee LAI in the shade did not result straightforwardly in greater leaf-level transpiration rate likely due to leaves overlapping in the dense coffee crown with low leaf surface and due to low photosynthetically active radiation under shade tree canopies. Moreover, as the leaf transpires, water vapour tends to humidify the air nearby the leaves reducing evaporative demand and transpiration rates.

However, coffee water consumption on a land area basis was greater in AFS compared to FS due to higher vegetative growth of coffee plants in shaded environment with 33% greater coffee LAI in agroforestry. The same trend of greater coffee LAI and transpiration per hectare in AFS was demonstrated by Van Kanten and Vaast (2006) in sub optimal environmental conditions in Costa Rica in which coffee was associated with different shade tree species (*Erythrina poeppigiana, Eucalyptus deglupta* or *Terminalia ivorensis*).

Coffee water use could be influenced by plant defoliation due to the coffee rust epidemics caused by *Hemileia vastatrix* that occurred in Central America in the period of the experiment. However, mean annual incidence of coffee rust measured by CATIE in the study area over the experiment was 1% and 8% in FS and AFS respectively in 2012 while in 2013 was 2% in both systems (data not published). But those rates may be considered low compared with the disease incidence of 32% in Nicaragua in the same period (Avelino *et al.* 2015). The overall low incidence may be related to agronomic practices and the monitoring routine in the study site that allows early detection and control of coffee rust. Higher coffee rust incidence in shade may be explained by buffering effect on temperature and radiation with increasing leaf wetness which favours the *Hemileia vastatrix* infection (Avelino *et al.* 2004). Another study also showed worse coffee rust occurrence in shade system (Staver *et al.* 2001). Despite the management influence, climate anomalies were found to be the most important factor of influence of the disease in the region over the period of study (Avelino *et al.* 2015). Overall, the climate conditions also seemed to be the most influence on coffee water use in the period of study with little effect from the coffee rust in the experiment results.

In this study, the influence of soil water availability in coffee transpiration was demonstrated by the contrast precipitation pattern in the two consecutive years of the experiment. In the first year, coffee transpiration tended to follow air saturation deficit in both the wet and dry seasons. This close relationship between transpiration and atmospheric parameters had been previously demonstrated (Fanjul et al. 1985; Gutiérrez et al. 1994b). In contrast, in the 2013 severe dry season lower transpiration rates occurred despite the high radiation (1015 W.m⁻²) and high vapour pressure deficit (2.9 kPa) as a response to very low plant available soil water that became the predominant limiting factor of transpiration. The high level of water stress was indicated by coffee leaf water potential that reached -1.94 MPa at midday in FS and -2.33 MPa in AFS. The general independence of coffee transpiration on soil water availability was demonstrated by Nunes and Duarte (1969) thereby our results suggest that the lower levels of transpiration in the harsh dry period may be associated with a high level of soil water depletion that induced water stress of coffee plants. Also, the lowest levels of leaf water potential and decline of coffee water use found in AFS in the 2013 restrictive soil water conditions indicated possible competition for water between coffee and shade trees in those unusual circumstances.

6.2.3 Shade tree water use

We demonstrated greater water consumption in agroforestry compared to coffee monoculture. However, the highest contribution to water use in agroforestry was due to coffee rather than shade trees. Deciduous Tabebuia rosea shade tree accounted for 17% and evergreen Simarouba glauca for 8% of the whole water transpired in the system. In the study site deciduous Tabebuia rosea water use pattern was mainly controlled by the degree of foliation ranging from 1.22 mm.d⁻¹ (S.E.=0.06) to 0.45 $mm.d^{-1}$ (S.E.=0.04) and from 0.67 $mm.d^{-1}$ (S.E.=0.10) to 0.09 $mm.d^{-1}$ (S.E.=0.01) in the wet and dry seasons in 2012 and 2013, respectively. A strong and positive relationship (R²=0.80) between Tabebuia rosea mean transpiration rate and LAI reinforced the conclusion that the water consumption pattern was affected by phenology. Deciduous tree species may be advantageous as coffee shade tree providing further light availability during dry periods and therefore, enhancing floral induction with effects on increasing fruit load (Beer et al. 1998). However, potential benefits from light inputs and reduction in the water use by *Tabebuia rosea* over the dry periods was overcome by the great water consumption in the late dry seasons to achieve full canopy expansion. Tabebuia rosea may be characterized as a water spender when compared to evergreen Simarouba glauca. Also, this study demonstrated that most water consumption by Tabebuia rosea occurred over the wet seasons which did not coincide with great water use by coffee that was during the dry periods. In this case, deciduousness favoured a complementarity in water use over time between coffee and shade tree. However, in a limited water supply condition as the severe dry season in the study site, deciduousness represented a competitive factor due to great shade tree water use which accentuated the low soil water available for coffee. This may constitute a good reason for deciduous Tabebuia rosea not being a good choice as coffee shade tree in sub optimal environmental condition for coffee growth. Water requirements by *Tabebuia rosea* that was determined by leaf phenology, soil water availability and environmental conditions contrasted with the tendency for lower and more stable water consumption by Simarouba glauca over the course of the experiment.

Evergreen *Simarouba glauca* water use was similar throughout the period of study. Mean *Simarouba glauca* daily transpiration rate ranged from 0.19 mm.d⁻¹ (S.E.=0.01) to 0.22 mm.d⁻¹ (S.E.=0.01) in the wet and dry season respectively likely due to the influence of greater evaporative demand over the dry period. Another study showed the tendency of increasing transpiration rates as the dry season progressed in evergreen timber trees such as *Eucalyptus tetrodonta* and *Eucalyptus miniata* (Grady et al. 1999). Simarouba glauca was characterized by a denser and more concentrated root system in deep soil layers (below 110 cm depth) with a clear root niche differentiation from coffee roots as we demonstrated in Chapter III. Most water uptake from deep soil layers was reinforced by the lower soil water content below 0.4 m of the soil profile found near Simarouba glauca when compared to Tabebuia rosea. Also, evergreen Simarouba glauca water use pattern and spatial belowground arrangement are supported by findings of Meinzer et al. (1999) in which species with small seasonal variability in leaf fall were able to exploit deeper soil layers with increasing drought conditions. The influence of drought on water use by evergreen timber tree as a coffee shade tree was reported in Costa Rica regarding E. deglupta (Jimenez and Alfaro 1999). These authors showed rapid water depletion and great competition with coffee in water limiting condition over a five month dry season while another study in high precipitation conditions showed no competition for water between coffee and trees (Schaller et al. 2003). Our study site, located in conditions where water can be limiting to coffee growth is clearly similar to the first case, where competition for soil water between coffee and shade trees was demonstrated during the drier year.

6.2.4. Shade tree effects on coffee production in sub optimal environmental condition for coffee cultivation

We did not measure coffee production in this experiment. Thus, in order to assess the shade tree effects on coffee yields we used the results of measurements carried out by the Centro Tropical de Investigación y Enseñanza (CATIE) at the study site, during the period of study and over the 10 years previous to this experiment. Despite the greater coffee LAI and the higher coffee transpiration rates demonstrated in the shade, coffee production was 25% lower in AFS compared to FS from averaged data of the 10 years period. This result may be explained by the shade effect on reduction of the number of nodes per branches, on inhibiting flower bud formation and, therefore, on diminishing fruit load previously reported by various authors (Cannell 1985; Carr 2001; Da Matta 2004). Globally, the effect of shade tree on coffee production has been variable depending on the shade density and environmental condition. Da Matta (2004) proposed, as a general rule, that the more the sub-optimal condition for coffee grown the site is, the greater would be the benefits from shade for coffee production. This is somehow related to the stress gradient hypothesis (Bertness and Callaway 1994) in

which facilitative and competitive interactions will vary inversely across abiotic stress gradients with positive and facilitative interactions being more common in conditions of high abiotic stress. In the sub optimal coffee growing condition studied, lower coffee production in the second year of study, compared to the mean coffee produced over the period of study, seemed to be an effect of the lower precipitation, rather than the mean shade level being 60% of full irradiance averaged over the experiment. In optimal environmental condition for coffee in Costa Rica, with less climatic stress condition than this study, shade cover up to 55% was found to favor the coffee fruit set maintenance (Franck and Vaast 2009). In a study in optimal conditions for coffee cultivation in Mexico, a positive effect was reported on yield within a range of 23% and 38% of shade cover whilst under shade up to 48% coffee yield was maintained and decreased under shade cover above 50% (Soto-Pinto et al. 2000). Great variation in results reported indicates the need for further studies regarding the wide source of influence beyond shade density on coffee production in agroforestry system.

The lower coffee yield in AFS was compensated by greater productivity of the whole system. Cumulative coffee green bean yield measured by CATIE over 2004-2013 (exception for missing data in 2009) consisted in 78,298 kg ha⁻¹ and 58,736 kg ha⁻¹ which corresponded to cumulative values of US\$ 36.445,00 and US\$ 26.538,00 b (adjusted for each year regarding local annual inflation index in the period)^c in FS and AFS, respectively. By the end of the same period, timber production consisted in 601 m³ ha⁻¹ from both the *Tabebuia rosea* and *Simarouba glauca* shade tree species in the study site (Cruz Castillo 2014), which was estimated in US\$ 174/m^{3 d}. The current tree density in the study site generated a shade density of 60% averaged over the period of study that would permit at least 10% of timber tree utilization with minimal impacts on coffee yield. Thus, it would allow obtaining at about US\$ 10.421,00 from timber which summed to the coffee yield inputs would mean 1.4% greater incomes in AFS compared to FS coffee, in that site conditions. Further financial advantages from wood could be accounted by adding the firewood amount produced in the shade system (not recorded). Moreover, studies have demonstrated strong effect of shade on coffee bean size and coffee quality improvements (Muschler 1997; Vaast et al. 2005; Vaast et al.

^b from historical data on annual coffee prices paid for growers - export value – available in International Coffee Organization: http://www.ico.org/new_historical.asp?section=Statistics ^c Available in: www.indexmundi.com/pt/nicaragua

^d from roundwood (non-coniferous tropical) exports value regarding Costa Rica as a reference available in International Tropical Timber Organization: http://www.itto.int/annual review output/

2006) which may enhance incomes in sub optimal condition for coffee cultivation, such as at the study site, due to the better prices in the coffee market. Moreover, it is very probably less risky to have incomes from diversified sources, and those sources combined have less variability than separately. Although the need of more detailed information in order to have the economic balance of both systems compared, data indicated better financial performance in AFS in the site conditions studied. Another study comparing coffee yields in the shade and in the open in Nicaragua by Clemens and Siman (1993) showed coffee yields of 315-630 kg h⁻¹ in shaded traditional technology while in the open sun with high input technology was 1,365-2,730 kg h⁻¹. However, they reported that the balance between the profit and the total costs regarding fertilizers and pesticides resulted in 73% and 28% of rate of return for shaded coffee and for coffee in the open sun, respectively. The great variability in shade tree canopy cover, coffee cultivars, the age of coffee stand, system management and biophysical environments precludes comparison on coffee yield and the whole system production between this experiment and other coffee agroforestry systems.

6.3. Water dynamics and use in coffee agroforestry in sub optimal conditions

6.3.1. Evapotranspiration partitioning

Coffee growth and yield are directly related to the hydrological process in which a very small proportion of the water absorbed by plants is used in the photosynthesis and therefore, in coffee production. Most of water budged is lost by transpiration, soil evaporation, rainfall interception, runoff and infiltration. In agroforestry, coffee and tree interactions affects the water flux in all its components. In this study, evapotranspiration which includes soil surface evaporation plus coffee and shade trees transpiration, as the main component of the water balance, was directly measured in agroforestry and full sun coffee.

We also measured rainfall interception, in which precipitation is captured by the vegetation canopy and subsequently evaporated. Our results demonstrated the effect of coffee and shade tree canopies combined on diminishing the water that reaches the soil in agroforestry. We found that mean annual interception was 24% of gross rainfall in agroforestry compared to 18% in FS. However, it was relatively difficult to compare our results with those of other authors, due to few studies in coffee agroforestry and the huge variations in interception data reported. Our results were found to be high

when comparing to other study with conditions relatively similar to this investigation site with interception of 15.5% and 8.9% of rainfall (Siles et al. 2010; Cannavo et al. 2011). Interception calculation is derived from the difference of rainfall and the sum of stemflow and throughfall, thereby different sources of error may be present. Bias was found on interception estimation, particularly in Tabebuia rosea tree canopy and in coffee canopy in AFS that may be attributed to underestimation of both the stemflow and throughfall measurements. Throughfall, as the largest component in rainfall partitioning seemed to be the major potential influence on the results. However, errors from stemflow measurements, equally hard to measure accurately, should not be discounted, especially during the big rainfall events. Also, we hypothesize that the spatial representativity of rain gauges related to the actual structure of the plantation may be another possible source of bias. A more regular pattern in the rain gauges spatial arrangement rather than the orthogonal distribution used in this experiment could be recommended in order to minimize this potential source of error, but new measurements will be required to test this hypothesis. Also, in this study interception was monitored only over a short period of 28 rainfall events during 2013 wet season, thereby a properly extrapolation of the data for the whole period of experiment was limited.

After being intercepted on the leaves and the bark of trunks and branches, rainfall water is evaporated. It is unlikely that much of this evaporation takes place on the day of the interception, process that could explain the high interception capacity of shade trees: it usually rains during the afternoon, and evaporation diminishes strongly during the night. On the other hand, when this water evaporates, it clearly decreases the net radiation reaching the ground surface, and thus the potential for transpiration or evaporation. That is, adding interception to transpiration or evaporation as components of the water balance is, at least in part, a double counting, from the point of view of the energy balance. It seems that the reality of this double counting depends on the height of the vegetation: in forest trees, the water accumulated by rainfall interception on the leaves evaporates much faster as compared to the transpiration rate; therefore, the double counting is limited. In smaller vegetation, as could be considered the coffee layer in our experiment, the evaporation is much slower and double counting could be an issue (Rutter 1975).

Evapotranspiration partitioning showed that transpiration was the greatest contribution in water consumption in both systems. We demonstrated that transpiration accounted for 83% and 69% of evapotranspiration while soil evaporation represented 17% and 30% in AFS and FS respectively. A similar order of magnitude was found in coffee monoculture in Brazil in which, transpiration and soil evaporation accounted for 65% and 35% of total evapotranspiration, respectively (Flumignan et al. 2011). However, in coffee agroforestry, studies regarding whole evapotranspiration components directly measured are lacking. Most studies of transpiration reliably measured in coffee agroforestry do not take evaporation from the soil surface into account.

Transpiration, which is of high relevance because is directly linked with growth and yield, was found to be greatly dependent on the leaf area. In this study, transpiration and coffee and tree leaf area were measured over short periods that included some days, in the dry and wet seasons, in 2012 and 2013. However, we did not measure leaf area over periods that lasted two-four months, in which great variability of climate conditions should have affected coffee and tree leaf area and therefore, transpiration rates, which was difficult to recover by modelling.

Our results demonstrated that although transpiration constituted the largest component of total evapotranspiration, soil surface evaporation represented a significant term and should be neglected neither in research nor in management of coffee. Soil surface evaporation was 19% greater in FS compared to AFS. Mean daily soil evaporation was 2.59 mm (S.E.=0.16) and 2.10 mm (S.E.=0.14) in FS and AFS, respectively. The significant effect of LAI on evaporation rate was demonstrated (p<0.0001). We also demonstrated that soil evaporation depended a lot on rainfall distribution, as was expected from existing studies in other cropping systems. Soil evaporation was greater in periods of scattered rainfall, due to great evaporation in the first phase after each rainfall compared to periods of large and infrequent rainfall. The lower evaporation rate found in the dry periods despite of higher evaporative demand was similar to findings of Wallace (1991) in arid lands, and may be explained by relatively low rates of water movement toward the surface in unsaturated soil. Indeed, in the dry periods soil evaporation was similar in the row or interrow in both systems while in the wet period differences on lysimeters location was found. In the wet period water loss by soil surface evaporation was always greater in the interrow rather than in the row, despite of the system.

This study showed that shade trees associated with coffee optimized soil water consumption by increasing productive water use as transpiration. Moreover, lower soil evaporation rate in agroforestry compared to full sun coffee may result in less waste of water to the atmosphere that could be available for crop and tree transpiration. The influence of seasonal pattern on plant transpiration and soil evaporation was demonstrated. Results demonstrated that, in the sub optimal condition for coffee cultivation, the dry periods favoured greater evaporative demand with increasing transpiration and at the same time reduction in water loss by soil surface evaporation due to low soil water availability.

6.3.2. Evapotranspiration by directly and indirect estimation compared

In this experiment, evapotranspiration estimates in AFS and FS systems was obtained by two independent approaches: 1) Direct estimates of evapotranspiration by direct measuring soil evaporation and plant transpiration (coffee and shade trees), as presented in Chapter V; 2) Indirect estimates of evapotranspiration by measuring changes in the water stock of the soil profile through Time Domain Reflectometry (TDR) probes monitored in different layers of the soil profile in both systems, as explained in Chapter III.

1) Evapotranspiration was estimated by direct measurement of the main components of water balance such as soil surface evaporation and transpiration of coffee in FS and of coffee and trees in AFS.

 $ET = SE + \Sigma T$

Where: SE = soil surface evaporation; ΣT = transpiration of coffee or coffee and tree.

2) Evapotranspiration was indirectly estimated by measuring changes in soil water stock of the soil profile by using Time Domain Reflectometers - TDR probes (CS616 - Campbell Scientific Inc.). Evapotranspiration was estimated regarding the assumption of lack of drainage and was simplified by the equation:

S1 - S2 + R = Et

where: S1 is the soil water stock in the soil profile at the beginning of the period (mm); S2 is the soil water stock in the soil profile at the end of the period (mm); R is the accumulated rainfall during the period (mm); Et is the evapotranspiration by the soilplant system accumulated during the period (mm). Water consumption comparing evapotranspiration estimation by directly measurements of soil evaporation and plant transpiration; and by evapotranspiration estimated from changes in soil water stock, was found to be of very similar order of magnitude in both systems over the periods studied in spite of the different sources of data and methods utilized in both studies. The similarity of the outcomes and the accuracy of methods and procedures utilized in both approaches reinforced the reliability of the results obtained (Fig 6.1).



Fig 6.1. Daily soil water uptake directly estimated (by soil evaporation and plant transpiration measuring) and indirectly estimated (by changes in the soil water stock) in FS and AFS and reference evapotranspiration over the period of study. Vertical bars represent the standard error of the mean.

We observed that on the whole, water uptake tended to be most influenced by soil water availability than the evaporative demand. Reference evapotranspiration was greater in the dry compared to the wet seasons. However, maximum values of daily water consumption were observed in the wet seasons in both years which suggested that soil water was the main driver of water consumption in both systems in the sub optimal environmental condition studied. Most water uptake in the wet seasons was influenced primarily by leaf area index of coffee and trees which was greater in the wet than in the dry seasons.

We demonstrated that in AFS water uptake was greater while coffee yield was lower compared to FS due to the effect of water use of shade trees and coffee combined. However, enough water availability (although lower than ideally required by coffee cultivation) and deeper root distribution pattern of shade tree did not represent a risk for coffee cultivation in AFS for most of the period of study. In 2012 precipitation was 34% lower than the mean annual rainfall of 1470 mm in the region and an extended and severe dry season occurred in 2013. In such conditions water uptake was very low in both systems and as an effect of coffee and shade tree combined water uptake tended to decrease earlier in AFS and competition for water occurred. This study demonstrated that in these drought conditions coffee transpiration reached its lowest values and coffee leaf water potential indicated a high level of water stress in the agroforestry system.

These results suggest that in the sub optimal environmental conditions studied where the rooting zone was limited to 2 m depth, temperature was high and rainfall was low for coffee cultivation, agroforestry could be a risk in the context of climate change. In such conditions further studies are required to improve design and management in order to minimize competition for water between coffee and shade tree.

6.4. GENERAL CONCLUSIONS

In the sub optimal environmental conditions for coffee cultivation in which this study was done, water was not limiting for most of the period of study and *Tabebuia rosea* and *Simarouba glauca* shade trees could take up water from deep layers reducing competition with coffee which suggest a complementary relationship.

Some complementarity of soil water exploration over the whole soil profile was demonstrated by shallower coffee roots and deeper shade tree root distribution pattern. Desirable root niche differentiation was demonstrated between coffee and *Simarouba glauca* tree that favoured water use from deep layers not available for coffee and that otherwise could be lost by deep drainage.

This study demonstrated that in the sub optimal condition for coffee cultivation most of water was used by transpiration that accounted for 77% and 67% of total evapotranspiration in AFS and FS. Evapotranspiration partitioning showed that water loss by evaporation from soil surface was far from negligible and represented 23% and 33% of total evapotranspiration in AFS and FS respectively. In agroforestry coffee had
the greater contribution to the total transpiration and not the shade trees. Greater transpiration and lower evaporation from the soil surface in agroforestry compared to full sun coffee suggest greater effectiveness in water use in shaded environment regarding the system as a whole but not necessarily for the coffee.

In this study a serious shade tree effect in constraining coffee water consumption was observed over the 2013 severe dry season when transpiration was stabilized and coffee leaf water potential reached its lowest value (-2.33 MPa) which suggested a high level of water stress and a high level of competition between coffee and shade trees.

Our results showed that the effect of shade tree species on coffee water consumption differed with tree species characteristics. We found a temporal differentiation in water use pattern between coffee and deciduous *Tabebuia rosea* which mitigate potential competitive relationship for most of the period of experiment. We also observed a spatial niche differentiation between coffee and evergreen *Simarouba glauca* root systems. Evergreen *Simarouba glauca* water use pattern such as taking up water from deeper soil layers and the lower and more constant water consumption seemed to be more suitable as coffee shade tree when compared to deciduous *Tabebuia rosea* in sub optimal conditions for coffee cultivation.

Results obtained about changes in soil water stock estimation from water loss measurements when compared with results from direct measurement of changes in soil water stock were consistent indicating that methods utilized in this investigation were reliable and suitable for the coffee agroforestry water balance in the study site conditions.

6.4.1. TESTING THE RESEARCH HYPOTHESES

6.4.1.1. Root distribution and water use in coffee shaded by *Tabebuia rosea* and *Simarouba glauca* compared to full sun coffee in sub optimal environmental conditions.

- Trees had no significant influence on coffee root distribution and no difference was found between coffee root system in agroforestry and full sun coffee.
- Coffee roots crossed the compacted soil layer but the restrictive soil conditions had a negative influence on coffee root growth regardless of the system.
- Coffee root growth was greater near *Simarouba glauca* rather than *Tabebuia rosea*.

- Evergreen *Simarouba glauca* and deciduous *Tabebuia rosea* exhibited different root systems behaviour. *Simarouba glauca* and coffee exhibited greater root niche differentiation than Tabebuia rosea and coffee, suggesting complementarity relationships.
- Coffee water uptake was severely restricted in agroforestry compared to full sun coffee over the 2013 severe dry season.

6.4.1.2. Rainfall interception, stemflow and throughfall in a coffee shaded by *Tabebuia rosea* Bertol. and *Simarouba glauca* DC.

- We confirmed that shade tree canopies modified water availability for coffee cultivation in agroforestry system.
- Shade tree characteristics such as leaf area and leaf traits had some influence on the amount of rainfall that reached the soil surface.
- Most of rainfall partitioning was distributed as throughfall.
- We confirmed negligible amount of stemflow in coffee and shade trees when compared to throughfall and rainfall interception.

6.4.1.3. Evapotranspiration in a coffee shaded by *Tabebuia rosea* and *Simarouba glauca* timber tree species compared to unshaded coffee in sub optimal environmental conditions.

- Transpiration comprised the major fraction of evapotranspiration in both systems.
- Agroforestry showed greater transpiration and lower soil evaporation when compared to full sun water use. Coffee transpiration was the greater contributor to the total transpiration in agroforestry.
- We confirmed the hypothesis that water loss by evaporation from the soil surface was reduced by the effect of shade trees associated with coffee crops, even in dry seasons when only evergreen *Simarouba glauca* maintain its leaves.
- Deciduous *Tabebuia rosea* and evergreen *Simarouba glauca* affected soil water budget in different ways. *Tabebuia rosea* exhibited great and highly variable pattern of water consumption compared to *Simarouba glauca*.
- In sub optimal conditions for coffee cultivation a complementarity relationship between coffee and *Tabebuia rosea* and *Simarouba glauca* as shade tree was demonstrated. The hypothesis on competition between coffee and shade tree was confirmed only at the atypically severe dry periods.

6.5. CONSIDERATIONS AND RECOMMENDATIONS FOR FUTURE RESEARCH ON COFFEE WATER RELATIONS IN AGROFORESTRY

Understanding of water dynamics and use by coffee and shade tree is determinant of successful coffee agroforestry design especially in water limiting environments. It is known that in arid and semi-arid environments agroforestry may be not the best option due to greater water consumption by the whole system. In the humid tropics knowledge of the hydrological processes and interactions with different crops and tree species are limited. Studies of water relations in coffee agroforestry are rare and could become more required in the future with climate change.

This study demonstrated that water loss by rainfall interception on the canopies surface is of relevant importance in the coffee agroforestry water balance in sub optimal environmental condition. Despite the great quantity of literature available on the topic little was found in coffee agroforestry. Also, available studies did not include all components for rainfall interception. Due to the inherent difficulty of measuring rainfall partitioning, studies have shown great variability in interception with different environmental conditions and shade tree species. Thus, consensual estimates of interception in coffee agroforestry are still lacking. Regarding a methodological approach we could recommend careful attention to throughfall and stemflow measurements under big rainfall events. Accuracy of results would require a revision in the rain gauges spatial distribution towards a more regular sampling pattern. The tree canopy characteristics and the rainfall interception effects on available rainfall water for coffee should be taken into account by farmers as criteria for shade tree species definition, especially in conditions of limiting water.

Soil surface evaporation as one of the most important components of water balance has been poorly investigated in agroforestry systems. Most of the studies on soil surface evaporation are related to arid and sub arid environments. The shade tree effects on soil evaporation in tropical environment have rarely been studied and very few are available in coffee agroforestry. In the study site conditions soil surface evaporation represented a significant term in the system water loss and should not be neglected in the management decisions. Thus, further research is needed in coffee agroforestry regarding seasonal pattern, different arrangements and environmental conditions in order to identify how to improve whole system water use by maximal reduction of water loss by soil evaporation. The study of root system spatial distribution was found to be an important tool to assess below ground interactions and potential complementarity or competition in agroforestry. Coffee and shade tree root systems evaluated in this study demonstrated the root niche differentiation between shallower coffee roots and *Simarouba glauca* root system that dominated deeper soil layers. This result, previously rarely demonstrated in coffee agroforestry suggested complementarity in water use between both species. However, given the dynamic characteristic of root systems and possible dominant root systems influence it would be important to check species-specific characteristics or behavioural effects on such distribution. Also, in a mature agroforestry system such as at the study site, in which average shade tree spacing was about 8 m, root competition between tree species could be expected, thereby we suggest a study of both coffee and shade tree root system by growing tree species separately and in agroforestry systems in the same environmental condition.

This study reinforced the strategic role of agroforestry on diminishing water loss by soil evaporation and enhancing soil water available for coffee growth and yield. The sub optimal condition for coffee cultivation studied is similar to a wide range of coffee farms all over the world which are predicted to fail as coffee land producer in the next decades. In such water-limiting condition suitable shade tree characteristics and functional traits would contribute to improve the system water use. The contrasting performance of deciduous and evergreen shade tree species on water consumption was demonstrated. Leaf traits were determinant on the water loss by rainfall interception from the shade tree canopies and therefore, on available soil water for coffee growth and yield. We also demonstrated that root system characteristics indicated competition or complementarity in water uptake between coffee and shade tree. However, given the heterogeneity and complex nature of the agroforestry systems, trait-based approaches and process-based models development are required in order to better understand the ecological processes and mechanisms that govern interactions between crops, trees and environment and also predict system behaviour in the current context of global climate change.

REFERENCES (Chapter I, II and VI)

- Albertin, A.; Nair, P.K.R. 2004. Farmers' Perspectives on the Role of Shade Trees in Coffee Production Systems: An Assessment from the Nicoya Peninsula, Costa Rica. Human Ecology 32(4): 443-463.
- Allen, R.G.; Pereira, L.S.; Raes, D.; Smith, M. 1998. Crop evaporation: guidelines for computing crop water requirements. Rome, FAO. 300 p. (FAO Irrigation and Drainage)
- Alvim, P.T. 1973. Factors affecting flowering of coffee. Genes, Enzymes, and Populations 2: 193-202.
- Avelino, J.; Willocquet, L.; Savary, S. 2004. Effects of crop management patterns of coffee rust epidemics. Plant Pathology 53: 541-547.
- Avelino, J.; Cristancho, M.; Georgiou, S.; Imbach, P.; Aguilar, L.; Bornemann, G.; Laderach, P.; Anzueto, F.; Hruska, A.; Morales, C. 2015. The coffee rust crises in Colombia and Central America (2008-2013): impacts, plausible causes and proposed solutions. Food Security 7: 303-321.
- Babbar, L.I.; Zak, D.R. 1994. Nitrogen cycling in coffee agroecosystems: net N mineralization and nitrification in the presence and absence of shade trees. Agriculture, Ecosystems & Environment 48: 107-113.
- Barradas, V.L.; Fanjul, L. 1986. Microclimatic characterization of shaded and opengrown coffee (*Coffea arabica*) plantations in Mexico. Agricultural and Forest Meteorology 38: 101-112.
- Beer, J. 1987. Advantages, disadvantages and desirable characteristics of shade trees for coffee, cacao and tea. Agroforestry Systems 5: 3-13.
- ______. 1995. Efectos de los arboles de sombra sobre la sostenibilidad del cafetal. Boletin Promecafe 68: 13-18.
- Beer, J.; Muschler, R.; Kass, D.; Somarriba, E. 1998. Shade management in coffee and cacao plantations. Agroforestry Systems 38: 139-164.
- Benegas, L.; Ilstedt, U.; Roupsard, O.; Jones, J.; Malmer, A. 2014. Effects of trees on infiltrability and preferential flow in two contrasting agroecosystems in Central America. Agriculture, Ecosystems & Environment 183: 185-196.
- Bengough, A.G.; McKenzie, B.M.; Hallett, P.D.; Valentine, T.A. 2011. Root elongation, water stress, and mechanical impedance: a review of limiting stresses and beneficial root tip traits. Journal of Experimental Botany 62(1): 59-68.

- Bennie, A.T.P. 1996. Growth and mechanical impedance. 2nd ed. Waisel, Y.; Eshel, A.; Kafkafi, U. eds. New York. Basel. Hong Kong, Marcel Dekker, Inc. 453-470 p. (Plant roots. The hidden half)
- Bertness, M.D.; Callaway, R.M. 1994. Positive interactions in communities. Trends in Ecology and Evolution 9: 191-193.
- Bierhuizen, J.F.; Nunes, M.A.; Ploegman, C. 1969. Studies on productivity of coffee. II Effect of soil moisture on photosynthesis and transpiration of *Coffea arabica* Acta Botanica Neerlandica 17(2): 367-374.
- Blanco, M.; Haggar, J.; Moraga, P.; Madriz, J.C.; Pavón, G. 2003. Morfología del café (*Coffea arabica* L.) en lotes comerciales. Nicaragua. Agronomia mesoamericana 14(1): 97-103.
- Bohm, W. 1979. Methods of studying root systems. Billings, W.D.; Golley, F.; Lange, O.L.; Olson, J.S. eds. Berlin Heidelberg New York, Springer-Verlag. 188 p. (Ecological Studies)
- Brodribb, T.J.; Holbrook, N.M.; Edwards, E.J.; Gutiérrez, M.V. 2003. Relations between stomatal closure, leaf turgor and xylem vulnerability in eight tropical dry forest trees. Plant, Cell and Environment (26): 443-450.
- Bull, R.A. 1963. Studies on the effect of mulch and irrigation on root and stem development in Coffea arabica L. Turrialba 13(2): 96-115.
- CABI. 2013. The CABI Encyclopedia of forest trees, CAB International. 536 p.
- Calder, I.R.; Hall, R.L.; Rosier, P.T.W.; H.G., B.; Prasanna, K.T. 1996. Dependence of rainfall interception on drop size: 2. Experimental determination of the wetting functions and two-layer stochastic model parameters for five tropical tree species. Journal of Hydrology 185(1-4): 379-388.
- Calder, I.R. 1998. Water use by forests, limits and controls. Tree physiology 18: 625-631.
- Camargo, M.B.P. 2010. The impact of climatic variability and climate change on arabic coffee crop in Brazil. Bragantia 69(1): 239-247.
- Cannavo, P.; Sansoulet, J.; Harmand, J.M.; Siles, P.; Dreyer, E.; Vaast, P. 2011. Agroforestry associating coffee and *Inga densiflora* results in complementarity for water uptake and decreases deep drainage in Costa Rica. Agriculture, Ecosystems & Environment 140(1-2): 1-13.
- Cannell, M.G.R. 1985. Dry matter partitioning in tree crops. Abbotts Ripton, Institute of Terrestrial Ecology. 160-193 p. (Attributes of trees as crop plants.)

- Cannell, M.G.R.; Van Noordwijk, M.; Ong, C.K. 1996. The central agroforestry hypothesis: the trees must acquire resources that the crop would not otherwise acquire. Agroforestry Systems 34: 27-31.
- Carr, M.K.V. 2001. The water relations and irrigation requirements of coffee. Experimental Agriculture 37(1): 1-36.
- Castaño, O.; Jaramillo, A. 1999. Escurrimiento de agua y nutrimentos por el tallo en árboles de guamo (*Inga* sp). Chinchiná, Cenicafé. 22 p.
- Charbonnier, F. 2013. Measuring and modeling light, water and carbon balance and net primary productivity in a coffee based-agroforestry system of Costa Rica. PhD. France, Université de Lorraine. 230 p.
- Clemens, H.; Siman, J.J. 1993. Tecnologia y desarrollo del sector cafetalero en Nicaragua. Managua, (CIES/ESECA) (93).
- Clowes, M.S.J.; Logan, W.J.C. 1985. Advances in coffee management and technology in Zimbabwe 1980 - 1985 Clowes, M.S.J.; Logan, W.J.C. eds., 181 p.
- Crisosto, C.H.; Grantz, D.A.; Meinzer, F.C. 1992. Effects of water deficit on flower opening in coffee (*Coffea arabica* L.). Tree physiology 10: 127-139.
- Crockford, R.H.; Richardson, D.P. 1990. Partitioning of rainfall in a eucalypt forest and pine plantation in southeastern Australia: II. Stemflow and factors affecting stemflow in a dry sclerophyll eucalypt forest and a Pinus radiata plantation. Hydrological Processes 4: 145-155.
- Cruz Castillo, J.B. 2014. Determinación del índice de área foliar, radiación solar no interceptada por espécies forestales y concentración de magnesio en hojas de café arabica (Coffea arabica L.) en sistemas agroforestales localizados en Masaya, Nicaragua. Graduación engeniero forestal. Managua, Nicaragua, UNA. 37 p.
- Da Matta, F.M.; Maestri, M.; Barros, R.S.; Regazzi, A.J. 1993. Water relations of coffee leaves (*Coffea arabica* and *C. canephora*) in response to drought. Journal of Horticultural Science 68: 741-746.
- Da Matta, F.M. 2004. Ecophysiological constraints on the production of shaded and unshaded coffee: a review. Fields Crops Researchs 86: 99-114.
- Da Matta, F.M.; Ramalho, I.D.C. 2006. Impacts of drought and temperature stress on coffee physiology and production: a review. Plant physiology 18(1): 55-81.
- Da Matta, F.M.; Ronchi, C.P.; Maestri, M.; Barros, R.S. 2007. Ecophysiology of coffee growth and production. Brazilian Journal of Plant Physiology 19(4): 485-510.

- Daamen, C.C.; Simmonds, L.P.; Sivakumar, M.V.K. 1995. The impact of sparse millet crops on evaporation from soil in semi- arid Niger. Agricultural Water Manage 27(3): 225-242.
- Dauzat, J.; Rapidel, B.; Berger, A. 2001. Simulation of leaf transpiration and sap flow in virtual plants: model description and application to a coffee plantation in Costa Rica. Agricultural and Forest Meteorology 109: 143-160.
- Di Rienzo, J.; Casanoves, F.; Balzarini, M.; Gonzalez, L.; Tablada, M.; Robledo, C. 2014. InfoStat Manual. Argentina., Universidad Nacional de Córdoba. 336 p.
- Dzib, B.; Vaast, P.; Harmand, J.M.; Llandera, T. 2006. Manejo, almacenamiento de carbono e ingresos económicos obtenidos de árboles maderables en fincas cafetaleras de tres regiones contrastantes de Costa Rica. Agroforesteria en las Americas 10(41-42): 44-53.
- Elfving, D.C.; Kaufmann, M.R.; Hall, A.E. 1972. Interpreting Leaf Water Potential measurements with a model of soil plant atmosphere continuum. Physiologia plantarum 27: 161-168.
- Escalante, M.; Somarriba, E. 2001. Diseño y manejo de los cafetales del occidente de El Salvador. Agroforesteria en las Americas 8(30): 12-16.
- Int. Symp. For. Hydrol. 1967. Interception and soil moisture distribution. Sopper, W.E., Hull, H.W. (Eds.). ed. Toronto, Pergamon Press. 191-200. p.
- Fanjul, L.; Arreola-Rodriguez, R.; Mendez-Castrejon, M.P. 1985. Stomatal Responses to Environmental Variables in Shade and Sun Grown Coffee Plants in Mexico. Experimental Agriculture 21 (3): 249-258.
- FAO, (Food and Agriculture Organization of the United Nations). 2009. Global Agriculture Towards 2050. Rome, 150 p. (How to feed the world 2050)
 - _____. 2013. Statistical Yearbook. Rome, 289 p. (World food and agriculture)
- Fisher, E.F.; Victor, B. 2014. High-end coffee and smallholding growers in Guatemala. Latin American Research Review 49(1): 155-177.
- Fitter, A. 1996. Characteristics and functions of root systems. *In.* 1996. Root plants. The hidden half. Tel Aviv. Israel, p. 1-20.
- Flumignan, D.L.; Faria, R.T.; Prete, C.E.C. 2011. Evapotranspiration components and dual crop coefficients of coffee trees during crop production. Agricultural Water Management 98: 791-800.
- Franck, N.; Vaast, P. 2009. Limitations of coffee leaf photosyntesis by stomatal conductance and light availability under different shade levels. Trees 23: 761-769.

- Franco, C.M.; Inforzato, R. 1946. O sistema radicular do cafeeiro nos principais tipos de solo do Estado de Sao Paulo. Bragantia 6: 443-478.
- Gash, J.H.C. 1979. An analytical model of rainfall interception by forests. Journal of the Royal Meteorological Society 105: 43-55.
- Geisler, C.; Nadrowski, K.; Kühn, P.; Baruffol, M.; Bruelheide, H.; Schmid, B.; Scholten,
 T. 2013. Kinetic Energy of Throughfall in Subtropical Forests of SE China –
 Effects of Tree Canopy Structure, Functional Traits, and Biodiversity.
 10.1371/journal.pone.0049618
- Gentry, A.H. 2002. Tropical forest biodiversity: distributional patterns and their conservational significance. OIKOS 63(1): 19-28.
- Germer, S.; Elsenbeer, H.; Moraes, J.M. 2006. Throughfall and temporal trends of rainfall redistribution in an open tropical rainforest, south-western Amazonia (Rondonia, Brazil). European Geosciences Union 10(3): 383-393.
- Goebes, P.; Bruelheide, H.; Härdtle, W.; Kröber, W.; Kühn, P. 2015. Species-specific effects on throughfall kinetic energy in subtropical forest plantations are related to leaf traits and tree architecture. 10.1371/journal.pone.0128084
- Gomez-Delgado, F.; Roupsard, O.; Moussa, R.; Le Maire, G.; Taugourdeau, S.; Bonnefond, J.M.; Perez, A.; Van Oijen, M.; Vaast, P.; Rapidel, B.; Voltz, M.; Imbach, P.; Harmand, J.M. 2010. Modelling the hydrological behaviour of a coffee agroforestry basin in Costa Rica. Hydrology and Earth System Sciences 7: 3015-3071.
- Goulden, M.L.; Field, C.B. 1994. Three methods for monitoring the gas exchange of individual tree canopies: ventilated-chamber, sap-flow and Penman-Monteith measurements on evergreen oaks. Functional Ecology 8: 125-135.
- Grady, A.P.O.; Eamus, D.; Hutley, L.B. 1999. Transpiration increases during the dry season: patterns of tree water use in eucalypt open-forests of northern Australia. Tree physiology 19: 591-597.
- Granier, A. 1985. Une nouvelle méthode pour la mesure du flux de seve brute dans le tronc des arbres. Annales des Sciences Forestieres 42: 193-200.
 - ______. 1987. Evaluation of transpiration in a Douglas-fir stand by means of sap flow measurements. Tree physiology 3: 309-320.

- Guiscafre-Arrilaga, J.; Gomez, L. 1938. Studies of the root system of Coffea arabica L.I. Environment conditions affecting the distribution of coffee roots in Coloso clay. Journal of Agriculture of the University of Puerto Rico 22(2): 227-262.
- Gurav, M.; Kumar, S.; G., K.C.; Vaast, P. 2012. Throughfall and Interception Loss in Relation to Different Canopy Levels of Coffee Agroforestry Systems. International Journal of Environmental Sciences 1(No.3): 145-149.
- Gutiérrez, M.V.; Harrington, R.A.; Meinzer, F.C.; Fownes, J.H. 1994a. The effect of environmentally induced stem temperature gradients on transpiration estimates from the heat balance method in two tropical woody species. Tree physiology 14: 179-190.
- Gutiérrez, M.V.; Meinzer, F.C. 1994. Estimating water use and irrigation requirements of coffee in Hawaii. Journal of the American Society for Horticultural Science 119(3): 652-657.
- Gutiérrez, M.V.; Meinzer, F.C.; Grantz, D.A. 1994b. Regulation of transpiration in coffee hedgerows: covariation of environmental variables and apparent responses of stomata to wind and humidity. Plant, Cell and Environment 17: 1305-1313.
- Gutierrez, P.S. 2007. Hydrological processes (water use and balance) in a coffee (Coffea arabica L.) monoculture and a coffee plantation shaded by Inga densiflora in Costa Rica. Ph.D., Nancy university. 51 p.
- Haggar, J.; Ayala, A.; Díaz, B.; Reyes, C.U. 2001. Participatory design of agroforestry systems: developing farmer participatory research methods in Mexico. Development in Practice 11(4): 417-424.
- Haggar, J.; Barrios, M.; Bolaños, M.; Merlo, M.; Moraga, P.; Munguia, R.; Ponce, A.; Romero, S.; Soto, G.; Staver, C.; de M. F. Virginio, E. 2011. Coffee agroecosystem performance under full sun, shade, conventional and organic management regimes in Central America. Agroforestry Systems 82(3): 285-301.
- Hare, P.D.; Cress, W.A.; Van Staden, J. 1998. Dissecting the roles of osmolyte accumulation during stress. Plant, Cell & Environment 21: 535-553.
- Harmand, J.M.; Avila, H.; Dambrine, E.; Skiba, U.; De Miguel, S.; Renderos Duran, R.V.; Oliver, R.; Jimenez, F.; Beer, J. 2007. Nitrogen dynamics and soil nitrate retention in a Coffea arabica-Eucalyptus deglupta agroforestry system in Southern Costa Rica. Biogeochemistry 85: 125-139.
- Harvey, C.A.; Komar, O.; Chazdon, R.; Ferguson, B.G.; Finegan, B.; Griffith, D.M.; Ramos, M.M.; Morales, H.; Nigh, R.; Soto-Pinto, L.; Van Breugel, M.; Wishnie,

M. 2008. Integrating Agricultural Landscapes with Biodiversity Conservation in the Mesoamerican Hotspot Conservation Biology 22(1): 8-15.

- Holt-Giménez, E. 2002. Measuring farmers' agroecological resistance after Hurricane Mitch in Nicaragua: a case study in participatory, sustainable land management impact monitoring. Agriculture, Ecosystems and Environment 93: 87-105.
- Horton, R.E. 1919. Rainfall interception. *In* Brooks, C.F. ed. 1919. Monthly Weather Review. N.Y., p. 603-623.
- Huxley, P. 1996. Biological factors affecting form and function in wood-non-woody plant mixtures. *In* Ong, C.K.; Huxley, P. eds. 1996. Tree-crop interactions: a physiological approach. UK, CAB International. p. 235-298.
- Huxley, P.A.; Patel, R.Z.; Kabaara, A.M.; Michell, H.W. 1974. Tracer studies with P on the distribution of functional roots of arabica coffee in Kenya. Annals of Applied Biology 77: 159-180.
- Imbach, A.C.; Fassbender, H.W.; Beer, J.; Borel, R.; Bonneman, A. 1989. Sistemas agroforestales de café (Coffea arabica) con laurel (Cordia alliodora) y café con poro (Erythrina poeppigiana) en Turrialba, Costa Rica. *In.* 1989. Acta Agronómica - Universidad Nacional de Colombia. Turrialba, p. 400-414.
- Jackson, N.A.; Wallace, J.S. 1999. Soil evaporation measurements in an agroforestry system in Kenya. Agricultural and Forest Meteorology 94: 203-215.
- Jackson, N.A. 2000. Measured and modelled rainfall interception loss from an agroforestry system in Kenya. Agricultural and Forest Metereology 100: 323-336.
- Jackson, R.B.; Canadell, J.E., J.R.; Mooney, H.A.S., O.E.; Schulze, E.D. 1996. A global analysis of root distributions for terrestrial biomes Oecologia 108: 389-411.
- Jaramillo, A. 2003. La lluvia y el transporte de nutrimentos dentro de ecosistemas de bosque y cafetales. Cenicafe 54: 134-144.
- Jaramillo, A.R.; Chaves, B.C. 1998. Intercepcion de la lluvia en un bosque y en plantaciones de *Coffea arabica* L. Cenicafe 42: 129-135.
- ______. 1999. Aspectos hidrológicos en un bosque y en plantaciones de café (*Coffea arabica* L.) al sol y bajo sombra. Cenicafe 50(2): 97-105.
- Jiménez, A.E.; Goldberg, D. 1982. Efecto de differentes estructuras vegetales sobre el balance hidrico del cafetal. *In* Jiménez, A.E., Gomez, P.A. (Eds.). ed. 1982. Estudios Ecologicos en Agroecosistema Cafetalero. Cuidad de Mexico, Editora Continental. p. 39-54.

- Jimenez, F.; Alfaro, R. 1999. Available soil water in *Coffea arabica-Erytrina poeppigiana, Coffea arabica-Eucalyptus deglupta* and *Coffea arabica* monoculture plantations. Proceedings of the 4th Semana Científica Catie: 203-206.
- Johnson-Maynard, J.L.; Graham, R.C.; Wu, L.; Shouse, P.J. 2002. Modification of soil structural and hydraulic properties after 50 years of imposed chaparral and pine vegetation. Geoderma 110: 227-240.
- Jonckheere, I.; Fleck, S.; Nackaerts, K.; Muysa, B.; Coppin, P.; Weiss, M.; Baret, F. 2004. Review of methods for in situ leaf area index determination Part I. Theories, sensors and hemispherical photography. Agricultural and Forest Meteorology 121: 19-35.
- Kool, D.; Agam, N.; Lazarovitch, N.; Heitman, J.L.; Sauer, T.J.; Ben-Gal, A. 2014. A review of approaches for evapotranspiration partitioning. Agricultural and Forest Meteorology 184: 56- 70.
- Kumar, S.; Anderson, S.H.; Brinckel, L.G.; Udawatta, R.P.; Gantzer, C.J. 2008. Benefits of agroforestry and grass buffers in grazed pasture system. Journal of Soil and Water conservation. 63(4): 224-232.
- Kunert, N.; Schwendenmann, L.; Ischer, D.H. 2010 Seasonal dynamics of tree sap flux and water use in nine species in Panamanian forest plantations. Agricultural and Forest Meteorology 150: 411-419.
- Laclau, J.P.; Arnaud, M.; Bouillet, J.P.; Ranger, J. 2001. Spatial distribution of Eucalyptus roots in a deep sand soil in the Congo: relationships with the ability of the stand to take up water and nutrients. Tree physiology 21: 129-136.
- Lawrence, D.M.; Thornton, P.E.; Oleson, K.W.; Bonan, G.B. 2006. The Partitioning of Evapotranspiration into Transpiration, Soil Evaporation, and Canopy Evaporation in a GCM: Impacts on Land–Atmosphere Interaction. Journal of hydrometeorology 8: 862-880.
- Levia, D.F.; Frost, E.E. 2003. A review and evaluation of stemflow literature in the hydrological and biogeochemical cycles of forested and agricultural ecosystems. Journal of Hydrology 274: 1-29.
- Lin, B.B. 2007. Agroforestry management as an adaptive strategy against potential microclimate extremes in coffee agriculture. Agricultural and Forest Meteorology 144: 85-94.

- Lin, B.B.; Perfecto, I.; Vandermeer, J. 2008. Synergies between Agricultural Intensification and Climate Change Could Create Surprising Vulnerabilities for Crops Bioscience 58(9): 847-854.
- Lin, B.B. 2010. The role of agroforestry in reducing water loss through soil evaporation and crop transpiration in coffee agroecosystems. Agricultural and Forest Meteorology 150: 510-518.
- Liu, C.; Zhang, X.; Zhang, Y. 2002. Determination of daily evaporation and evapotranspiration of winter wheat and maize by large-scale weighing lysimeter and micro-lysimeter. Agricultural and Forest Meteorology 111: 109-120.
- Llanderal, T.; Somarriba, E. 1999. Tipologias de cafetales en Turrialba, Costa Rica Agroforesteria en las Americas 6(23): 30-32.
- López-Gómez, A.M.; Williams-Linera, G.; Manson, R.H. 2008. Tree species diversity and vegetation structure in shade coffee farms in Veracruz, Mexico. Agriculture, Ecosystems and Environment 124: 160-172.
- Lunka, P.; Patil, S.D. 2016. Impact of tree planting configuration and grazing restriction on canopy interception and soil hydrological properties: implications for flood mitigation in silvopastoral systems. Hydrological Processes 30(6): 945-958.
- Marin, C.T.; Boutena, W.; Sevink, J. 2000. Gross rainfall and its partitioning into throughfall, stemflow and evaporation of intercepted water in four forest ecosystems in western Amazonia. Journal of Hydrology 237: 40-57.
- Mazzafera, P.; Teixeira, J.P.F. 1989. Prolina em cafeeiros submetidos à déficit hídrico. IICA Turrialba 39(3): 305-313.
- Meinzer, F.C.; Grantz, D.A.; Goldstein, G.; Saliendra, N.Z. 1990. Leaf water relations and maintenance of gas exchange in coffee cultivars grown in drying soil. Plant physiology 94: 1781-1787.
- Meinzer, F.C.; Grantz, D.A. 1991. Coordination of stomatal, hydraulic, and canopy boundary layer properties: Do stomata balance conductances by measuring transpiration? Physiologia plantarum 83: 324-329.
- Meinzer, F.C.; Andrade, J.L.; Goldstein, G.; Holbrook, N.M.; Cavelier, J.; Wright, S.J. 1999. Partitioning of soil water among canopy trees in a seasonally dry tropical forest. Oecologia 121: 293-301.
- Méndez, V.E. 2008. Farmers' livelihoods and biodiversity conservation in a coffee landscape of El Salvador Cambridge, U.S.A., MIT Press. 207-236 p. (Confronting the coffee crisis: Fair Trade, sustainable livelihoods and ecosystems in Mexico and Central America)

- Morales, E.; Beer, J. 1998. Distribución de raíces finas de *Coffea arabica* y *Eucalyptus deglupta* en cafetales del Valle Central de Costa Rica. Agroforesteria en las Americas 5: 44-48.
- Mulia, R.; Dupraz, C. 2006. Unusual fine root distributions of two deciduous tree species in southern France: What consequences for modelling of tree root dynamics? Plant and Soil (281): 71-85.
- Muschler, R.G. 1997. Effectos de sombra de Erythrina poeppigiana sobre Coffea arabica vars. Caturra y Catimor. 18º Simposio Latinoamericano de Caficultura. San Jose, Costa Rica, 157-162 p.
- Muschler, R.G.; Bonnemann, A. 1997. Potentials and limitations of agroforestry for changing land-use in the tropics: experiences from Central America Forest Ecology and Management 91(1): 61-73.
- Nair, P.K.R. 1984. Role of trees in soil productivity and conservation. In: Soil productivity aspects of agro-forestry. The International Council for Research in Agro-forestry. Nairobi. (Chapter 3): 29-49.
- Nunes, M.A.; Duarte, N.T. 1969. Resistencia a secura em *Coffea arabica*. Estudo comparativo de duas seleções. Garcia de Orta, Lisboa 17(1): 83-96.
- Nutman, F. 1933a. The root-system of *Coffea arabica* L. Part I. Root.system in typical soils of British East Africa. Empire Journal of Experimental Agriculture 1: 271-284.
- Nutman, F. 1934. The root-system of *Coffea arabica.* Part III. The spacial distribution of the absorbing area of the root. Empire Journal of Experimental Agriculture 2: 293-302.
- Nutman, F.J. 1933b. The root system of *Coffea arabica* Part II. The effect of some soil conditions in modifying the 'normal' root system. Empire Journal of Experimental Agriculture 1: 285-296.
- Nutman, F.J. 1937. Studies of the Physiology of *Coffea arabica*: I. Photosynthesis of Coffee Leaves under Natural Conditions. Annals of Botany 1(3): 353-367.
- Ong, C.K.; Black, C.R. 1994. Complementarity of resource use in intercropping and agroforestry systems. *In* Monteith, J.L.; Scott, R.K.; Unsworth, M.H. eds. 1994. Resource Capture by Crops. Loughborough, Nothinghan University Press. p. 255-278.
- Ong, C.K. 1995. The 'dark side' of intercropping: manipulating of soil resources. The ecophysiology of Tropical Intercropping. : 45-66.

- Ong, C.K.; Black, C.R.; Marshall, F.M.; Corlett, J.E. 1996. Principles of resource capture and utilization of light and water Huxley, C.K.O.a.P. ed. University Press, Cambridge. UK, CAB International and ICRAF. (Tree-crop interactions. A physiological approach)
- Ong, C.K.; Anyango, S.; Muthuri, C.W.; Black, C.R. 2007. Water use and water productivity of agroforestry systems in the semi-arid tropics. Annals of Arid Zone 46(3-4): 255-284.
- Park, A.; Cameron, J.L. 2008. The influence of canopy traits on throughfall and stemflow in five tropical trees growing in a Panamanian plantation. Forest Ecology and Management 255: 1915-1925.
- Penman, H.L. 1963. Vegetation and hydrology. Commonwealth Burg. of soils 53: 124.
- Pinheiro, J.C.; Bates, D.M. 2009. Mixed-Effects Models in S and S-PLUS. 2nd edition ed. New York, Springer. 530 p. (Statistics and computing)
- Ramos, L.C.S.; Carvalho, A. 1997. Shoot and root evaluations on seedlings from Coffea genotypes. Bragantia 56 (1): 59-68.
- Rao, M.R.; Nair, P.K.R.; Ong, C.K. 1998. Biophysical interactions in tropical agroforestry system. Agroforestry Systems 38: 3-50.
- Rapidel, B.; Roupsard, O. 2009. Sap flow measurement in coffee tree: comparison of two methods, sources of error and correction methods. *In* II World Congress of Agroforestry (Nairobi, Kenya) 2009. ICRAF. ed. p.
- Rice, R.A.; Ward, J.R. 1996. Coffee, Conservation and Commerce in the Western Hemisphere Washington, DC, Smithsonian Migratory Bird Center.
- Ritchie, J.T. 1971. Dryland Evaporative Flux in a Subhumid Climate: I. Micrometeorological Influences. Agronomy Journal 63(1): 51-55.
 - ______. 1972. Model for predicting evaporation from a row crop with incomplete cover. Water Resource Research 8(5): 1204-1213.
- Roupsard, O.; Ferhi, A.; Granier, A.; Pallo, F.; Deppomier, D.; Mallet, B.; Jolly, H.I.; Dreyer, E. 1999. Reverse Phenology and dry season water uptake by *Faidherbia albida* (Del.) A Chev. in agroforestry parkland of Sudanese West Africa. Functional Ecology 13: 460-472.
- Rutter, A.J. 1963. Studies in the Water Relations of *Pinus Sylvestris* in Plantation Conditions I. Measurements of Rainfall and Interception. Journal of Ecology 51(1): 191-203.

- Rutter, A.J.; Kershaw, K.A.; Robins, P.C.; Morton, A.J. 1971. A predictive model of rainfall interception in forests, 1. Derivation of the model from observations in a plantation of *Corsican pine*. Agricultural Meteorology 9: 367-384.
- Rutter, A.J. 1975. The hydrological cycle in vegetation. *In* 1975. Monteith, J.L. ed. Vegetation and the atmospere. London, Academic Press. p. 111-154.
- Samper, K. 1999. Trayectoria y Viabilidad de las Caficulturas Centroamericanas Desafios de la Caficultura en Centroamerica: 1-68.
- Sanchez, P.A. 1995. Science in agroforestry. Agroforestry Systems 30: 5-55.
- Schaller, M.; Schroth, G.; Beer, J.; Jimenez, F. 2003. Species and sites characteristics that permit the association of fast-growing trees with crops: the case of Eucalyptus deglupta as coffee shade in Costa Rica. Forest Ecology and Management 175: 205-215.
- Scholander, P.F.; Hammel, H.T.; Bradstreet, E.D.; Hemmingsen, E.A. 1965. Sap pressure in vascular plants. Science 148: 339-346.
- Schroth, G. 1995. Tree root system characteristics as a criteria for species selection and systems design in agroforestry. Agroforestry Systems 30: 125-143.
- Schroth, G.; Silva, L.F.; Wolf, M.A.; Teixeira, W.G.; Zech, W. 1999. Distribution of throughfall and stemflow in multi-strata agroforestry, perennial monoculture, fallow and primary forest in central Amazonia, Brazil. Hydrological Processes 13: 1423-1436.
- Siles, P. 2007. Hydrological process (water use and balance) in a coffee (Coffea arabicaL.) monoculture and a coffee plantation shaded by Inga densiflora in CostaRica. PhD. Nancy Université France.
- Siles, P.; Harmand, J.M.; Vaast, P. 2009. Effects of Inga densiflora on the microclimate of coffee (*Coffea arabica* L.) and overall biomass under optimal growing conditions in Costa Rica. Agroforestry Systems 78(3): 269-286.
- Siles, P.; Vaast, P.; Dreyer, E.; Harmand, J.M. 2010. Rainfall partitioning into throughfall, stemflow and interception loss in a coffee (*Coffea arabica* L.) monoculture compared to an agroforestry system with *Inga densiflora.* Journal of Hydrology 395: 39-48.
- Sklenicka, P.; Lhota, T.; Cecetka, J. 2002. Soil porosity along a gradiente from forest edge to field Bodenkultur 53: 191-197.
- Smith, D.M.; Allen, S.J. 1996. Measurement of sap flow in plant stems. Journal of Experimental Botany 47(305): 1833-1844.

- Smith, M.; Burgess, S.S.O.; Suprayogo, D.; Lusiana, B.; Widianto, M. 2004. Uptake, partitioning and redistribution of water by roots in mixed species agroecosystems. *In* Van Noordwijk, M. ed. 2004. Below-ground interactions in tropical agroecosystems : concepts and models with multiple plant components. Cambridge, MA, USA, CABI Publishing. p. 157-170.
- Somarriba, E.; Harvey, C.A.; Samper, M.; Anthony, F.; Gonzalez, J.; Staver, C.; Rice, R. 2004. Biodiversity conservation in neotropical coffee (*Coffea arabica*) plantations. *In* Schroth, G. ed. 2004. Agroforestry and biodiversity conservation in tropical landscapes. Island press. p. 198-226.
- Staver, C.; Gurahay, F.; Monterroso, D.; Muschler, R.G. 2001. Designing pest supressive multistrata perennial crop systems: shade grown coffee in Central America. Agroforestry Systems 53: 151-170.
- Sylvain, P. 1955. Some observations on *Coffea arabica* L. in Ethiopia. Turrialba 5: 37-53.
- Tabios, G.Q.; Salas, J.D. 1985. A comparative analysis of techniques for spatial interpolation of precipitation. Journal of the American Water Resources Association 21: 365-380.
- Tardieu, F. 1988. Analysis of the spatial variability of maize root density I. Effect of wheel compaction on the spatial arrangement of roots. Plant and Soil (107): 259-266.
- Taugourdeau, S.; Le Maire, G.; Avelino, J.; Jones, J.R.; Ramirez, L.G.; Quesada, M.J.;
 Charbonnier, F.; Gomez-Delgado, F.; Harmand, J.M.; Rapidel, B.; Vaast, P.;
 Roupsard, O. 2014. Leaf area index as an indicator of ecosystem services and
 management practices: an application for coffee agroforestry. Agriculture,
 Ecosystems & Environment 192: 19-37.
- Tausend, P.C.; Goldstein, G.; Meinzer, F.C. 2000. Water utilization, plant hydraulic properties and xylem vulnerability in three contrasting coffee (*Coffea arabica*) cultivars. Tree physiology 20: 159-168.
- Tilman, D.; Cassman, K.G.; Matson, P.A.; Naylor, R.; Polasky, S. 2002. Agricultural sustainability and intensive production practices Nature 418: 671-677.
- Udawatta, R.P.; Krstansky, J.J.; Henderson, G.S.; Garrett, H.E. 2002. Agroforestry Practices, Runoff, and Nutrient Loss: A Paired Watershed Comparison. Journal of Environmental Quality (31): 1214-1225.

- Udawatta, R.P.; Anderson, S.H.; Gantzer, C.J.; Garrett, H.E. 2006. Agroforestry and grass buffer influence on macropore characteristics: a computed tomography analysis. Soil Science Society of America Journal 70: 1763-1773.
- Udawatta, R.P.; Anderson, S.H.; Motavalli, P.P.; Garrett, H.E. 2011. Calibration of a water content reflectometer and soil water dynamics for an agroforestry practice. Agroforestry Systems 82: 61-75.
- Vaast, P.; Kanten, R.V.; Siles, P.; Dzib, B.; Franck, N.; Harmand, J.; Genard, M. 2005. Shade: a key factor for coffee sustainability and quality. *In* ASIC 2004. 20th International Conference on Coffee Science, Bangalore, India, 11-15 October 2004. 2005. Association Scientifique Internationale du Café (ASIC). p. 887-896.
- Vaast, P.; Bertrand, B.; Perriot, J.; Guyot, B.; Genard, M. 2006. Fruit thinning and shade improve bean characteristics and beverage quality of coffee (Coffea arabica L.) under optimal conditions. Journal of the Science of Food and Agriculture 86: 197-204.
- Vaast, P.; Van Kanten, R.; Siles, P.; Angrand, J.; Aguilar, A. 2007. Biophysical interactions between timber trees and Arabica Coffee in suboptimal conditions of Central America. Towards Agroforestry Design: an ecologycal approach 4(Chapter 9): 135-148.
- van Kanten, R.; Vaast, P. 2006. Transpiration of arabica coffee and associated shade tree species in sub-optimal, low-altitude conditions of Costa Rica. Agroforestry Systems 67: 187-202.
- Van Noordwijk, M.; Widianto, M.; Heinen, M.; Hairah, K. 1991. Old tree root channels on acid soils in the humid tropics: Important for crop root penetration, water infiltration and nitrogen management. Plant and Soil 134: 37-44.
- Van Noordwijk, M.; Lawson, G.; Soumaré, A.; Groot, J.J.R.; Hairiah, K. 1996. Root distribution of trees and crops: competition and/or complementarity. *In* Huxley, C.K.O.a.P. ed. 1996. Tree-crop interactions. CAB International. p. 319-359.
- Van Noordwijk, M.; Brouwer, G.; Meijboom, F.; Do Rosario, G.; Oliveira, M.; Bengough,
 A.G. 2000. Trench profile techniques and core break methods. *In* Smit, A.L.;
 Bengough, A.G.; Engels, C.; Van Noordwijk, M.; Pellerin, S.; Van der Geijn, S.C.
 eds. 2000. Root Methods, a Handbook. Berlin, Springer Verlag. p. 211-233.
- Vertessy, R.A.; Benyon, R.G.; O'Sullivan, S.K.; Gribben, P.R. 1995. Relationships between stem diameter, sapwood area, leaf area and transpiration in a young mountain ash forest. Tree physiology 15: 559-567.

- Vogel, A.W.; Acuña Espinales, E. 1995. Reference soils of the Pacific Coastal Plain with a hardpan (Talpetate). International Soil Reference and Information Centre and Universidad Nacional Agraria. 21 p. (Soil brief Nicaragua) (3).
- Wallace, J.S. 1991. The measurement and modelling of evaporation from semi-arid land. *In* Sivakumar, M.V.K.; Wallace, J.S.; Renard, C.; Giroux, C. eds. 1991. Soil Water balance in the Sudano-Sahelian one Wallingford, UK, IAHS. p. 131-148.
- Wallace, J.S.; A., J.N.; Ong, C.K. 1997. Soil evaporation in a Grevillea robusta agroforestry system. Modelling the effects of tree canopy shade. Agricultural and Forest Meteorology.
- Wallace, J.S.; Jackson, N.A.; Ong, C.K. 1999. Modelling soil evaporation in an agroforestry system in Kenya. Agricultural and Forest Meteorology 94(189-202).
- Wilcox, B.P.; Breshears, D.D.; Seyfried, M.S. 2003. Water balance on rangeland. *In* Howell, B.A.S.a.T.A. ed. 2003. Encyclopedia of water science. New York, Marcel Dekker, Inc. p. 791-794.
- Williams, D.G.; Cable, W.; Hultine, K.; Hoedjes, J.C.B.; Yepez, E.A.; Simonneaux, V.; Er-Raki, S.; Boulet, G.; de Bruin, H.A.R.; Chehbouni, A.; Hartogensis, O.K.; Timouk, F. 2004. Evapotranspiration components determined by stable isotope, sap flow and eddy covariance techniques. Agricultural and Forest Meteorology 125: 241-258.
- Wormer, T.M. 1965. Some physiological problems of coffee cultivation in Kenya. Café 6: 1-20.
- Xiao, Q.; McPherson, E.G.; Ustin, S.L.; Grismer, M.E.; Simpson, J.R. 2000. Winter rainfall interception by two mature open-grown trees in Davis, California Hydrological Processes (14): 763-784
- Xu, Z.; Yang, H.; Liu, F.; An, S.; Cui, J.; Wang, Z.; Liu, S. 2008. Partitioning evapotranspiration flux components in a subalpine shrubland based on stable isotopic measurements. Botanical Studies 49: 351-361.
- Yepez, E.A.; Williams, D.G.; Scott, R.L.; G., L. 2003. Partitioning overstory and understory evapotranspiration in a semiarid savanna woodland from the isotopic composition of water vapor. Agricultural and Forest Meteorology 119(1-2): 53-68.
- Yocum, W.W. 1937. Root development of young delicious apple trees as affected by soils and by cultural treatments. University of Nebraska Research Bulletin (95): 1-55.

Yunusa, I.A.M.; Walker, R.R.; Lu, P. 2004. Evapotranspiration components from energy balance, sapflow and microlysimetry techniques for an irrigated vineyard in Inland Australia. Agricultural and Forest Meteorology 127: 93-107.